

Qu, Chunlin (2024) *Advanced hydrogen-terminated diamond field effect transistor technology*. PhD thesis.

https://theses.gla.ac.uk/84831/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

### Advanced Hydrogen-Terminated Diamond Field Effect Transistor Technology

Chunlin Qu B.Eng., M.Sc. Supervisor: Prof. David A. J. Moran

Submitted in fulfilment of the requirements for the Degree of Doctor of Philosophy

James Watt School of Engineering College of Science and Engineering University of Glasgow



December 2024

## Abstract

Diamond demonstrates excellent properties for use in high-power and high-frequency electronic devices due to its wide band gap, high carrier mobility, and thermal conductivity. Traditional doping methods prove difficult in diamond due to its strong carbon to carbon bonds and the deep energy levels of most dopants. To overcome these difficulties, transfer doping has emerged as a viable alternative, particularly for hydrogen terminated p-type doped diamond. Hydrogen-terminated diamond is widely used for MOSFET fabrication because of its low thermal activation energy.

Given the sensitive nature of hydrogen termination of diamond, this work explores multiple alternative fabrication methods to replace the conventional gold sacrificial layer and protect the hydrogen termination from damage caused by plasma or air exposure during the fabrication process. H-diamond MESFETs with and without ALD  $Al_2O_3$  encapsulation were fabricated and measured. Device properties such as drain current and gate leakage current were observed to improve after  $Al_2O_3$  deposition. After the encapsulation, the drain current improved from  $16.3 \ mA/mm$  to  $37.5 \ mA/mm$ , while the gate leakage current dropped from  $2.49 \ mA/mm$  to  $0.05 \ mA/mm$ . However, high-temperature measurements revealed device degradation at elevated temperatures up to 450K, with the drain current decreasing from 39.6 mA/mm at room temperature to 10 mA/mm at 450K.

Accumulation channel MOSFETs on H-diamond were fabricated using high-temperature annealing and both PVD and ALD  $Al_2O_3$ . With this method, MOSFETs on H-diamond do not rely on transfer doping formed by air adsorbates and hydrogen termination. However, hydrogen termination is still required under the gate to form an accumulation channel. These devices exhibited normally-off properties ( $I_{on}/I_{off} = 10^9$ ) and achieved a maximum drain current of 110 mA/mm. The threshold voltage ( $V_{th} < -5 V$ ) represents the lowest state-of-the-art value for hydrogen-terminated diamond MOSFETs. The carrier concentration and mobility extracted by the CV analysis is  $2 \times 10^{12} cm^{-2}$  and  $110 cm^2/V \cdot s$ , respectively.

## Contents

Al	bstrac	t	i		
A	cknow	vledgements	xi		
As	Associated Publications xii				
De	eclara	tion	xiii		
1	Intr	oduction	1		
2	Bacl	kground: Diamond as a Semiconductor	3		
	2.1	Diamond Structure	3		
	2.2	Synthetic Diamond Production	6		
	2.3	Doping of Diamond	8		
	2.4	Surface transfer doping of diamond	11		
	2.5	Metal semiconductor interfaces	14		
		2.5.1 Schottky Contact	14		
		2.5.2 Ohmic Contact	17		
		2.5.3 MOS Contact	19		
	2.6	Field Effect Transistor operation	22		
		2.6.1 MOSFET	22		
		2.6.2 MESFET	24		
	2.7	Chapter Summary	25		
3	Lite	rature Review	26		
	3.1	Surface transfer doping	26		
	3.2	Diamond surface hydrogen termination	32		
	3.3	Spillover mechanics	33		
	3.4	Contact metal on hydrogen terminated diamonds	35		
	3.5	Field Effect transistors	35		

	3.6	Chapter Summary	39
4	Fabr	rication	40
	4.1	Sample Preparation	40
	4.2	Resist spinning	45
	4.3	Photo-lithography exposure	46
	4.4	Electron beam lithography	48
	4.5	Resist Development	50
	4.6	Metallisation & lift-off	50
	4.7	Selective Etching of metals	53
	4.8	Oxide deposition	54
	4.9	Diamond MESFETs Fabrication Flow	58
	4.10	Accumulation Channel diamond MOSFETs Fabrication Flow	63
	4.11	Chapter Summary	69
5	Cha	ractorization and Massuramont	70
3	5 1	Surface Profiling	70
	5.1	5.1.1 SEM	70
		5.1.2 AEM	70
		5.1.2 ArW $\ldots$ Surface Profiler DektakXT	73 77
	52	Flectrical Measurement	78
	5.2	5.2.1 Transmission Line Measurement (TLM)	78
		5.2.1 Transmission Line Weasurements (VDP)	80
		5.2.2 Van Der Fadwinkensenenens (VDF)	83
	5.3	Chapter Summary	86
	<b>a</b>		0.
6	Surf	ace Termination Results	87
	6.1		87
	6.2	Spillover with optimal parameters	90
	6.3	Chapter Summary	96
7	Alte	rnative Fabrication Results	97
	7.1	Lift-off Resist as protective layer	98
	7.2	Hydrogen Plasma resist residue removal	108
	7.3	Pd as contact metal	112
	7.4	Chapter Summary	115
8	Hyd	rogen terminated Diamond MESFETs	116
	8.1	VDP and TLM structures with Au as contact metal	116
	8.2	MESFET measurement before and after air exposure	122

A	Арро	endix	158
	10.2	Future work	156
	10.1	Conclusions	155
10	Cone	clusions and Future Work	155
	9.4	Chapter Summary	153
	9.3	Split CV measurement and mobility extraction	146
	9.2	MOSFETs measurement	141
	9.1	$PVD Al_2O_3$ deposition	139
9	Accu	mulation Channel H-Diamond MOSFETs	138
	8.5	Chapter Summary	136
	8.4	MESFET high temperature measurement	131
	8.3	MESFET with thermal ALD $Al_2O_3$	127

## **List of Tables**

2.1	Semiconductor properties comparison between CVD Diamond, Si, GaN and	
	4H-SiC [19], [21]–[23]	6
4.1	Samples surface roughness $R_q$ before and after scaife polish	41
4.2	Sheet resistance change after ALD deposition under different growth conditions.	57
6.1	Experiment parameters of three different samples	89
6.2	Sheet resistance of SC16 after spillover (100nm of Pd strip)	92
6.3	Sheet resistance of SC22 after spillover (100nm of Pt metal strip)	92
6.4	Sheet resistance of SC10 after spillover (Blank control group)	94
6.5	Surface roughness of SC16 (Pd) before and after spillover	94
6.6	Surface roughness of SC22 (Pt) before and after spillover	94
6.7	Surface roughness of SC10 (control) before and after spillover	95
7.1	VDP measurement of Sample A and B before and after the fabrication	101
8.1	Sheet resistance and contact resistance after 30 days of air expose	118
8.2	VDP measurement of the sample.	121
8.3	Comparison of device before and after 14 days of air expose	127
8.4	Comparison of device before and after oxide deposition.	131
8.5	Extracted resistance from the output characteristics plot at different temperature.	134
9.1	Hall Kit sheet resistance measurement of the sample (1) in air (2) after 1 hour $300 ^{\circ}C$ annealing and 15 nm of $Al_2O_3$ deposition (3) $Al_2O_3$ removed by 5 minutes BOE wet etch.	140
A.1	Parameters for the resist used in this work. PMMA is e-beam resist, and S1805 is photo-resist. LOR stands for Lift-off resist which is not sensitive to either UV	
	light nor e-beam, however it can be lift-off and etched by TMAH	158
A.2	Parameters for the two stages acid cleaning process used in this work	158
A.3	Parameters for diamond hydrogen termination provided by Dr. Alastair Stacy	
	from Australia.	159

# **List of Figures**

2.1	Carbon atomic structure and ground state electronic configuration [13]		
2.2	Graphite atomic Structure	4	
2.3	Diamond lattice structure [17]	5	
2.4	Carbon phase under different conditions and the conditions for HPHT diamonds		
	and CVD diamonds. Direct HPHT and indirect synthesis refer to the process		
	without or with diamond seeds [25]	7	
2.5	Diagram of CVD growth chamber [29]	8	
2.6	N-type and P-type doped diamond lattice with Phosphorus and Boron as example.	9	
2.7	The energy levels of different dopants in diamond	10	
2.8	The atomic structure of diamond surface with different terminations	11	
2.9	Energy band diagram of diamond with different surface terminations	12	
2.10	Using air adsorbates as an example, energy band diagram for surface transfer		
	doping on a hydrogen-terminated diamond [43]	13	
2.11	Energy band diagram of P-type Schottky contact.	15	
2.12	P-type Schottky contact with forward bias	16	
2.13	P-type Schottky contact with reverse bias	16	
2.14	Band diagram of p-type semiconductor ohmic contact.	18	
2.15	Ideal MOS Structure	19	
2.16	Energy band diagram of MOS contact.	20	
2.17	Energy band diagram of MOS contact.	21	
2.18	p-channel MOSFET cross section. Enhancement mode (left) and depletion		
	mode (right)	22	
2.19	p-channel MOSFET $I_{DS}$ - $V_{DS}$ curve	23	
2.20	p-channel depletion mode MESFET with no bias and positive bias.	24	
3.1	Energy levels of various molecular orbitals and H-diamond with different sur-		
	face orientation [44].	28	
3.2	Band diagram of $MoO_3$ /H-diamond before and after thermal equilibrium [49].	29	

3.3	Sheet Resistance of H-diamond with and without transition metal as surface acceptor[62].	30
3.4	Electron affinities of different metal oxides compared to the bandgap of diamond	
	with no termination, oxygen termination and hydrogen termination[62].	31
3.5	Hydrogen spillover process.	33
3.6	Hydrogen spillover on different support material.	34
3.7	Structures of first H-diamond MESFETs[99].	36
3.8	Structures of Crawford demonstrated FETs[93]	37
3.9	Cross section of T-shaped gate MOSFETs[104]	38
4.1	AFM scan results of sample surface before polish	41
4.2	AFM scan results of sample surface After polish	42
4.3	AFM scan results of sample surface after the cleaning process	42
4.4	AFM scan results and Hall kit measurement of sample surface before and after	
	hydrogen plasma treatment.	44
4.5	Resist before and after spinning.	45
4.6	Resist thickness versus spinning speed with ebeam resist as example[116]	45
4.7	Photolithography with photo-mask contact with the sample surface	46
4.8	Photolithography Mask Aligner 6 (MA6)	47
4.9	Marker used in photo-lithography.	47
4.10	Simplified representation of EBPG. Information from [117]	49
4.11	Typical marker sets used in Ebeam lithography.	49
4.12	Resist development and ashing	50
4.13	Simplified electron beam evaporation chamber.	51
4.14	Lift-off process for sample with single layer of resist (a) and sample with double	
	layers of resist (b)	52
4.15	Gold wet etch with isotropic etching profiles	54
4.16	The inner structure of Plassys 5 (used for PVD of oxide materials)	55
4.17	PVD and ALD oxide deposition profile	55
4.18	A complete ALD cycle with ALD $Al_2O_3$ as an example	56
4.19	Fabrication of hydrogen terminated diamond MESFETs with gold sacrificial	
	layer to protect the surface conductivity and ALD $Al_2O_3$ as encapsulating layer.	59
4.20	Marker used in photo-lithography	60
4.21	Sample with gold after isolation step	61
4.22	MESFETs after Bond Pad lift off.	62
4.23	Active region fabrication before and after gold etch.	62
4.24	MESFETs after gate lift off.	63
4.25	Resist over bond pad and gate pad is exposed and developed. Thermal ALD	
	$Al_2O_3$ is wet etched by BOE to provide access to bondpads for probing	63

4.26	Normally-off "Accumulation Channel" hydrogen terminated diamond MOS-	61
4 07		04
4.27	The L-edit design of MOSFETS and images of isolation and marker structure.	00
4.28		6/
4.29	Optical microscope image of bond pad oxide etching	68
4.30	Optical microscope image of bond pad after lift off.	68
4.31	Optical microscope image of gate level fabrication.	69
5.1	Illustration of signals generated by the interaction of incident electrons and the	
	sample surface in Scanning Electron Microscopy.	71
5.2	General SEM configuration. [126]	72
5.3	General configuration of Atomic Force Microscopy	74
5.4	AFM tips and sample surface interaction curve.	75
5.5	Image of different width of AFM tips scan a surface with high aspect ratio and	
	density.	75
5.6	Double image caused by chipped AFM tip.	76
5.7	Scanner and Feedback-related artifacts of AFM	77
5.8	Configuration of Surface Profiler Dektak XT. [130]	78
5.9	Optical Microscope Image of TLM structure on hydrogen terminated diamond	
	surface.	79
5.10	TLM gap graph	79
5.11	Ideal Plot of $R_T$ against $L_x$ .	80
5.12	Hall effect	81
5.13	Nanometrics 550 Hall Kit 4 probes hall measurement system	81
5.14	Microscope image of VDP structure for hall effect measurement	82
5.15	Simplified FET DC characteristics graph.	83
5.16	Example FET output characteristics.	84
5.17	Simplified FET DC characteristics for $R_{on}$	84
5.18	Example FET transfer characteristics.	85
		~ ~
6.1	Image of masking the samples for metal deposition.	88
6.2	Diamond sample were split into 25 areas and 4 probes hall measurement is car-	
	ried in each area following Spillover	91
6.3	Contact angle measurement of SC16 before (left) and after (after) the spillover.	95
6.4	Contact angle measurement of SC22 before (left) and after (after) the spillover.	95
6.5	Contact angle measurement of SC10 before (left) and after (after) the spillover.	95
7.1	Fabrication steps of LOR method (Left) and gold sacrificial laver method(Right).	99
7.2	Structures of gold sacrificial layer methods (left)[20] and LOR lift off method	-
	(right)	100

7.3	Optical image of VDP structure for Hall measuremnt	101
7.4	Optical image of TLM structure	101
7.5	TLM result of sample A fabricated with LOR (Applied voltage 0 - 5 V)	102
7.6	TLM result of sample A fabricated with LOR below 0.5V	102
7.7	TLM results analyse of Sample A when data extracted below 0.5 V	103
7.8	TLM result of sample B fabricated with LOR	104
7.9	Sample surface roughness after four times of LOR spin and etch	105
7.10	Simplified fabrication flow of resist residue test.	106
7.11	Optical image of LOR residue test structure fabrication. The first step of hori-	
	zontal array written (left), exposure of vertical array (mid), vertical array LOR	
	wet etch (right)	106
7.12	2D and 3D rendering AFM image of area between the first expose and second	
	expose	107
7.13	Histogram graph of AFM scan	107
7.14	CAD image of e-beam lithography job design for hydrogen plasma etching test.	108
7.15	Resist thickness of pattern A before (left) and after (Right) dry etch with coil	
	power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins	110
7.16	Resist thickness of pattern B before (left) and after (Right) dry etch with coil	
	power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins	110
7.17	Resist thickness of pattern C before (left) and after (Right) dry etch with coil	
	power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins	110
7.18	Hydrogen plasma etching rate of PMMA resist versus coil power change	111
7.19	Hydrogen plasma etching rate of PMMA resist versus hydrogen flow rate change.	111
7.20	Fabrication flow of TLM with Pd or Au as ohmic contact metal	113
7.21	Au as ohmic metal TLM measurement results	114
7.22	Pd as ohmic metal TLM measurement results	114
81	Optical image of sample with completed structures and devices febricated with	
0.1	and as sacrificial layer	117
87	TI M results with sample exposed to air day 1 and day 7 after AI D $A_{la}O_{a}$	11/
0.2	deposition	118
83	Sheet resistance and contact resistance before and after $A_{\rm L}O_{\rm c}$ and under differ	110
0.5	ent temperature	110
81	Output characteristics for 2 <i>um</i> MESEETs with H diamond surface expose to air	112
0.4 9 5	Transfer characteristic of $2\mu m$ MESEETs with H diamond surface expose to air.	122
0.J 8.6	$\begin{array}{c} \text{Prime} \text{ for } 2\mu m \text{ WESFETs with re-diamond surface expose to all.} \\ \text{Output characteristics for } 2\mu m \text{ MESFETs with atmospheric adoptates ofter } 14 \end{array}$	123
0.0	Super characteristics for $2\mu m$ westers with atmospheric ausorbates after 14 days of air expose	104
87	Transfer characteristic of $2\mu m$ MESEETs with atmospheric adsorbates ofter 14	124
0./	Transfer characteristic of $2\mu m$ messers with atmospheric adsorbates after 14 days of air expose	125
	uays of all expose	1 <i>L</i> J

8.8	Output characteristics multiple sweep with gate bias $V_{gs} = -2.5V.$	126
8.9	Off-state break test of devices with gate lengths of $2\mu m$ and $5\mu m$	126
8.10	FET structure with atmospheric adsorbates (top) and 15nm of ALD $Al_2O_3$ (bot-	
	tom)	127
8.11	Output characteristics for 5 $\mu m$ MESFETs with atmospheric adsorbates after 14	
	days of air expose.	128
8.12	Output characteristics for 5 $\mu m$ MESFETs with ALD $Al_2O_3$	128
8.13	Transfer characteristics for 5 $\mu m$ MESFETs with atmospheric adsorbates after	
	14 days of air expose.	130
8.14	Transfer characteristics for 5 $\mu m$ MESFETs with ALD $Al_2O_3$	130
8.15	Output characteristics at 300 K, 350 K and 350 K for 30 mins.	132
8.16	Output characteristics at 400 K, 400 K after 30 mins and 450 K.	133
8.17	Transfer characteristics at different temperature.	135
8.18	Transconductance at different temperature.	136
9.1	Energy diagram between the gate channel and source drain area	139
9.2	MOSFETs cross section structure.	141
9.3	25 $\mu$ m wide, 3 $\mu$ m gate length AC-MOSFETs output characteristics	142
9.4	25 $\mu$ m wide, 3 $\mu$ m gate length AC-MOSFETs transfer characteristics	142
9.5	MOSFETs off-state breakdown measurement	145
9.6	Benchmarks of threshold voltage and drain current maximum of the reported	
	state-of-the-art normally-off hydrogen termination diamond MOSFETs	146
9.7	MOSFETs CV measurement at 1 MHz	147
9.8	The capacitor distribution in the AC-MOSFETs	147
9.9	Carrier density versus gate voltage in logarithmic scale	148
9.10	25 $\mu m$ wide, 1 $\mu m$ gate length AC-MOSFET Output characteristics	149
9.11	25 $\mu$ m wide, 1 $\mu$ m gate length AC-MOSFET Transfer characteristics	150
9.12	TLM measurement results after $Al_2O_3$ etched off	150
9.13	Sheet resistance versus gate source voltage.	151
9.14	Mobility versus gate source voltage.	152
9.15	Benchmarks of carrier mobility versus the carrier density of the reported state-	
	of-the-art hydrogen termination diamond.	153

## Acknowledgements

I would like to take this opportunity to express my heartfelt thanks to everyone who has supported me throughout my PhD journey over the past few years.

Firstly, I would like to thank my supervisor, Prof. David Moran, for offering me this position. He has been an excellent mentor, helping me tackle various problems and find the best solutions when needed. I wouldn't have come this far without your guidance and patience.

To my second supervisor, Dr. Alexey Ganin, thank you for providing suggestions and access to equipment in the School of Chemistry to complete my research.

To Dr. Alastair Stacey, thank you for all the amazing hydrogen termination processes and your suggestions regarding hydrogen plasma etching.

To all the staff in the JWNC and Microwave and Terahertz Lab, thank you for your help with fabrication and auto probe measurement. Your hard work in maintaining the lab made all these experiments possible.

To all my friends and colleagues, life is much more colourful with your company. Thanks for all the laughter, jokes, food, movies, games, coffee, gossip, and holidays. Most importantly, thank you for being there when I needed your help.

To my families, especially my parents, no words can adequately express my appreciation. You have always stood behind me, supporting every decision I made without hesitation. I wouldn't make it without your support.

## **Associated Publications**

### **Journal Entries**

Chunlin Qu, Isha Maini, Qing Guo, Alastair Stacey, and David A. J. Moran, (2024) *Extreme Enhancement-Mode Operation Accumulation Channel Hydrogen-Terminated Diamond FETs with*  $V_{th} < -6$  V and High On-Current. (Under Review)

Chunlin Qu, Isha Maini, Jingyi Zhang, Alexey Ganin, David A. J. Moran, (2024) *Catalytic*enhanced thermal hydrogen-temination of diamond for electronic applications. (Under Review)

### Conferences

Chunlin Qu, Isha Maini, Kevin G. Crawford and David A.J. Moran, (2022) *Lift-off Resist as a Protective Layer for Hydrogen-terminated Diamond MOSFET Fabrication*, Surface and Bulk Defects in Diamond (SBDD) workshop 2022, Hasselt, Belgium 9 - 11 March. (Poster)

Chunlin Qu, Isha Maini, Kevin G. Crawford and David A.J. Moran, (2022) *Aluminium as a Protective Layer for Hydrogen-Terminated Diamond MOSFET Fabrication*, UK Semiconductors 2022, Sheffield, UK 6 - 8 July. (Oral)

## Declaration

With the exception of chapters 1, 2 and 3, which contain introductory material, all work in this thesis was carried out by the author unless otherwise explicitly stated.

## Chapter 1

### Introduction

Diamond possesses several remarkable properties that make it highly desirable for various applications, including electronic devices.

Firstly, its wide band gap of 5.47 eV enables diamond-based electronic devices, such as MOS-FETs, to potentially operate at high temperatures without excess excitation of electrons or holes, ensuring stable performance even in extreme conditions. Additionally, this wide band gap contributes to its exceptionally high breakdown field of 10 MV/cm, surpassing that of other materials like 4H-SiC and GaN [1], [2]. This high breakdown field allows diamond to withstand high voltages without experiencing impact ionization or avalanche breakdown.

Moreover, diamond exhibits extraordinary thermal conductivity, four times higher than 4H-SiC and twenty times higher than GaN [3], [4]. This outstanding thermal conductivity, stemming from its efficient phonon transport facilitated by strong chemical bonding and light atomic mass, enables efficient thermal conduction, minimizing thermal losses in high-power diamond-based devices.

Another advantageous property of diamond is its high carrier mobility. In intrinsic single-crystal CVD diamond, the reported achieved highest mobility of electrons and holes in diamond is 1945  $cm^2/V \cdot s$  and 2285  $cm^2/V \cdot s$ , respectively [5]. In comparison, widely used high-temperature electronic materials like SiC have significantly lower carrier mobility for electrons (950  $cm^2/V \cdot s$ ) and holes (260  $cm^2/V \cdot s$ ) [6], [7]. This high carrier mobility in diamond allows carriers to reach saturation velocity faster, enhancing the overall performance and speed of electronic devices.

Overall, the combination of these outstanding properties makes diamond an attractive material for various electronic applications, offering potential high-temperature operation, high break-

#### CHAPTER 1. INTRODUCTION

down voltage, efficient thermal management, and superior carrier mobility.

Boron and phosphorus were well explored as dopants in diamond to prepare p-type and n-type diamond [8], [9]. However, the high thermal activation energy of Boron-doped diamond (0.37 eV) and Phosphorus (0.6 eV) limited its application. Hydrogen-terminated diamond has been extensively studied in the last decades as a substitution for p-type diamond. The lowest sheet resistance, 719  $\Omega/\Box$ , of hydrogen-terminated diamond was reported by Sato et al. [10]. And the hole mobility of the hydrogen-terminated diamond has been reported between 3  $cm^2/V \cdot s$  to  $300 \ cm^2/V \cdot s$  [11], [12].

In this work, the fabrication of Field Effect Transistors with hydrogen-terminated diamond is explored. This thesis starts with the background and literature review of hydrogen-terminated diamond and related theoretical introduction.

In Chapter 4, the fabrication process is discussed in detail. Each fabrication step used in this work is discussed thoroughly. The last two sections of this chapter give the fabrication flows of hydrogen-terminated diamond MESFETs and MOSFETs in detail. Chapter 5 gives a brief introduction to the equipment (SEM, AFM, and Surface Profiler) used for surface profiling and methods used for electrical measurement.

Chapter 6 explores a new method, "Spillover", to hydrogen-terminate the diamond surface rather than using hydrogen plasma. Low-concentration hydrogen (5%), high temperature (900°C), and catalyst metals (Pd and Pt) were used in this work to complete the termination process.

Chapter 7 concludes alternative fabrication methods explored to protect the hydrogen termination on the diamond surface during the fabrication process or form a better ohmic contact. Lift-off resist (LOR) was explored to be used as a Au sacrificial layer substitution and Pd was explored to be used to form an ohmic contact on hydrogen-terminated diamond.

Chapter 8 discusses the electrical characteristics of hydrogen-terminated MESFETs. The device properties were measured during air exposure and at elevated temperatures. TLM and VDP structures were fabricated on the same sample and measured to help understand the ohmic contact behaviour against air exposure duration and elevated temperature.

Chapter 9 reports on accumulation-mode hydrogen-terminated diamond MOSFETs fabricated with physical vapor deposition (PVD)  $Al_2O_3$ . Devices' transfer and output characteristics are discussed to understand the devices' behaviour. The Split C-V method is used to extract the carrier mobility and concentration. Record low threshold voltage (<-5 V) combined with high drain current (110 mA/mm) were achieved with these devices.

### Chapter 2

### **Background: Diamond as a Semiconductor**

Diamond possesses many properties, such as wide band gap, robustness and chemical inertness, making it an attractive material for many areas, including electrical applications. This chapter will discuss diamonds as a promising material for semiconductor electronic device applications.

The chapter commences by examining the diamond crystal structure and delves into the advantages that diamond derives from its stringent arranged crystal lattice. Following this is a brief introduction to the diamond's growth methods. Then impurities doping and surface transfer doping of diamond are discussed and compared in detail, which leads to the reason that later choice is employed in this research. Succeeding this is an introduction of metal-semiconductor (MES-FET) and metal-oxide-semiconductor (MOSFET) theories to explain the operation of hydrogenterminated diamond FETs.

### 2.1 Diamond Structure

Carbon, a fundamental element in the periodic table, serves as the cornerstone of life on our planet. The atomic structure of carbon is shown in Figure 2.1. There are 2 electrons on 1s orbital and two on 2s orbital while the remaining two electrons are on p orbitals. Figure 2.1 shows the electronic configuration of ground-state carbon. Two electrons on 2s orbitals can migrate to the p orbital in the excited state and hybridise with other electrons from different atoms to form various materials.



Figure 2.1: Carbon atomic structure and ground state electronic configuration [13].

In nature, elemental carbon has two allotropes: graphite with sp2-hybridisation and diamond with sp3-hybridisation [14]. In sp2-hybridisation, one 2s orbital and two p orbitals hybridise into three orbitals with an angle of  $120^{\circ}$  between each orbital. This hybridisation is also called trigonal hybridisation. In the case of graphite, one carbon atom is connected to 3 adjacent carbon atoms with covalent bonding on a layer plane. The overlap of sp2-hybridisation orbitals forms three  $\sigma$ -bonding, the most robust covalent bonding, and leaves a loose  $\pi$ -electron that plays a vital role in graphite electronic properties [15]. On the opposite of solid bonding between each carbon atom on the same plane layer, each plane layer is bonded to others with weak Van der Walls force that makes graphite shows different properties in the vertical direction. The presence of Van der Walls force in the vertical direction makes it possible to extract a solitary layer of graphite out of the bulk graphite. And this single layer of graphite is known as graphene.



Figure 2.2: Graphite atomic Structure.

In sp3-hybridisation, one 2s orbital and three p orbitals hybridise into four orbitals with an angle of 109.5° between two adjacent orbitals [16]. This hybridisation is also named tetrahedral hybridisation and is applied in the diamond structure. When assembling many carbon atoms with sp3-hybridisation, each carbon atom can connect to 4 other carbon atoms, and the overlapped orbitals form 4  $\sigma$ -bonding in a three-dimensional space. This structure contributes to many of diamond's properties. First, those strong carbon-carbon covalent bonds make diamond one of the hardest and chemically inert materials. In the excited state, all four electrons are occupied with forming the bonds in the diamond, and no extra electron can be excited to the conduction band with visible wavelength. Therefore pure diamond shows a transparent appearance [16].



Figure 2.3: Diamond lattice structure [17].

Because carbon atoms are small in size, they can approach each other relatively closely before experiencing a repulsive force, which is a characteristic that contributes to the formation of short carbon bonds [18]. This short bond and the overlap of orbitals give rise to a significant energy separation between antibonding orbital and bonding orbitals. That means a big forbidden band sits between the conduction and valence bands. In diamond, the forbidden band gap has an energy of  $5.47 \ eV$  at room temperature (300 K) [19]. Therefore, an intrinsic diamond is usually considered an insulator or ultra wide-band gap semiconductor. That makes diamond potentially suitable for high-temperature electronic applications because the wide band gap can stop excess excitation of electrons from the top of the valence band to the bottom of the conduction band [20].

As a wide-bandgap material, diamond lacks enough carriers to travel through the material and carry thermal energy for heat dissipation. However, the rigid carbon structure and covalent bonds help diamonds to transport phonon effectively, which gives diamonds the highest thermal conductivity ( $24 W cm^{-1}K^{-1}$ ) among known wide band gap semiconductors [19]. The high thermal conductivity makes diamonds a promising candidate for high-power devices.

$$V_s = \sqrt{\frac{8E_{opt}}{3\pi m^*}} tanh(E_{opt}/2kT)$$
(2.1)

Applied high electric fields increase carriers' drift velocity in semiconductors. Nevertheless, the interaction between carriers and crystal lattice makes carriers lose their energy rapidly by optical phonon emission. Together with other scattering mechanisms, semiconductor carriers always reach a velocity that can not be exceeded; that is known as the saturation velocity ( $V_s$ ) as shown in Equation 2.1. Saturation velocity benefits from high optical energy ( $E_{opt}$ ) and low effective mass ( $m^*$ ), and diamond has the highest optical phonon energy that gives rise to a very high saturation velocity,  $1.2 \times 10^7 \text{ cm s}^{-1}$  and  $2.7 \times 10^7 \text{ cm s}^{-1}$  for holes and electrons, respectively [19]. A high saturation velocity makes diamonds a potential material for the high-frequency application.

Intrinsic Materials Properties				
	CVD Diamond	Si	GaN	4H-SiC
Bandgap( <i>eV</i> )	5.47	1.1	3.44	3.2
Holes Saturation velocity( $\times 10^{-7} cm s^{-1}$ )	1.2	0.72	0.63	1
Electron Saturation velocity( $\times 10^{-7} cm s^{-1}$ )	2.7	0.86	2.5	3
Breakdown field( $MVcm^{-1}$ )	10	0.3	5	3
Hole mobility( $cm^2V^{-1}s^{-1}$ )	3800	480	200	120
Electron mobility( $cm^2V^{-1}s^{-1}$ )	4500	1450	440	900
Thermal conductivity( $Wcm^{(-1)k^{(-1)}}$ )	24	1.5	1.3	5

Table 2.1: Semiconductor properties comparison between CVD Diamond, Si, GaN and 4H-SiC [19], [21]–[23].

Except for those properties mentioned above, the carrier mobility in ultra-pure and single crystal diamond has been measured at 2285  $cm^2/V$  for electrons and 1945  $cm^2/V$  for holes, which is also the highest recorded values among wide-bandgap materials [5]. That means carriers moving through a diamond can reach the saturation velocity at lower electric field. Table 2.1 shows the comparison of diamond and other semiconductor materials.

### 2.2 Synthetic Diamond Production

In nature, carbon-containing materials, such as organic matter, experience extreme pressure and heat near the earth's mantle area, forming natural diamonds. However, the high cost and uncontrollable impurities of natural diamonds mean that they are not suitable for electronic device fabrication. Compared to natural diamonds, synthetic diamonds exhibit higher purity, lower cost, better consistency and customizable size and shapes [24]. Those advantages make synthetic diamonds a better choice for device fabrication.



Figure 2.4: Carbon phase under different conditions and the conditions for HPHT diamonds and CVD diamonds. Direct HPHT and indirect synthesis refer to the process without or with diamond seeds [25].

From a thermodynamic perspective, diamond is unstable compared to graphite, meaning that if a suitably high pressure and high temperature are applied, graphite can be transformed into diamond [26]. In the 1950s, General Electric Company announced the first high-pressure and high-temperature (HPHT) diamond [27]. Diamond seeds were placed in a chamber with a pressure of 5-6 Gpa and temperature of 1300 °C to 1600 °C. The carbon source, usually graphite and catalyst, would be sent into the growth chamber to stimulate diamond synthesis [25].

HPHT method synthesises diamonds by applying temperature and pressure at the region where the diamond is at its stable regime. Diamonds can also be formed at their meta-stable regime, which leads to chemical vapour deposition (CVD). Hydrogen and hydrocarbons, such as  $C_2H_2$  or  $CH_4$ , the mixture is introduced in the reaction chamber where different techniques like plasma, hot surface and combustion flames are used to activate reactants to create radicals, molecular, and ionic species [28]. This process happens at a pressure of 2 to 50 Kpa, much lower than that of HPHT methods. Meanwhile, a lower temperature, compared to the HPHT method, on the substrate surface is kept at 600 °C to 1200 °C[25]. Cleaned substrates with or without diamond seeds are used to prepare single-crystal or polycrystalline diamonds. Because of the use of hydrogen, CVD synthetic diamond shows a hydrogen-terminated surface. Figure 2.5 illustrates the plasma assisted CVD diamond growth chamber and the growth process.



Figure 2.5: Diagram of CVD growth chamber [29].

Compared to the HPHT method, the CVD method is more widely used because of the lower cost benefits from lower pressure, higher purity, easier doping, and higher scalability. In this work, all of diamonds used for devices fabrication are CVD-grown diamonds.

### 2.3 Doping of Diamond

Impurity doping intentionally adds small amounts of foreign atoms (impurities) to semiconductor materials to modify its electrical properties. The impurities are usually from a different element group of elements than the semiconductor and act as either donor or acceptor, creating excess electrons (*n*-type doping) or holes (*p*-type doping).

Ion implantation and diffusion are two critical methods for impurity doping [30]. In the diffusion doping process, semiconductor materials are placed under a high-temperature environment with a gas containing the desired dopant flow through the surface. The chamber pressure influences the doping depth and concentration. Since the impurities need to travel through the semiconductors, this method may lead to a deviated uniformity in the vertical direction. In addition, diamonds' tight and stringent lattice structure makes diffusion difficult [31].

Compared to diffusion doping, ion implantation has better uniformity. In the ion implantation process, energetic ions beam with energy between  $300 \ eV$  and  $5 \ MeV$  are introduced into the

semiconductor. The ion beam can travel up to 10  $\mu m$  under the substrate's surface, so ion implantation has better uniformity than diffusion [30]. However, the energetic ion beams are always accompanied by lattice damage. In diamond, the metastable sp3 configuration can be converted to graphite with stable sp2 configuration in the following annealing process that is aimed to remove the lattice damages caused by the ion beams [32].

Thus, during-growth-doping is more commonly used for diamonds. As previously mentioned, chemical vapour deposition (CVD) and high-pressure high-temperature (HPHT) are used for diamond synthetic production. The substance that contains desired dopants can be introduced into the growth chamber to acquire doped diamonds [33]. Though the arrangement of the atoms in the diamond structure limits the choice of available impurities. Boron, phosphorus and nitrogen are widely used in doping diamonds.



Figure 2.6: N-type and P-type doped diamond lattice with Phosphorus and Boron as example.

As mentioned above, carbon has four electrons in the valence shell. An *n*-type doped diamond requires an element, such as nitrogen, phosphorus or sulfur, with more than four electrons in the valence shell [34]. When the desired dopant substitutes one carbon atom's position, the valence shell's additional electron is introduced into the diamond lattice. The most common impurity in diamond is nitrogen [35]. Diamond can be classified into two types, Type I and Type II , in terms of the existence of nitrogen. While in natural diamonds, nitrogen tends to be aggregated (Type Ia), in synthetic diamonds, it is typically present as single substitution atoms (Type Ib). Nitrogen acts as a donor in diamond lattice, but the ionisation energy level ( $1.7 \ eV$ ) is far too deep to be used as a useful donor at room temperature [36]. At room temperature, sulfur-doped diamond has been observed to exhibit mobility between  $250 - 600 \ cm^2 V^{-1} s^{-1}$ , along with a thermal activation energy of  $0.38 \ eV$  [37]. Nevertheless, later research shows that unintentionally induced boron impurities could cause it. Currently, the only unambiguous n-type dopant for diamond is phosphorus which has an activation energy level of  $0.6 \ eV$  at room temperature [8].



Figure 2.7: The energy levels of different dopants in diamond.

For a p-type doped diamond, an element with fewer electrons on valence orbit is required to create a vacancy at the donor's level. Boron-doped p-type diamonds are found in nature and developed in synthetic diamonds [35]. The addition of boron-containing gas, such as diborane, into the mixture gas of hydrogen, carbon contains gas during the CVD growth process achieves boron-doped p-type diamonds [38]. Though the boron-doped diamonds are well developed, compared to the activation energy level of typical boron-doped silicon (0.045 eV), the boron-doped diamond still shows a relatively high activation energy of 0.37 eV.

Because of the high activation energy level of both p-type doped diamond and n-type doped diamond, at room temperature, the concentration of carriers in doping diamond is deficient. Higher dopant concentration contributes to a better carrier concentration. However, the carriers' mobility reduces significantly due to the additional impurity scattering in the diamond lattice. A potential strategy to alleviate this issue in bulk materials is delta doping. Delta doping is a method where a thin layer of boron atoms is introduced into the diamond lattice, forming a two-dimensional layer of dopants at a thickness of a few atomic layers. The confinement of carriers in a thin layer causes them to move away from the regions with ionised impurities, called delocalisation. It has been proposed that this delocalisation can reduce the impurity scattering in the highly doped diamond, thus exhibiting both high concentration and high mobility at room temperature [39]. The high precision requirement of the boron-doped monolayer in synthetic diamonds thus far has limited the development of this method. with little improvement in p-type diamond observed experimentally thus far using these delta doping techniques.

### 2.4 Surface transfer doping of diamond

The free dangling bond of carbon on the diamond surface can be terminated by various elements such as oxygen and hydrogen, which dramatically changes the surface electronic properties and allows for an alternative doping process known as "surface transfer doping".

Electron affinity ( $\chi$ ) is referred to as the energy required to excite electrons from the bottom of the conduction ( $E_{CBM}$ ) band to the vacuum level ( $E_{vac}$ ). Moreover, ionisation potential (I) can be calculated by the addition of band gap ( $E_g$ ) and electron affinity because it is defined as the energy required to excite electrons from the top of the valence ( $E_{VBM}$ ) band to the vacuum level. As an intrinsic diamond with no surface termination, the electron affinity and ionisation potential are 0.4 eV and 5.87 eV, respectively.



Figure 2.8: The atomic structure of diamond surface with different terminations.
(a) Diamond surface without termination, carbon atoms form double-bonded dimers. (b)
Hydrogen terminated diamond surface forms (2 × 1) : H reconstruction. (c) Oxygen terminated diamond surface form (1 × 1) : O reconstruction in the form of "ether" or "ketone".

When the surface of the diamond is terminated with oxygen, the surface carbon atoms reconstruct with oxygen atoms to  $(1 \times 1)$ : O structure where the oxygen either double bonded with one carbon atom to form "ketone" or single bond to form ether structure [40]. Due to the higher electronegativity of oxygen than that of carbon atoms, the electron affinity of the diamond surface increases to 1.7 *eV*, which also brings the ionisation potential of the diamond to 7.2 *eV*.





(b)



Figure 2.9: Energy band diagram of diamond with different surface terminations. (a) Clean surface (b) Hydrogen termination (c) Oxygen termination. Due to the electronegativity between the carbon atoms and hydrogen/oxygen atoms, the electron affinity of diamond surface with different termination decreased (with hydrogen) or increased (with oxygen). However, with hydrogen as the surface termination, the surface reconstructs to  $(2 \times 1)$ : H structure [41]. Hydrogen atoms form a single bond with carbon atoms. The electronegativity difference between carbon and hydrogen atoms creates a dipole on the diamond surface. Usually, a semiconductor's vacuum level situates above the conduction band minimum to stop low-energy electrons from escaping to the vacuum. But the dipole on the hydrogen-terminated diamond surface drags the vacuum level below the conduction band minimum by 1.3 eV and brings ionisation potential to 4.2 eV [42]. This property is called negative electron affinity (NEA).



Figure 2.10: Using air adsorbates as an example, energy band diagram for surface transfer doping on a hydrogen-terminated diamond [43].

The relatively small ionisation potential induced by NEA provides a small barrier for the diamond to transfer electrons from the valence band to an energetically favourable electron acceptor on the diamond surface. Such a material should have the lowest unoccupied molecular orbital (LUMO) or single occupied molecular orbital (SOMO) close to or below the valence band maximum of the diamond [44]. In other words, materials with electron affinity greater than hydrogen-terminated diamond ionisation potential (4.2 *eV*) are generally required. It is reported that air-exposed hydrogen-terminated diamond surface creates a series of chemical groups, such as  $H_3O^+$  and  $HCO_3^-$ , to behave as an electron acceptor on the diamond surface [43]. When those air adsorbates come into contact with the hydrogen-terminated diamond surface, excess electrons in the diamond valence band transfer to the lowest unoccupied molecular orbital in those chemical groups and pull up and bend the energy band to create a holes accumulation layer near the surface. Except for the air adsorbates, other materials like  $MoO_3$ ,  $V_2O_5$ ,  $ReO_3$ , and  $WO_3$  also act as surface acceptors to transfer excess electrons on the diamond surface [45]–[50].

### 2.5 Metal semiconductor interfaces

Due to the work function difference, when metal and semiconductor materials are brought together, electrons transport contributes to fermi level form new thermal equilibrium and different types of contact [30]. This section introduces the two most common metal-semiconductor contact: ohmic contact and Schottky contact. In addition, metal oxide semiconductor (MOS) contact will be introduced to understand field effect transistors operation.

#### 2.5.1 Schottky Contact

Work function ( $\phi$ ) is the energy difference between the Fermi level within a material and the external vacuum level. If a metal with a work function ( $\phi_m$ ) is smaller than the work function of a p-type semiconductor ( $\phi_s$ ), when they are in contact, electrons in the metal flow to the semiconductor where the Fermi level is lower than that of metal, which also neutralises the majority carriers, holes, in the semiconductor and creates a depletion layer near the metal-semiconductor interface. The depletion layer has a barrier that stops electrons from flowing from metals to semiconductors.

Figure 2.11 shows the band diagram of Schottky contact made with a p-type semiconductor before and after equilibrium.  $\phi_{Bp}$  is defined as the ideal barrier height from the metal side. Equation 2.2 given the expression of this barrier height, where  $E_g$  is the bandgap energy.

$$q\phi_{Bp} = E_g - (q\phi_m - q\chi) \tag{2.2}$$

From the side of the semiconductor, the height of the built in barrier can be expressed as equation 2.3.

$$V_b = \phi_s - \phi_m \tag{2.3}$$



Figure 2.11: Energy band diagram of P-type Schottky contact. (a) The diagram before contact. (b) Metal and semiconductor making contact and achieving new thermal equilibrium.

Due to the barrier in the metal and semiconductor interface, Schottky contact has rectifying properties. That means that when there is a forward bias applied to the contact, current can flow through more effortlessly than a reverse bias. In the case of metal and p-type semiconductors, the forward bias is when a negative bias is applied to the metal. Figure 2.12 shows the band diagram of p-type Schottky contact with forward bias. The forward bias (negative) attracts the majority carriers (holes) in the semiconductor close to the interface and pulls down the barrier, which makes it easier for electrons to travel from the metal to the semiconductor.



Figure 2.12: P-type Schottky contact with forward bias.



Figure 2.13: P-type Schottky contact with reverse bias.

If the bias on metal changes to positive, the contact works in the reverse mode. The positive bias repels holes in semiconductor away and widens the depletion region. The increased barrier makes it more difficult for electrons travel from the metal to the semiconductor.

This rectifying behaviour of Schottky contact can be used as a gate terminal for a MESFET. At forward bias, devices are turned on and off at reverse bias. Though the reverse bias can turn off the contact, there is still some current flow through the interface. There are three ways electrons can overcome the barrier, field emission, thermionic emission and thermionic field emission [30]. Field emission happens when the depletion layer is too thin, and quantum mechanical tunnelling may occur. Field emission depends on the carrier concentration and width of the depletion layer. Thermionic emission usually happens when the electrons acquires enough thermal energy, increasing exponentially as the temperature increases. The last one, thermionic field emission, occurs when electrons obtain enough thermal energy to tunnel the barrier above the Fermi level.

#### 2.5.2 Ohmic Contact

Different from the Schottky contact, Ohmic contact is a non-rectifying contact. That means when metal and semiconductors make a contact, there is a negligible barrier in the interface, which allows current to pass through the interface with a small voltage drop compared to the drop across the whole device.[30] For a p-type semiconductor, ohmic contact happens when  $\phi_m$  is greater than  $\phi_s$ .

Figure 2.14 shows the band diagram of a p-type semiconductor before and after forming an ohmic contact. Because the work function of the metal is greater than that of the semiconductor, electrons in the semiconductor flow into the metal where there is a lower energy band, which leaves holes in the semiconductor and forms a holes accumulation layer in the interface, which makes it easier to transport electrons from the metal to the semiconductor. That is how ohmic contact forms on a p-type semiconductor, creating very small barrier for current travel.



Figure 2.14: Band diagram of p-type semiconductor ohmic contact. (a) before contact (b) after contact.

#### 2.5.3 MOS Contact

Metal-Oxide-Semiconductor contacts plays a crucial role in MOSFET operation, which will be talked about in the next section. Figure 2.15 shows an ideal MOS structure.  $t_{OX}$  indicates the thickness of the oxide layer, and  $\varepsilon_{OX}$  is the dielectric constant of the oxide.



Figure 2.15: Ideal MOS Structure.

An ideal MOS contact can be treated as a parallel capacitor with the metal and semiconductor as upper and lower plates. If a p-type semiconductor is used, when a negative bias is applied on the metal side, an electric field point to the metal plate is created. When the electric field crosses through the semiconductor, the majority carriers (holes) in the semiconductor are pulled to the oxide-semiconductor interface. In the band diagram, a hole's accumulation layer is created near the oxide-semiconductor interface.

However, suppose a positive bias is applied to the metal gate. In that case, it will produce an electric field in the opposite direction to the electric field generated when a negative bias voltage is applied to the gate. That electric field repels holes away from the oxide-semiconductor interface and leaves ionised charges to form space charge region. When the applied bias increases, more negative charges are left near the oxide-semiconductor interface. The Fermi level is closer to the conduction band than valence, which means the semiconductor near the oxide-semiconductor region has been turned into n-type semiconductor.



Figure 2.16: Energy band diagram of MOS contact. (a) Ideal MOS energy band diagram when the semiconductor and metal work function is the same. (b) When a negative bias is applied to the metal gate.



Figure 2.17: Energy band diagram of MOS contact.

(a) When there is a positive bias applied to the metal gate, space charge region formed (b) Keep increasing positive bias, a inversion layer forms near the oxide-semiconductor interface.

A MOS contact is used in FET devices to adjust the channel depth by changing the bias applied on the metal gate terminal to turn on and off the device.
### 2.6 Field Effect Transistor operation

A field effect transistor (FET) is a device that uses an electric field to control the current flow through a semiconductor material. In an FET, a voltage applied to the gate creates an electric field in the semiconductor, which controls the current flow between the source and drain terminals. There are several types of FETs, including junction field transistors (JEFTs), metal-oxide-semiconductor transistors (MOSFETs), and metal-semiconductor transistors (MESFETs). In this work, MOSFETs and MESFETs will be reviewed in detail.

#### **2.6.1 MOSFET**



Figure 2.18: p-channel MOSFET cross section. Enhancement mode (left) and depletion mode (right).

A MOSFET contains three terminals source (S), gate (G), and drain (D). Depending on the channel types, MOSFETs are divided into n-channel (NMOS) and p-channel (PMOS). MOSFET is divided into enhancement mode and depletion mode depending on whether there is a channel under the gate terminal when no voltage is applied. For enhancement mode p-channel MOSFET (PMOS-enhancement), there is no channel between two highly doped p regions, which means the device is off when there is no voltage applied to gate terminal. A negative voltage applied to the gate ( $V_G$ ) repels electrons doped in the substrate. When the bias is high enough, threshold voltage ( $V_{th}$ ), an inversion layer of charge, a layer where the original minimum carries turns to the majority, forms under the gate terminal and performs as a channel to allow carriers to travel through the source and drain. However, in a depletion mode p-channel MOSFET (PMOS-depletion), a channel exists between the source and drain without bias on the gate terminal-a positive bias is required on the gate to repel holes in the channel and turn off the device.

When the MOSFET is in the on-state ( $V_G > V_{th}$ ), a bias applied on the drain terminal can push the carriers to migrate between source and drain. Traditionally, the majority carriers are expected to travel from source to drain. With increasing the bias applied on the drain, FETs work under two regions. The first region is when a small bias is applied between the source and drain ( $V_{GS}$ ). This



Figure 2.19: p-channel MOSFET I<sub>DS</sub> - V<sub>DS</sub> curve.

region is linear or ohmic region where drain current  $(I_D)$  increases proportionally to the increase of drain voltage because the channel behaves as a resistor. Equation 2.4 expresses the current in the linear region, and W and L are the width and length of the gate terminal, respectively.  $C_{ox}$  is the capacitor between the gate metal and semiconductor, and  $\mu_p$  is the mobility of holes in the semiconductor.

$$I_D = \frac{W\mu_p C_{ox}}{2L} [2(V_{SG} + V_{th})V_{SD} - V_{SD}^2]$$
(2.4)

With drain bias increasing, at some point, the bias reaches saturation voltage ( $V_{sat}$ ), where the thickness of inversion layer near drain terminal reduces to zero; that is called pinch-off point.

$$V_{SD(sat)} = V_{SG} + V_{th} \tag{2.5}$$

Beyond this point, the voltage keeps constant with drain bias increasing. That means the amount of carriers transported from the source is constant, and the current no longer changes with the drain voltage. This region is the saturation region.

$$I_{D(sat)} = \frac{W\mu_{p}C_{ox}}{2L}(V_{SG} + V_{th}^{2})$$
(2.6)

Figure 2.19 shows the  $I_D$  and  $V_D$  curve under the ideal situation. The slope on the graph indicates the on-resistance ( $R_{on}$ ) of the FET device. This resistance comprises a series of resistance that includes the contact resistance ( $R_c$ ), channel resistance ( $R_a$ ) and under gate resistance ( $R_{undergate}$ ).

$$R_{on} = 2R_c + 2R_a + R_{undergate} \tag{2.7}$$

#### **2.6.2 MESFET**

Instead of using oxide to create a capacitor as the gate terminal, MESFET is a device fabricated with Schottky contact as the gate terminal. As mentioned in the previous section, due to the work function difference, the metal-semiconductor contact creates a depletion layer when the electrons in the metal diffuse into the semiconductor. Depends on if the depletion area can totally break the channel, MESFET can be divided into depletion mode, which is normally on at zero bias and enhancement mode, which is normally off at zero bias. The barrier thickness in the interface changes with the applied bias on the metal terminal. For a p-channel MESFET, a positive bias increases the barrier and allows less current to get through. For a depletion mode MESFET, the device can be turned off if the bias is big enough.



Figure 2.20: p-channel depletion mode MESFET with no bias and positive bias.

For a p-channel depletion mode MESFET, the working mode is similar to p-type MOSFET. When the bias applied on drain terminal is small, the device works on linear region. However, with the bias applied on the drain increase, the Schottky barrier on the drain side is more reversed-biased. When the bias is high enough, the depletion region on the drain side reaches the insulating substrate, and that is called pinch-off. Beyond this point, the current reaches the saturation current  $I_{D(sat)}$  and no longer changes.

$$I_{D(sat)} = \frac{\mu_n \varepsilon_s W}{2L} (V_{GS} - V_T)^2$$
(2.8)

# 2.7 Chapter Summary

This chapter briefly describes synthetic diamond and the advantages of using diamond as electronic device material over other wide band gap materials. Challenges of doping diamond have been introduced, and an institutional way, surface transfer doping, of applying diamond as an electronic devices is reviewed. In addition, fundamental theories of FET devices are discussed.

# Chapter 3

# **Literature Review**

Continuing from the previous chapter, this section delves deeper into surface transfer doping of diamond, providing in-depth insights and highlighting recent advancements in this field. Sub-sequently, it explores methods for hydrogen termination of diamond surfaces, introducing an alternative technique known as Spillover, which was employed in this study to hydrogenate the diamond surface. Following this, there is a review of contact metals on hydrogen-terminated diamond surfaces, aiming to elucidate the formation of ohmic contacts on diamond. Lastly, the chapter concludes with an exploration of the cutting-edge technology in the realm of diamond surface transfer doping Field-Effect Transistors (FETs).

### **3.1** Surface transfer doping

In the previous chapter, the challenges associated with utilizing boron or phosphorus doped diamond at room temperature was discussed, as well as alternative strategies like delta doping and surface transfer doping. However, due to the stringent precision demands of boron delta doping and the limited enhancements observed in p-type delta doping diamond, the practical applicability of this method is constrained. On the other hand, surface transfer doping shows greater promise for the fabrication of high performance diamond-based devices.

10The first experimentally observed surface transfer doping was reported in 1989 when Landstrss et al. measured a fairly low resistivity of  $10^6 \ \Omega \cdot cm$  on type IIa single crystal diamond without any impurity doping [51]. This value is much lower than that of diamond without hydrogen termination  $(10^{16} \ \Omega \cdot cm)$  [29]. This phenomenon was initially explained by the diffusion of hydrogen into the diamond to contribute as an acceptor state [52]. However, in 1990, Grot et al.[53] proposed that the hydrogen induced surface conductivity can be removed by chemical cleaning process which can only remove non-carbon contamination on the diamond surface.

This discovery contradicts the previously mentioned theory about the surface conductivity of hydrogen terminated diamond. Soon after this discovery, Junji et al.[54] reported that the Kelvin Probe measurement and X-Ray Photoelectron Spectroscopy shows evidence that electrons on hydrogen terminated diamond surface transferred into unknown acceptor states on the surface. It provided a strong evidence for a hole accumulation layer sitting below the hydrogen terminated diamond surface.

As discussed in the previous chapter that hydrogen terminated diamond surface exhibits negative electron affinity (NEA), and this property can be used to verify if the surface termination on diamond has been changed. Negative electron affinity on hydrogen terminated diamond was first observed by Himpsel et al.[55] in 1979 with observation of excited electrons when the photons energy is lower than that of theoretical prediction.

In 2000, with the help of NEA, Maier et al.[43] proposed a series of experiments verifying a few crucial facts about the surface conductivity of hydrogen terminated diamond. In their experiment, a hydrogen terminated diamond is kept in an ultra-high vacuum (UHV) system and experienced high temperature annealing. With in-situ measurement, the surface conductivity is observed to drop after annealing, while the measured negative electron affinity (NEA) confirms the existence of hydrogen termination on the diamond surface. Following the experiment, half of the sample surface is protected, with the other half exposed to electron beams aiming at removing hydrogen termination. The measurement of electron affinity confirms that the protected surface remains the termination; however, the half exposed to electron beams loses the termination. After exposing the sample to air for three days, the protected half regains the surface conductivity, while the part exposed to electron beams remains highly resistive. They indicate that the surface conductivity of hydrogen terminated diamond not only comes from the hydrogen termination, but also caused by the air adsorbates acting as acceptor. Because the lowest unoccupied molecular orbitals of the air adsorbates are below the valence band top of the hydrogen terminated diamond, electrons transfer from the diamond to the surface and show surface conductivity, and this process is governed by the following redox reaction,

$$2H_3O^+ + 2e^- \rightleftharpoons H_2 + 2H_2O \tag{3.1}$$

Later experiment carried by other groups proposed that oxygen-ion are responsible for the surface conductivity of hydrogen terminated diamond [56]–[58]. Back to 1999, Gi et al.[59] reported that the surface resistivity of hydrogen terminated diamond decreases when the sample is exposed to  $NO_2$  flow. Following this report, many groups proposed  $NO_2$  induced surface conductivity on the hydrogen terminated diamond surface. Until 2013, the lowest  $NO_2$  induced sheet resistance of hydrogen terminated diamond, 719  $\Omega/\Box$ , was reported by Sato et al.[10].



Figure 3.1: Energy levels of various molecular orbitals and H-diamond with different surface orientation [44].

In 2013, Takagi et al.[60] conducted a detailed investigation on the transfer capability of various of airbone species on H-diamond based on experimental results and first-principles calculations. Their report indicates that the airbone species such as  $NO_2$ ,  $SO_2$ , NO and  $O_3$ , with lowest unoccupied molecular orbitals (LUMO) below or near the valence band top (VBT) contributes to the surface conductivity. However, airbone species such as  $CO_2$ ,  $N_2O$  and  $H_2O$ , with LUMO higher than the VBT of diamond has little or no effect on surface conductivity. This was later verified by Kubovic et al.[56], who reported that samples exposed to  $O_3$  lead to the largest increase of carrier concentration and exhibited good stability.

With these early experiments, the mechanism of high surface conductivity on H-diamond becomes clearer. Airborne adsorbates play crucial roles in this process, and molecules with LUMO near the VBT of H-diamond act as electron acceptors that can improve surface conductivity. Additionally, metal oxides are explored to improve both the stability and conductivity of Hdiamond. In 2013, Russell et al.[61] reported a boost in surface carrier concentration with  $MoO_3$ deposition on H-diamond. In 2014, Tordjman et al. [49] reported the same  $MoO_3$ -induced effect on H-diamond, with carrier densities up to  $1 \times 10^{14} \text{ cm}^{-2}$ . Both experiments indicate that  $MoO_3$ acts as an electron acceptor, with the conductance band minimum situated below the valence band maximum of H-diamond. The migration of electrons from H-diamond to  $MoO_3$  drives the energy band of H-diamond upward, forming a hole accumulation layer near the surface of H-diamond.



Figure 3.2: Band diagram of MoO<sub>3</sub>/H-diamond before and after thermal equilibrium [49].

This work demonstrates the potential of applying metal oxides with high electron affinity (EA) to act as acceptors and improve the surface conductivity of H-diamond. Transition metal oxides such as  $ReO_3$ ,  $WO_3$ ,  $V_2O_5$ ,  $Nb_2O_5$ , and  $CrO_3$  have been reported following the success of  $MoO_3$  [45], [46], [48], [50]. In Verona et al.'s report, the sheet resistance of samples covered with transition metal oxide is only around one-tenth of that of samples exposed to air. Additionally, the carrier concentration of samples covered with transition metal oxide is ten times that of air-exposed samples. High-temperature stability is also reported up to 300 °C with  $V_2O_5$  and  $MoO_3$  coverage [45], and up to 400 °C for  $ReO_3$  covered samples [48].



•  $MoO_3 V_2O_5 WO_3 Nb_2O_5 ReO_3 Al_2O_3$ 

Figure 3.3: Sheet Resistance of H-diamond with and without transition metal as surface acceptor[62].

Transition metal oxides have been reported to improve the surface conductivity and thermal stability, making them ideal materials for encapsulating H-diamond devices. However, another group of materials, high dielectric constant materials, has drawn attention from researchers. Hiraiwa et al. [63] reported that the use of  $Al_2O_3$  can encapsulate and stabilize the surface conductivity of H-diamond surfaces. In 2016, Verona et al. [50] compared the thermal stability of H-diamond with or without oxide protection. Their results show that the surface conductivity of air-exposed H-diamond drops drastically between 25 °C and 300 °C, whereas the surface conductivity of H-diamond with high work function oxide protection or  $Al_2O_3$  protection can remain stable up to 250 °C and 400 °C, respectively. Kawarada et al. [64] also reported that the surface stability of H-diamond depends on the deposition temperature of  $Al_2O_3$ , and they reported a device that remains stable up to 400 °C with  $Al_2O_3$  deposited under 450 °C.

Figure 3.4 illustrates the electron affinities of various materials compared to the band diagram of diamond with different terminations. As previously discussed, the conduction band minimum (CBM) of all transition metal oxides is situated below the valence band maximum (VBM) of hydrogen-terminated diamond, allowing them to act as electron acceptors. However, high dielectric materials, such as  $Al_2O_3$ , have their CBM situated above the VBM of H-diamond, indicating that  $Al_2O_3$  should theoretically be unable to contribute to the surface conductivity of H-diamond. Nonetheless,  $Al_2O_3$  has been found to improve the surface conductivity of H-diamond, but only when deposited by Atomic Layer Deposition (ALD).



Figure 3.4: Electron affinities of different metal oxides compared to the bandgap of diamond with no termination, oxygen termination and hydrogen termination[62].

In 2012, Kawarada et al. [65] demonstrated that H-diamond with  $Al_2O_3$  deposited by ALD on the surface does not require air exposure to achieve surface conductivity. Soon after, they proposed that unoccupied energy levels may exist in  $Al_2O_3$ , situated below the VBM of H-diamond [66]. Later, using XPS, Ren et al. [67] observed a large band offset of 3.3 eV between the H-diamond and  $Al_2O_3$  interface. They also suggested the presence of interstitial aluminium and oxygen vacancy defects, which act as electron acceptors. Currently, all reports of  $Al_2O_3$ improved conductivity on H-diamond surfaces have been accomplished by atomic layer deposition (ALD) technique. In contrast, Verona et al. [50] utilized electron beam deposition of  $Al_2O_3$ , however, the sheet resistance of H-diamond increased to 38.6  $k\Omega/\Box$ , which is more than two times the sheet resistance before deposition.

In conclusion, the mechanism behind the conductivity of H-diamond has been investigated over the past several years. The most convincing explanation is that air adsorbates on the surface act as electron acceptors and create a hole accumulation layer on the diamond surface. Subsequent studies have also explored other materials, such as high electron affinity materials and high dielectric constant materials, to substitute air adsorbates in order to improve conductivity as well as stability.

### **3.2** Diamond surface hydrogen termination

Diamond surfaces can be hydrogen-terminated using various methods. As discussed in the previous chapter, diamond can be grown using the chemical vapor deposition (CVD) method, where hydrogen and hydrocarbons serve as precursors and act as the source of hydrogen termination. Nowadays, the most popular and optimized method for diamond growth and in-situ hydrogen termination is microwave plasma-enhanced chemical vapor deposition (PCVD) [68].

Ex-situ plasma-assisted hydrogen termination methods are widely used in research areas. In this method, oxygen-terminated diamond is exposed to hydrogen plasma in a vacuum chamber with temperatures ranging between 500 °C and 800 °C [69], [70]. Although this method can complete hydrogen termination in a few minutes, the roughness of the diamond surfaces varies depending on factors such as the power of the plasma, chamber temperature, time, and other parameters. In Crawford's work [71], atomic force microscopy (AFM) measurements showed that the surface roughness of diamond samples increased after the hydrogen termination process, with roughness worsening at higher plasma powers. In 2011, Philip et al. [70] observed that longer durations of hydrogen plasma smoothed the diamond surface. They did not observe hydrogen plasma etching pits reported in Crawford et al.'s work with this prolonged plasma treatment. However, this effect was only observed after 2 hours of treatment. With properly set parameters, one can modify the diamond surface to achieve a smooth hydrogen-terminated surface using ex-situ plasma-assisted methods [72].

Thermal hydrogenation is another method for hydrogenating diamond surfaces. In 1993, Ando et al. [73] achieved hydrogenation on oxygen-terminated diamond surfaces. The experiment was conducted under atmospheric pressure and high temperatures (below 900 °C), with pure hydrogen flowing through the chamber for several hours. A similar experiment was carried out in 2013 by Seshan et al. [74], where hydrogen-terminated diamond was obtained with pure hydrogen flow at 800 °C. These experiments reveal the possibility of thermally hydrogenating diamond. Earlier than these two reports, Pate et al. hydrogenated diamond surfaces with hydrogen atoms rather than molecular atoms [75]. In their experiment, a heated tungsten filament (1800 °C) was used to generate atomic hydrogen and terminate the diamond surface. High-resolution electron energy loss spectroscopy (HREELS) and photo-stimulated ion desorption (PSID) results revealed a diamond surface with hydrogen termination after exposure to atomic hydrogen under ultra-high vacuum.

Diamond surface hydrogenation has been well studied in the past several years. Ex-situ plasmaassisted hydrogenation methods contribute to achieving a good hydrogen-terminated diamond surface. In addition, thermal hydrogenation with molecular or atomic hydrogen also shows good potential for terminating diamond surfaces with hydrogen.

## 3.3 Spillover mechanics

Spillover refers to a process where  $H_2$  molecules are dissociated by a metal catalyst, such as Pt and Pd, into H atoms, and these H atoms migrate on the contact support materials [76]. The essence of this process lies in the interaction between active hydrogen atoms and the support materials, wherein hydrogen migrates from the hydrogen-rich phase to a hydrogen-poor phase.





Catalytic metal (Pd or Pt) dissociation of the hydrogen molecule into atoms those migrate to the non-metallic support surface (diamond). [77].

In 1964, Khoobiar reported that yellow  $WO_3$  is reduced by  $H_2$  to blue  $WO_{3-x}$  when Pt is used as a catalyst [78]. This phenomenon was explained as the dissociative chemisorption of  $H_2$ molecules on the Pt surface, followed by migration of the resulting H atoms to the yellow  $WO_3$ particles, leading to their reduction to blue  $WO_{3-x}$ . Soon after, Boudart et al. coined the term "spillover" to describe this process, wherein dissociated H atoms diffuse from the metal catalyst to support materials such as carbon-based materials and  $TiO_2$  [79]. Recent research on spillover has shown that the proton-electron pairs formed by H atoms can migrate onto support materials such as  $MoO_3$  [80], [81].

In 2010, Triwahyono et al. conducted a study on the kinetics of hydrogen adsorption over  $Pt/MoO_3$ . Their results showed that H atoms are formed when  $H_2$  passes through the catalyst metal, such as Pt, rather than on the surface of the support material ( $MoO_3$ ). Subsequently, the electrons reduce the reducible cations of  $MoO_3$ , while the protons bind to the surface oxygen anions [82].

Hydrogen spillover onto reducible materials such as  $TiO_2$ ,  $WO_3$ , and  $CeO_2$  has been well studied and confirmed [83]–[85]. In 2017, Karim et al. conducted a series of experiments and verified that spillover occurs on reducible support materials such as  $TiO_2$  irrespective of the distance. However, on non-reducible materials such as  $Al_2O_3$ , this effect is barely detectable when the distance between the catalyst metal and the oxide is increased [86].



Figure 3.6: Hydrogen spillover on different support material.

Left, diagram of reduction grade on reduction support material  $TiO_2$ . Right, diagram of reduction grade on non-reduction support materials  $Al_2O_3$ . The green dots is the catalyst metal *Pt* in both images [86].

The spillover phenomenon on non-reducible support materials remains controversial. Despite reports in recent years about hydrogen spillover on non-reducible support materials such as  $Al_2O_3$  and  $SiO_2$ , the process is energetically unfavorable [87], [88]. In order for hydrogen spillover to occur onto the support material, a new bond between hydrogen atoms and support materials with energy similar to that of the metal-hydrogen bond is required. However, non-reducible materials cannot form this bond with hydrogen atoms [76]. The observed hydrogen spillover on non-reducible materials has also been explained as being due to defects on the material surface contributing to the breakdown of the metal-hydrogen bond [89].

Spillover on carbon-based materials is also energetically unfavorable. The calculated C - H binding energy is approximately 0.8 eV, which is smaller than that required to break the metal-hydrogen binding, such as the 3.8 eV bond energy in the case of Pt - H binding [90]. This means that hydrogen atoms on the metal surface cannot migrate to the surface of carbon-based materials. However, this only applies to carbon-based surfaces without defects. If defects exist on the material surface, one can expect hydrogen atoms to migrate from the metal surface to unsaturated carbon-based materials to form C - H bonds. In 2003, Mitchell et al. reported that C - H bonds form on carbon edge sides where free dangling bonds are observed before the spillover process [91]. This supports the theory that spillover can occur on carbon-based surfaces with free dangling bonds.

When it comes to the hydrogenation of diamond surfaces, the surface can first be oxidized with an acid cleaning process, which will be discussed in the next chapter. Then, defects, or free dangling carbon bonds, can be created by heating the oxygen-terminated diamond under vacuum. The temperature required to desorb the oxygen on the (100) diamond surface is reported to be 600 °C [92]. With free dangling carbon bonds exposed, hydrogen atoms dissociated by transition metals such as Pd and Pt are able to migrate to the diamond surface and form C - H bonds, thus hydrogenating the diamond surface.

Hydrogen spillover has been detected on both reducible materials and non-reducible materials with defects on the surface. To make the process energetically possible for carbon-based materials, free dangling carbon bonds are required for the migration of hydrogen atoms dissociated by catalyst metals. These findings provide theoretical support for hydrogen-terminated diamond surfaces via thermal spillover processes.

# **3.4** Contact metal on hydrogen terminated diamonds

In the previous chapter, we introduced the metal-semiconductor interface. In this section, we will delve into the metal-hydrogen-terminated diamond contact to review the metal contacts used in device fabrication.

As discussed previously, to form an ohmic contact on a p-type semiconductor, a metal with a work function  $(\phi_m)$  greater than that of the semiconductor  $(\phi_s)$  is desirable. This allows electrons from the semiconductor to flow into the metal, bending the semiconductor's band and forming an accumulation area at the interface to facilitate easy current flow. Au has been the most commonly used ohmic metal for hydrogen-terminated diamond due to its high work function. Reported lowest contact resistance formed by Au on H-diamond is 0.9  $\Omega \cdot mm$  [93]. Contact resistance reported by other groups varies from 3  $\Omega \cdot mm$  to 5  $\Omega \cdot mm$  [94]–[96]. Additionally, metals with high work functions like Pd have also been explored for use as ohmic contact metals, with reported contact resistance as low as 0.42  $\Omega \cdot mm$  [97].

To form Schottky contacts on hydrogen-terminated diamond, a metal with a lower work function is used. This allows electrons in the metal to migrate into the hydrogen-terminated diamond, forming a depletion area and creating an energy barrier to hinder electron transfer when the junction is reverse biased. Metals like Al, Pb, Zn, and Ti have been reported to form large Schottky barriers on hydrogen-terminated diamond surfaces [98]. The extrapolated Schottky barrier heights vary from 0.58 eV to 0.39 eV.

# 3.5 Field Effect transistors

As discussed in the previous chapter, hydrogen-terminated diamond (H-diamond) exhibits great potential for applications such as high-power, high-frequency, and high-temperature metal-oxide-semiconductor field-effect transistors (MOSFETs). Despite its limitations in thermal sta-

bility, appropriate passive layers and surface carrier acceptors can significantly improve both the thermal stability and conductivity.

The first hydrogen-terminated diamond FETs were demonstrated by Kawarada et al. in 1994 [99]. In their experiment, a chemical vapor deposition (CVD) grown hydrogen-terminated (100) diamond was used as the substrate. Low work function Al and high work function Au were utilized to form Schottky and ohmic contacts, respectively. With a gate length of 10  $\mu m$ , this device achieved a highest drain current of 1.6 mA/mm and a highest transconductance of 200  $\mu S/mm$ . To fully turn off the device, the gate length was increased to 40  $\mu m$ . However, this resulted in a drop in drain current to 0.14 mA/mm and transconductance to 20  $\mu S/mm$ . Following this pioneering device, subsequent devices with adapted geometry and improved surface conditions were fabricated.



Figure 3.7: Structures of first H-diamond MESFETs[99].

Russell et al. demonstrated a metal-semiconductor field-effect transistor (MESFET) with Al as the Schottky contact in 2012 [96], with the gate length reduced to 50 nm. They achieved a maximum drain current exceeding 295 mA/mm with a transconductance of 78 mS/mm. However, the measured leakage current reached up to 0.02 mA/mm.

In contrast to using metals with low work function for Schottky gate contacts, devices with gate dielectric layers insulators were fabricated to create a metal-insulator-H-diamond interface, aiming to improve breakdown voltage potential, reduce leakage current, and enhance surface termination stability. Transition metal oxides have been utilized to improve the surface conductivity of hydrogen-terminated diamond by acting as electron acceptors. In 2016, Verona et al. reported metal-insulator-semiconductor field-effect transistors (MISFETs) on hydrogen-terminated diamond with  $V_2O_5$  as the gate dielectric [100]. They observed stable surface conductivity of hydrogen-terminated diamond up to 200 °C, and at room temperature, the surface conductivity remained stable after 30 days of air exposure. Additionally, the maximum drain

current extracted from MISFETs with a gate length of 2  $\mu m$  was 280 mA/mm, which was 80 mA/mm higher than the MESFETs with the same gate length. However, the gate leakage current of 40  $\mu A$  was even higher than that of the previously discussed MESFETs [96].

In 2017, Ren et al. demonstrated the first hydrogen-terminated MOSFETs with  $MoO_3$  as the insulator layer [101]. The maximum drain current for a gate length of 4  $\mu m$  was 33 mA/mm. However, the breakdown voltage (roughly 2 V) was relatively low and might restrict the maximum current. Moreover, the leakage current of the device with 10 nm of  $MoO_3$  was similar to that of the same devices with 3 nm of  $Al_2O_3$ , indicating that a higher thickness of transition metal oxide is required to restrict the drain leakage.

Crawford et al. reported the highest drain current of 700 mA/mm for FETs with a 50 nm gate fabricated on hydrogen-terminated diamond [93]. Instead of using  $V_2O_5$  as the insulator layer under the gate, the oxide was used to encapsulate the entire device. However, the device suffered from drain-induced barrier lowering and thus poor off-state performance.



Figure 3.8: Structures of Crawford demonstrated FETs[93].

In summary, the utilization of transition metal oxides as dielectric materials to enhance the surface conductivity of hydrogen-terminated diamond and improve device properties requires further investigation. Reported metal-insulator-semiconductor field-effect transistors (MISFETs) with transition metal oxides as the insulator layer exhibit limited improvement and suffer from similar leakage issues as metal-semiconductor field-effect transistors (MESFETs) devices.

Another group of oxides with high dielectric constants, such as  $Al_2O_3$ ,  $SiO_2$ ,  $TiO_2$ , and  $HfO_2$ , have been explored as gate dielectric materials. Hydrogen-terminated diamond metal-oxide-semiconductor field-effect transistors (MOSFETs) with  $SiO_2$  as the dielectric material were first demonstrated in 1999 [102]. However, the maximum drain current achieved with this device was only 18 mA/mm, significantly lower than that reported for hydrogen-terminated MESFETs

and MISFETs with transition metal oxides as the dielectric.

In 2013, Liu et al. demonstrated normally-off diamond MOSFETs with  $HfO_2$  as the gate dielectric, showcasing low leakage drain current, indicating the potential of using high dielectric materials to mitigate drain leakage issues. The maximum drain current of 37.6 mA/mm also represented an improvement compared to the first MOSFETs built with *SiO*<sub>2</sub>.

Among the mentioned materials,  $Al_2O_3$  is the most popular and widely used one as both a dielectric layer and channel passivation layer. Hirama et al. illustrated a MOSFET with alumina gate insulator in 2007 [103]. Their device featured a 3 nm layer of aluminum deposited on the gate position, followed by self-oxidation of this metal layer to form  $Al_2O_3$ . Subsequently, aluminum was deposited on the oxide to form the gate stack. The measured maximum drain current of 790 mA/mm and an  $I_{on}/I_{off}$  ratio of 10<sup>9</sup> demonstrated a drastic improvement compared with MESFETs.

In 2019, Yu et al. demonstrated a T-shaped gate with atomic layer deposition (ALD)  $Al_2O_3$  as the dielectric layer [104]. In this device, thermal-assisted ALD was utilized to grow  $Al_2O_3$  as the dielectric layer under a 90 °C to reduce the leakage current. The final DC measurements of the device showed a maximum drain current of 703 mA/mm and a gate leakage current as low as 0.6 nA/mm.



Figure 3.9: Cross section of T-shaped gate MOSFETs[104].

To mitigate off-state leakage current and enhance the breakdown voltage, various high-dielectric materials have been tested on hydrogen-terminated diamond, including multi-layer dielectric structures such as  $HfSiO_4/Al_2O_3$ ,  $ZrO_2/Al_2O_3$ ,  $LaAlO_3/Al_2O_3$ , and  $YSZ/Al_2O_3$  [105]–[108]. However, a common issue among all these reported devices is the low drain current ( $\leq 20$  mA/mm). The highest drain current achieved with passivation was reported in 2012 [109]. By exposing hydrogen-terminated diamond to  $NO_2$ , Hirama et al. demonstrated a MISFETs using  $Al_2O_3$  as passivation with a maximum drain current of 1.35 A/mm, which remains the highest recorded drain current thus far.

Due to the intrinsic properties of air-exposed hydrogen-terminated diamond, most of the demonstrated devices discussed above are "normally-on" devices. However, for applications requiring minimized off-state leakage current and improved device properties under high-power usage, "normally-off" hydrogen-terminated diamond devices are more desirable. High-dielectric materials are effectively used to create a larger energy barrier to reduce gate leakage and achieve "normally-off" FETs on hydrogen-terminated diamond. Nevertheless, the low drain current limits the applications of these devices. Other strategies, such as partially oxidizing the hydrogenterminated surface, have been reported to fabricate "normally-off" devices, but the maximum drain current achieved was only recorded as 21 mA/mm [110]. To achieve higher drain current and "normally-off" devices on hydrogen-terminated diamond, further exploration of additional methods is necessary.

# 3.6 Chapter Summary

The chapter provides a comprehensive literature review covering several key aspects related to diamond surface transfer doping, hydrogen termination methods, spillover mechanisms, contact metals, and the development of diamond surface transfer doping field-effect transistors (FETs).

Surface transfer doping emerges as a promising approach to enhance diamond surface conductivity by introducing hydrogen and other acceptor materials. Various methods, including ex-situ plasma-assisted hydrogenation and thermal hydrogenation, are explored for hydrogen termination of diamond surfaces, aiming to improve their electrical properties.

Spillover mechanisms, extensively studied, elucidate the migration of hydrogen atoms from metal catalysts to support materials, including diamond surfaces. This understanding is crucial for optimizing spillover processes and enhancing surface conductivity.

In terms of contact metals, investigations focus on forming ohmic contacts using metals like Au and Pd, as well as forming Schottky contacts using metals such as Al, Pb, Zn, and Ti. Understanding the properties of these contacts is essential for device fabrication and performance optimization.

The development of diamond surface transfer doping FETs represents significant progress in utilizing diamond for electronic applications. Various device configurations and dielectric materials are explored to improve device performance and stability. However, challenges persist in achieving low gate leakage current and high maximum drain current for practical applications.

Overall, the literature review provides valuable insights into the current state and future directions of research in this field, highlighting both advancements and remaining challenges.

# Chapter 4

# **Fabrication**

The scaling of electronic devices to nanometre dimensions necessitates advanced fabrication techniques, which become increasingly challenging when working on a diamond substrate with processing-sensitive hydrogen termination on the surface. This chapter introduces the fabrication processes and techniques established in this work.

The chapter commences with the preparation of diamond samples, encompassing the cleaning and surface hydrogenation. Subsequently, it introduces and compares in detail two of the most important lithography techniques: photolithography and e-beam lithography. Metal contacts on the sample surface are formed through physical vapour deposition and wet etching, both of which will be reviewed in this chapter. Additionally, two techniques, atomic layer deposition (ALD) and e-beam evaporation, employed to deposit the oxide layers, will be introduced in detail. To protect the surface, various alternative fabrication processes are explored, with the results of those alternative fabrication processes discussed. Finally, the chapter concludes with the discussion of a customised fabrication process for diamond FET devices.

### 4.1 Sample Preparation

A clean and uncontaminated material surface is essential for nanofabrication, particularly for fabrication on hydrogen-terminated diamond surfaces, as the device performance strongly depends on the surface quality and cleanliness. A rough surface with defects increases the trapped charge and scattering mechanics, thereby decreasing the carrier mobility[111], [112]. The initial step in creating a device on a diamond surface involves smoothing the surface and reducing surface defects.

 $4.5 mm \times 4.5 mm$  CVD grown (100) single crystal diamond samples were purchased from Element Six with surface roughness between 2 nm to 7 nm. To smooth the surface down to the

sub-nanometer scale, samples were sent to Technical Diamond Polishing for scaif polish. AFM is used to measure the surface roughness before and after polishing. Table 4.1 represents the AFM measurement results of 5 samples before and after polishing.

Surface	Sample 1	Sample 2	Sample 3	Sample 4	Sample 5
Roughness					
$R_q$ Before	1.32	3.72	2.4	1.98	3.05
Polish(nm)					
$R_q$ After	0.71	0.86	0.44	0.65	0.31
Polish(nm)					

Table 4.1: Samples surface roughness  $R_q$  before and after scaife polish. Data measured with AFM in 5 $\mu$ m scanning size.

Figure 4.1 shows the AFM results of the sample surface before scaife polishing. As these samples are recycled from previous experiments, the surface is in poor condition, with etch pits and scratches clearly visible.



Figure 4.1: AFM scan results of sample surface before polish.

Figure 4.2 depicts the AFM results of the sample surface after scaife polishing, revealing a reduction in surface roughness  $R_q$  from a few nanometers to sub-nanometer level. Additionally, the polish grains are observable post-polishing. It is noteworthy that a few white dots persist on the sample surface after scaife polishing. These white dots, observed with AFM, are presumed to be contamination remnants from the polishing process. Subsequently, acid cleaning and solvent cleaning are employed to terminate the surface with oxygen and eliminate the contamination.

To eliminate metal trace contaminants and accumulated hydrocarbon and oxygenated compounds from the sample surface, a two step acid cleaning process was carried out after the polish. The initial step involves immersing the diamond samples in boiled Aqua Regia (a 1 : 1 mixture of  $HNO_3$  and HCl) for 15 minutes. Subsequently, the samples were rinsed in RO water three times to remove any residual acid remaining on the surface. The second acid cleaning steps entails rinsing the samples in a boiled acid mixture of concentrated ( $\geq 95\%$ )  $H_2SO_4$  and



Figure 4.2: AFM scan results of sample surface After polish.

concentrated (65%)  $HNO_3$  in a 3 : 1 ratio. After 20 minutes in the boiled acid, the samples were rinsed in RO water three times. The second step is primarily serves for surface oxidation, which is crucial for the subsequent surface hydrogenation process.

After the acid clean, samples may still retain some acid residue on the surface. To rectify this, samples underwent a solvent clean process. A 5-minute rinse in acetone with ultrasonic water bath facilitates the removal of any grease and unwanted particles on the sample surface. Subsequently, a 5-minute rinse in IPA with ultrasonic water bath aims to eliminate any residual acetone. It is imperative to transfer the sample into IPA promptly to prevent acetone from drying on the sample surface, as this can be difficult to remove and may impact subsequent fabrication processes. Figure 4.3 displays the AFM scan results of samples after the acid clean and solvent clean process. In comparison to the scan results before cleaning, the presence of white dots on the sample surface is significantly reduced.



Figure 4.3: AFM scan results of sample surface after the cleaning process.

The samples were then packed in membrane box to prevent samples flips during the transit. They were subsequently dispatched to RMIT University in Australia for plasma hydrogen termination. In this process, oxygen-terminated diamond samples were placed in a vacuum chamber with the sample holder heated up to 800 °C. This temperature was maintained to prevent the etching of the diamond surface, which can dramatically roughen the surface and prevent the thermal desorption of hydrogen termination[92], [113]. Subsequently, hydrogen plasma with a power of 1000 *W* was directed onto the sample surface. After termination was complete, but before

#### CHAPTER 4. FABRICATION

the plasma were turned off, the holder temperature is reduced to 700 °C. Finally, the chamber temperature is gradually decreased back to the room temperature.

The sample surface may undergo modification by the hydrogen plasma, where both treatment duration and substrate temperature can influence the surface roughness[71], [114], [115]. AFM is utilised post-termination to ensure the surface remains sufficiently smooth enough for sub-sequent fabrication steps. Additionally, Hall measurements are conducted to confirm surface conductivity over time, ensuring hydrogen termination has been achieved. Figure 4.4 illustrates the surface roughness of diamond sample before and after hydrogen plasma treatment. On average, the surface roughness of more than half of samples increased by 0.3 *nm*, while a small proportion decreased by 0.1 *nm*. Overall, the surface remains in good condition to progression to the next step.

Subsequent to the roughness measurement, Hall kit measurements were conducted to assess surface conductivity. Four probes placed on four corners of the sample and made contact with the sample directly. The conductivity is measured six times over a span of ten days. This aims to characterise any changes in the conductivity over time and determine the duration required for conductivity stabilisation. Figure 4.4 illustrates a significant drop in conductivity across all samples within 2 to 3 days, followed by stabilisation or a slight increase in conductivity after day 9.

After all those aforementioned preparation, a solvent clean, as mentioned above, is conducted to remove any potential contamination induced during transit and plasma treatment. At this stage, the samples are ready to proceed to the subsequent stages.



Figure 4.4: AFM scan results and Hall kit measurement of sample surface before and after hydrogen plasma treatment.

### 4.2 Resist spinning



Figure 4.5: Resist before and after spinning.

Resist spinning is a process used to spread photoresist or e-beam resist into a uniform layer on the sample surface for subsequent exposure and development steps. Cleaned samples are mounted in the middle of a chuck, which is held by a vacuum system underneath. A specific amount of resist, typically enough liquid to cover 70% of the sample surface, is dispensed onto the sample surface using a pipette. Following the addition of the resist, the sample is spun at a high speed (thousands of revolutions per second), whereby centrifugal force spreads the resist into an uniform layer. The final thickness of the resist is highly dependent on the spinning speed; higher speed generally result in a thinner resist layers, although the thickness gradually stabilises when the spinning speed reaches a certain level[116]. Samples with a uniform resist layer are then transferred to hotplate or oven to bake the resist, evaporating the solvent from the resist. Different baking duration and temperature are applied to different resist to achieve optimal profiles after development.



Figure 4.6: Resist thickness versus spinning speed with ebeam resist as example[116].

The resist used in this thesis include Lift Off Resist (LOR), 50 K 12% and 200 K 4% e-beam resist and photoresist S1805. With customised spinning speed and baking durations, they yield resist thickness of 300 nm, 310 nm, 150 nm and 500 nm respectively.

# 4.3 Photo-lithography exposure

Photo-lithography is a process where photo-resist is employed to transfer desired patterns from a photomask to the sample surface. Photoresist typically consists of long chain polymer whose structure can be altered by UV light with specific wavelengths. In positive photoresist, exposure to UV light breaks the polymer chains into shorter ones, enabling selective removal of the exposed area by specific solvents such as TMAH and diluted Microposit. In contrast, in negative photoresist, the polymer is hardened by UV light. The machine utilised in this thesis is SUSS MicroTec Mask Aligner 6 (MA6).



Figure 4.7: Photolithography with photo-mask contact with the sample surface. The opaque area made of chrome blocked the UV light from exposing the resist.

A photomask is a partially opaque and partially transparent mask made of chrome and quartz. Desired patterns are written onto the mask, where the chrome blocks the UV light while the quartz areas allow the UV light to travel through and expose the photoresist.

After resist spinning, samples are loaded on the Mask Aligner 6 (MA6) holder and held steady by a vacuum system under the holder. The photomask is secured by vacuum between the sample and the UV light source. A customised exposure duration and contact mode is set up in the system, after which UV light is directed to expose the sample surface through the photomask.

When a design necessitates multiple times of exposure to form the final patterns, a manual alignment process is required. Markers, typically cross patterns, are employed for alignment purposes, placed at the four edges of the sample during the initial exposure. The sample holder in MA6 is adjustable in the X and Y directions and can be rotated in the  $\theta$  direction. For all subsequent exposures following the initial one, samples are adjusted in three directions to align the markers fabricated in the first exposure with those on the photomask. This step ensures that each exposure is aligned with the previous one.

#### CHAPTER 4. FABRICATION



Figure 4.8: Photolithography Mask Aligner 6 (MA6)



Figure 4.9: Marker used in photo-lithography.

The left is the marker made on the sample surface. The middle is the process of adjusting the marker on the sample according to the marker on the mask. The right one is after alignment, the cross on the mask overlaps with the marker on the sample surface.

The smallest feature size achievable via photolithography is highly determined by the wavelength of the light source. To enhance the resolution of photolithography, a light source with a shorter wavelength is required. Currently, Deep UV light, Extreme UV (EUV) light and X-ray are widely employed to improve the resolution. ASML company has achieved resolution of 5 *nm* and 3 *nm* with EUV photo-lithography. The mask aligner used in this thesis provides a minimum resolution of around 2  $\mu m$ .

# 4.4 Electron beam lithography

Electron beam lithography (EBL) is a technique used to expose electron beam sensitive resist on the sample surface to transfer desired patterns. EBL utilises an electron beam instead of UV light.

One advantage of using EBL over photolithography is that the patterns are designed using CAD software, and the files are exported as GDS file. These GDS files are then fractured using another software, Beamer, and expored as GPF files. Cjob is used to read GPF files and setup the exposure. The final step involves submitting the job on James Watt Nanofabrication Centre (JWNC) website. Subsequently, the machine can read the design files and use them directly to move the electron beams on the sample surface and expose the designed area, providing more flexibility to EBL compared to photo-lithography.

As the exposure is carried out by electron beams, in the case of insulated samples such as diamond, electrons may accumulate on the sample surface. When a sufficient amount accumulates, the electrons can deflect incident electron beams, leading to distortion of patterns and affecting accuracy. To mitigate this issue, a charge dissipation layer (CDL), typically a thin layer of metal, is added on the ebeam resist. This CDL channels all electrons to the ground. In this thesis, Au, Al and sputtered Au and Pd mix are utilised as CDL in different stages.

The second advantage of using EBL over photo-lithography is that EBL provides higher resolution. As discussed in the previous part that we can only achieve a resolution of 2  $\mu m$  with MA6. To achieve finer feature, EBL is necessary. The smallest feature size is determined by the minimum spot size the electron can produce, which is in turn determined by the wavelength of electron emitted. It is also determined by the amount of electron backscatter, which is dependent on the accelerating voltage and material being written.

The equipment utilised in this thesis is the EBPG5200, with Figure 4.10 depicting the simplified structure of the e-beam system. Electron beams, generated by thermally assisted field emission sources, are accelerated by the anode. The accelerated electron beams pass through a set of magnetic deflection coils, which tilt and shift the incident alignment onto the substrate. A series of apertures are employed in the system to converge and exclude scattered electrons, thereby controlling spot size and current density. The final patterns are defined by the main and subfield deflection coils. Analogue driven signals, generated by the patterns generator, deflect the electron beam to produce patterns designed by the user. The beam blanker in the column is utilised to turn the electron beams on and off.

Most of jobs completed in this work requireed multiple steps, necessitating alignment to ensure that each step overlays with the previous exposure. In electron beam lithography, a different



Figure 4.10: Simplified representation of EBPG. Information from [117]

markers set are employed compared to photolithography. Instead of manual alignment, the EBPG can automatically search for registered marker positions. A set of markers, consisting of individual squares sized at  $20 \,\mu m$ , are fabricated on the four corners of the sample. By uploading the markers coordinates, the machine can search for the correct position and form electron beam images at each position. These images then are analyses by algorithms, enabling the machine to adjust the position and offset of the sample to complete the alignment process. To ensure clear electron beam images, metals with higher atomic numbers are recommended, as they have more back-scattered electrons to form a clear image. Au and Pd are used in this work.



Figure 4.11: Typical marker sets used in Ebeam lithography.

## 4.5 Resist Development

After exposure, resist should be developed in a specific solvent. In the case of using positive resist, which is compatible with both e-beam and photolithography, the exposed area is supposed to be removed while the unexposed area remains on the surface. Ideally, there should be no resist left on the area where resist exposed to the UV light or the electron beams. However, in many cases, the unexposed area usually is affected lightly, and the exposed area may retain a small amount of resist on the sample surface, which is known as resist residue.

Resist residue can cause a series of issues for the next fabrication steps. The residue affects the adhesion between the metal and substrate. For metal-substrate contact, the residue also builds up a capacitor which influences the properties of the devices. To remove the resist residue, a low-power oxygen plasma is typically used to clean the surface[118]. This process, known as ashing or descumming, involves oxygen plasma under vacuum chamber interacting with the resist, oxidising it into gaseous species such as carbon dioxide ( $CO_2$ ) or carbon monoxide (CO)[119]. The plasma power and duration can be adjusted to adapt to the different types of resist applied on the surface. Extra care required in this step, as hydrogen termination can be easily damaged by oxygen plasma. Some alternative methods have been explored to avoid the damage to hydrogen termination, which will be talked about later.





(a) Sample with positive resist after exposure. (b) Sample with positive resist after development with resist residue left on the exposed area. (c) After ashing, the resist residue is removed.

For both photo and e-beam resist, development is affected by temperature, development duration, solvent concentration, and resist thickness. Additionally, the exposure duration used in photolithography and dose applied in e-beam lithography influences the development as well. The parameter for the resist used in this work are listed in the table below.

## 4.6 Metallisation & lift-off

Metalisation is a crucial step for pattern transfer and metal-semiconductor contact formation. Various techniques, such as physical vapor deposition (PVD), chemical vapor deposition (CVD) and electroplating, are employed for metal deposition. In this work, PVD is used for this purpose.

PVD is a process whereby solid-state target metals are transformed into the vapor phase, and vaporised metal is then condensed onto the sample surface. To vaporise the target metal, either thermal evaporation or electron beam evaporation can be used. Electron beam evaporation is utilised in this work for metal deposition. It is imperative that the deposition chamber maintains high vacuum conditions; otherwise, gaseous molecules and other particles may disrupt the film uniformity and introduce contamination. In addition, the high vacuum chamber prevents oxidisation of the metal.



Figure 4.13: Simplified electron beam evaporation chamber.

Figure 4.13 illustrates the simplified inner structure of an electron beam evaporation machine. The chamber is separated into two main chambers: the source chamber and the sample load chamber. The chamber containing the metal target is maintained under high vacuum to prevent contamination and oxidation of the metals. After loading the sample onto the sample holder, the loading chamber is evacuated using a rough pump and cryo pump to reduce the pressure inside the chamber. When the pressure in the loading chamber approaches that of the source chamber, the shutter between the two chambers is opened. Electron beams are then used to heat up the

#### CHAPTER 4. FABRICATION

metal target to its melting point. Once the evaporation rate stabilises at the desired rate, the substrate and source shutter are opened to allow the metal deposit onto the sample surface.

The entire process is completed under high vacuum conditions, and the deposition rate and thickness are monitored by a gold-plated Quartz Crystal Microbalance (QCM). The QCM is positioned near the sample location. As material is deposited on the crystal, its resonance frequency changes. This change is correlated with the mass change, providing valuable information about the deposition rate and thickness.



Figure 4.14: Lift-off process for sample with single layer of resist (a) and sample with double layers of resist (b).

Lift-off is a process whereby the unexposed resist is removed by specific solvent, leaving the metal deposited only on the developed area of the surface. This process completes the pattern transfer. Electron beam evaporation is a non-conformal process which means that the deposition is unable to cover the edge wall of the sample unless the sample is tilted to a specific angle. However, with a single layer of resist, the metal might still adhere to the resist edge wall, potentially causing the metal on the developed area to be partially or totally stripped off during the lift-off process. To address this issue, a double layer resist is applied to create a structure known as an under-cut, which assists the lift off process.

Figure 4.14 illustrates the differences between single layer resist lift-off and double layer lift-off techniques. With a single layer of resist, the metal edges adhere to the resist edges, which can cause the lift-off process to fail. However, using two layers of resist with different sensitivities to the UV light or electron beams creates a structure called an "under-cut". This structure serves two purposes: firstly, it isolates the metal from the resist, and secondly, it provides spaces between the resist and the metal, which helps the lift-off solvent dissolve the resist from the inside.

In this work, LOR is utilised as the bottom layer for photo-lithography. Since LOR is not sensitive to UV light, the development of the top layer S1805 does not affect it. However, LOR can be etched by Tetramethylammonium Hydroxid (TMAH). As TMAH etches LOR faster than S1805, with proper controll of duration, LOR can create an under-cut. For e-beam lithography, resist with different molecular weight and concentrations are chosen to establish the under-cut. Molecular weight is linked to resist sensitivity, and concentration is associated with resist thickness. A lower molecular weight implies higher sensitivity, while a higher concentration results in a thicker resist after baking. Therefore, 50 K 12% PMMA is employed as the bottom layer, while 200 K 4% PMMA serves as the top layer.

The lift-off is usually conducted in a 50 °C water bath, with the sample immersed in a specific solvent for few hours or overnight. The solvent used for LOR lift-off is SVC-14, while the solvent used for PMMA lift-off is acetone.

### 4.7 Selective Etching of metals

Selective etching of metals is crucial for subtractive pattern transfer. Dry etching and wet etching are commonly employed techniques to remove unwanted materials from the sample surface.

Dry etching is a process that utilises plasma to etch away unwanted materials. The process involves chemical reactions between the ions and materials, and physical bombardment. In the vacuum chamber, all by-products are pumped out of the system. Dry etching is an anisotropic process, meaning that it produces a controlled and directional etch profile. For example, ashing, a mild dry etch process using oxygen plasma, is commonly employed to remove the resist residue. However, the ions and radicals generated during the dry etching can potentially damage the hydrogen termination.

To protect the diamond surface termination, wet etching is necessary. Wet etching is a purely chemical process involves soaking samples in specific chemicals to remove unwanted materials. This process is isotropic, meaning it etches uniformly in all directions. Wet etch offers advantages such as simple equipment requirements, high throughput and high selectivity compared to dry etching. In this study, wet etching is employed to remove unwanted gold from the diamond surface. The solvent used for wet etching is a mixture of  $KI/I_2$ . The etching rate can be regulated by diluting the solvent with RO water and adjusting the reaction temperature, thus enabling a more controlled etch profile.

$$2Au + I_2 \rightleftharpoons 2AuI \tag{4.1}$$

In this work, a gold 'sacrificial' layer is employed to protect the hydrogen-terminated diamond surface from contamination by resist residue or damage from plasma. Following photo or e-

beam lithography, the sample is developed and then subjected to ashing in oxygen plasma. The gold layer acts as a barrier, protecting the termination from the plasma. The subsequent step involves wet etching of the gold to transfer the pattern to the sample surface or open the windows in the Au layer.



Figure 4.15: Gold wet etch with isotropic etching profiles.

## 4.8 Oxide deposition

Deposition of the oxide layer is a crucial aspect of the fabrication process. It can serve as a gate oxide to create a capacitor. As discussed in the previous chapter, the oxide with a LUMO near the hydrogen-terminated diamond valence band maximum acts as an electron acceptor and improves surface conductivity. It is essential to choose the appropriate method for oxide deposition for the intended purpose.

There are many different methods to put oxide on the sample surface, such as chemical vapour deposition (CVD), physical vapour deposition (PVD), atomic layer deposition (ALD). In this work, ebeam PVD, thermal PVD and ALD are applied to deposit oxide for different purpose.

As we talked in the previous part, PVD is commonly employed for metal deposition. It may also be used for oxide deposition. The fundamentals principles of PVD metal and PVD oxide deposition are similar. In the case of  $MoO_3$  and  $V_2O_5$  deposition, the process is carried out using thermal PVD deposition within Plassys 5 system. The main distinction between thermal PVD and e-beam PVD lies in the method of vaporization. In thermal PVD, the oxide is vaporized by heating the holder to the material's melting point, after which the evaporated material condenses onto the sample surface to form a thin film. A gold-plated quartz crystal microbalance is employed to monitor the deposition rate and thickness.

In contrast, the deposition of  $Al_2O_3$  differs slightly from other oxide due to its higher melting point, which is approximately a few thousand degrees Celsius. In this case, electron beams are



utilized to heat up and evaporate the target from a solid state to a vapor state.

Figure 4.16: The inner structure of Plassys 5 (used for PVD of oxide materials). The left part of the chamber is designated for thermal deposition, where materials such as  $Mo_2O_3$  and  $V_2O_5$  are used. Conversely, the right part of the chamber is designated for electron beam evaporation, which is suitable for materials with high melting point, such as  $Al_2O_3$ . A shutter between the target and the sample holder prevents materials from condensing on the sample surface until the deposition process stabilises. Additionally, a Gold Crystal Microbalance (GCM) situated between the sample holder and target material is utilised to monitor the deposition rate and thickness.

ALD is also employed in this work for the deposition of  $Al_2O_3$ . ALD is a thin film deposition technique that enables precise control over film thickness at the atomic level. It operates through a sequential, self-limiting surface reaction based on the temporal separation of two or more reactants. This approach allows for the deposition of conformal and uniform thin films on substrates with complex geometries [120].



Figure 4.17: PVD and ALD oxide deposition profile.

ALD deposition provides a better thickness control, uniformity, conformal profile and complex geometries.

Precursors are chemicals that contain the elements necessary for the target material deposition in ALD. Typically, two or more precursors are used for the deposition of a single material. The precursor is introduced into the vacuum chamber, where it reacts with the substrate surface to form a single layer of chemical groups containing one of the element required in the final material. Before the next precursor reacts with the previous products, the chamber is purged to remove all remaining precursor and potential by-products. Using  $Al_2O_3$  deposition as an example, precursor 1, contains Al, reacts with the substrate, followed by a chamber purge to clean the chamber. Subsequently, precursor 2, containing O, is introduced into the chamber to form  $Al_2O_3$  with the previous Al. The chamber is purged again to remove all by-products. This constitutes a complete deposition cycle for ALD, where each cycle only grows one single atomic layer of  $Al_2O_3$  due to the self-limiting nature of this process. This cycle is repeated to achieve the desired thickness of materials.

To activate the substrate surface and facilitate reaction with the precursors, thermal or plasma energy is applied during the reaction step. Plasma-enhanced ALD offers several advantages, including a wider choice of materials, increased growth rate, improved material properties and reduced substrate temperatures [120]. However, for sensitive samples like hydrogen-terminated diamond, exposure to plasma during this process can potentially damage the surface. Therefore, caution must be exercised when employing plasma-enhanced ALD on such substrates.



Figure 4.18: A complete ALD cycle with ALD  $Al_2O_3$  as an example. If the process is a thermal process,  $H_2O$  is used as the second precursor to provide oxygen. If the process is plasma enhanced,  $O_2$  plasma is used as the source of oxygen.

In the case of  $Al_2O_3$  deposition,  $H_2O$  serves as the source of oxygen in thermal deposition, while  $O_2$  plasma is utilised in the plasma-enhanced deposition. In this work, to preserve the integrity of the hydrogen-terminated diamond surface, all  $Al_2O_3$  deposition is conducted using thermal ALD.

An experimental investigation was designed to explore the effects of thermal ALD and plasmaenhanced ALD on the surface conductivity of hydrogen-terminated diamond. Three 3 *mm* single polished diamond samples (A,B and C) underwent an acid cleaning process and hydrogen termination with partners in Australia. Prior to proceeding to the next step, all three samples underwent solvent cleaning to remove potential contamination on the diamond surface post-hydrogen termination.

Following the cleaning process, four-probe VDP measurement were conducted. Silver dag was applied to the four corners of the samples to ensure good contact between the probes and sample surface, particularly after the oxide deposition. The sheet resistance of these three samples in air was measured as  $9 k\Omega/\Box$ ,  $11 k\Omega/\Box$  and  $9 k\Omega/\Box$  separately.

Subsequently, the three samples underwent  $Al_2O_3$  ALD under different parameters to investigate the influence of different ALD processes on the hydrogen termination. Sample A was subjected to a standard thermal ALD  $Al_2O_3$  deposition recipe, with a reaction activation temperature of 200 °C. Sample B underwent the same recipe as sample A, but with an additional in-situ anneal at 400 °C for one hour anneal before the deposition, aimed at removing any potential air adsorbates on the hydrogen-terminated diamond surface. Sample C underwent  $O_2$  plasma-enhanced  $Al_2O_3$ deposition without annealing. All three samples received a 10 nm thickness of  $Al_2O_3$  on the surface.

Following the deposition, VDP measurements were conducted again to assess surface conductivity. The addition of silver dag before deposition facilitates easy probing through the silver stack to make contact with the diamond surface. Post-deposition, the sheet resistance of sample A dropped from  $9k\Omega/\Box$  to  $6k\Omega/\Box$ , Sample B sheet resistance increased to  $20k\Omega/\Box$  from  $11k\Omega/\Box$  and Sample C sheet resistance is too high to be measured by the machine.

Sample	A	В	С
Process	Thermal ALD	Thermal ALD	Plasma ALD
	with $400^{\circ}C$ an-		
		nealing	
<i>R<sub>s</sub></i> Before	$9 k\Omega/\Box$	$11 k\Omega/\Box$	$9 k\Omega/\Box$
deposition			
$R_s$ After	$6 k\Omega/\Box$	$20 k\Omega/\Box$	$>2 M\Omega/\Box$
deposition			

Table 4.2: Sheet resistance change after ALD deposition under different growth conditions.

The decrease of the sheet resistance of sample A suggests that the thermal ALD process has enhanced the surface conductivity of the hydrogen-terminated diamond. Although the exact mechanism behind this improvement remains unclear (because the electron affinity of  $Al_2O_3$  should not be high enough to act as an electron acceptor on H-diamond), it is hypothesized that specific defects exist in the oxide layer act as acceptor, thereby enhancing conductivity [121].

The sheet resistance of sample B increased by  $9 k\Omega/\Box$  after the deposition. This may be contributed to two factors: Firstly, the annealing process removed air adsorbates, potentially depleting acceptors on the surface. Secondly, the high temperature annealing may have damaged
the hydrogen termination. To discern the cause of the increased sheet resistance, Sample B was treated with Buffered Oxide Etchants (BOE) to remove all the  $Al_2O_3$ , followed by VDP measurement. The sheet resistance remained the same after wet etching of the oxide as when measured with  $Al_2O_3$  encapsulation and a few days exposure in air, indicating that the annealing process most likely desorbed part of hydrogen termination from the diamond surface. Given the vacuum level in the ALD chamber is between 200 mTorr and 80 mTorr, potentially partial oxygen termination in the chamber could have occurred during the annealing process for sample B.

Sample C subjected plasma-enhanced ALD, exhibited no measurable conductivity after Plasma ALD  $Al_2O_3$  deposition. The oxygen plasma used in the second precursor can remove hydrogen and oxygen-terminates the diamond surface, rendering it non-conductive. After BOE removal of the Plasma ALD  $Al_2O_3$  and exposure to air, the VDP measurement indicated no measurable surface conductivity, confirming that the hydrogen termination has most likely been removed/damaged by the ALD plasma.

Overall, to preserve the hydrogen termination on the diamond surface and maximise conductivity, the optimal approach for ALD  $Al_2O_3$  deposition in this study is thermal ALD without the annealing process. Plasma-enhanced deposition can damage the hydrogen termination. Further experiments are warranted to elucidate the effects of annealing on the sample , as hydrogen termination should not be removed by a one-hour 400 °C annealing [122], [123].

# 4.9 Diamond MESFETs Fabrication Flow

In a typical hydrogen-terminated diamond MESFETs fabrication process, several key steps are involved, including marker placement, isolation, bondpad, active region etching, gate formation and oxide deposition. The smallest feature size of the diamond MESFET in this study is 2  $\mu m$ . Thus photo lithography is used for the fabrication. Figure 4.19 illustrates the fabrication for the hydrogen-terminated diamond MESFETs, wherein ALD  $Al_2O_3$  serves as the encapsulating oxide.



Figure 4.19: Fabrication of hydrogen terminated diamond MESFETs with gold sacrificial layer to protect the surface conductivity and ALD  $Al_2O_3$  as encapsulating layer.

### Sample clean

A 4.5  $mm \times 4.5 mm$  double-polished (100) single crystal diamond sample was purchased from Element six. A hot acid clean was employed to eliminate the potential metal contamination and oxidize the diamond surface. Subsequently, the samples underwent solvent cleaning procedures before being sent to Australia for hydrogen termination using hydrogen plasma. Following the hydrogen termination, a solvent clean consists of 5 minutes in acetone followed by 5 minutes IPA with ultrasonic was conducted to remove any residue contamination.

## Gold sacrificial layer deposition

The initial stage involved depositing a layer of gold onto the diamond surface to shield the surface termination from potential damage during subsequent ashing processes. Additionally, this thin layer serves as the ohmic contact pad in the subsequent steps. Samples were secured onto the holder, and an 80 *nm* layer of gold was evaporated onto the diamond surface using an electron beam.

## Markers

The fabrication involves multiple layers of lithography to be completed. To align each layer with the previous one, makers are fabricated on the sample surface during first exposure. Double layers of resists are employed for the convenience of lift-off process. LOR and S1805 are used as the bottom and top layer, respectively. After the development of S1805, oxygen plasma asher is utilized to remove the resist residue. Subsequently, CD-26 (2.5% TMAH) was applied to remove the LOR underneath the developed top layer of resist, with over-etching applied to create an undercut. Another round of ashing is conducted to eliminate LOR residue. Following this, a gold wet etch is employed to remove the gold underneath. The metallization for markers used in this work consists of 20 *nm* of Titanium and 80 *nm* of Gold. Titanium aids in ensuring gold adherence to the diamond surface. Subsequently, the sample is immersed in SVC-14 for an overnight water bath at 50 °C. Upon retrieval from the water bath, a pipette is used to fully remove the resist, followed by cleaning the sample surface with IPA.



Figure 4.20: Marker used in photo-lithography

#### Isolation

Isolation is a crucial step aimed at electrically isolating devices on the substrate from each other, ensuring that the current is confined to a single device. To achieve this, the hydrogen termination over the surface, where insulation is required, must be removed. The sample with marker on is spun with a single layer of resist. An undercut is not necessary in this step since it is defined by subtract process(wet etch) instead of addition process (lift-off). The MA6 is utilized to align the pattern with the previous marker exposure. Subsequently, the sample is developed and ashed to remove the unwanted resist. Following this, gold etching is applied to remove the gold where the photo-resist has been developed. The final step involves oxygen plasma ashing to replace all the hydrogen termination with oxygen, resulting in the formation of the isolation area. An overnight acetone with water bath is utilized to strip off the resist. The remaining gold on the surface serves as both a protection layer for the hydrogen termination and an ohmic contact metal.



Figure 4.21: Sample with gold after isolation step.

## **Bond pads**

The ohmic pad is formed from 80 nm of gold that is too thin and fragile for the measurement by physical probing directly without causing damage to the contacts. Therefore a more robust and thicker bond pad layer is fabricated over the ohmic pad. Similar to the fabrication of marker level, a double layer of resist is spun onto the surface. After the development and etching of the two layers of resist, gold is wet etched to open the window for the bond pad. Ebeam PVD is used to deposit 10 nm of Ti and 180 nm of Au as the bond pad metal stack. Titanium here performs as an adhesion layer, increasing the robustness of the top gold layer. The sample is then rinsed in SVC-14 for lift off.

### Active device region

Test structures like TLM and VDP require partial removal of gold to expose the hydrogen termination surface for measurement. Similar to the isolation step, wet etching is used to transfer



Figure 4.22: MESFETs after Bond Pad lift off.

the pattern, a single layer of photo-resist is sufficient for the process. S1805 is spun and baked onto the sample surface. The MA6 aligns the active area with the exist patterns. A 1:4 ratio of Microposit to RO water is used to develop the exposed resist. Since the smallest feature in this step is 2  $\mu m$ , a hot gold etch is employed to control the etch rate and achieve a better etch profile. One portion of the Au etch is diluted with ten portions of RO water. Then, the solvent is heated to 60 °C on a hotplate. The sample is then immersed in the solvent for 10 seconds to remove the unwanted gold. An overnight water bath in SVC-14 is required after the gold etch to strip off the resist.



Figure 4.23: Active region fabrication before and after gold etch.

### Gate

The next step for a diamond MESFET is the gate level. A lift-off process is required for this additional step. LOR and S1805 are used as the bottom layer and top layer, respectively. After exposure and development, ashing is employed to remove all the resist residue. Then, a warm gold etch is used to achieve smooth edges. This gold etch not just open the window for gate metal deposition but also separate the remaining gold into two ohmic metals. Then ebeam PVD is used to deposit 20 *nm* of Aluminium, 20 *nm* of platinum and 80 *nm* of gold on the sample surface as gate metal stack. Lift off is performed after the deposition, and a pipette is used to remove the resist and metal carefully.



Figure 4.24: MESFETs after gate lift off.

## **Oxide deposition**

To improve the thermal stability and surface conductivity, following the gate step, 10 nm of thermal ALD  $Al_2O_3$  is deposited on the sample surface to encapsulate the device. To ensure proper contact between the measurement probes and the bond pad metal, an additional step is carried out to open a window over the bond pad and gate pad. Subsequently, the sample is immersed in 2% of buffered oxide enchants (BOE) for 100 seconds to remove the unwanted  $Al_2O_3$ .



Figure 4.25: Resist over bond pad and gate pad is exposed and developed. Thermal ALD  $Al_2O_3$  is wet etched by BOE to provide access to bondpads for probing.

# 4.10 Accumulation Channel diamond MOSFETs Fabrication Flow

Normally-off, "Accumulation Channel" diamond MOSFETs in this work are fabricated with electron beam lithography, which shares similarities with the photo lithography process used for MESFETs. This subsection will primarily focus on providing additional details regarding the different fabrication steps while simplifying those that are similar. The key steps include markers, isolation, ohmic pad, bond pad, oxide deposition and gate.



Figure 4.26: Normally-off "Accumulation Channel" hydrogen terminated diamond MOSFETs fabrication flow.

#### **Sample Preparation**

A single crystal (100) single-polished diamond sample is purchased from element six. Thorough cleaning, including acid cleaning and solvent cleaning, is conducted to oxidize the surface and remove potential contamination. Subsequently, the sample is sent to our collaborator in Australia for hydrogen termination. After the termination process, the sample is cleaned with acetone and IPA. Following the termination process, the sample is exposed in air for a few days, and Hall kit is used to monitor the surface conductivity. Once the surface conductivity reaches the optimal value and becomes stable, a sacrificial layer of 80 *nm* of gold is deposited on the diamond surface.

### **Markers and Isolation**

The makers used for Ebeam lithography are different from those used in photolithography. Ebeam lithography markers consist of several 20  $\mu m$  squares. A double layer of PMMA is used for the lift-off marker fabrication. After spining and baking, a thin layer of metal (a mix of Au and Pd) is sputtered onto the sample surface as a charge dissipation layer (CDL) to channel all electrons to the ground and prevent pattern distortion. The marker layers in the design file are uploaded to EBPG5200.

The first step after exposure is to rinse the sample in pure gold wet etch for 30 seconds to remove the CDL. Subsequently, the sample is developed in MIBK:IPA 1:1 developer for 30 seconds, followed by a 30 seconds rinse in IPA. Due to the sensitivity of the developer to temperature, a water bath is used to warm up the solvent to 23 °C before development. As the bottom layer of PMMA is more sensitive to the electron beams, an undercut is formed after the development.

Prior to the wet etch of the gold layer, an 80W 2-minute oxygen plasma ashing is employed to remove the resist residue. Then,a 40-second room temperature gold etch with a 1:4 gold etch to RO water ratio is used to remove the unwanted gold. Following this, 20 nm of Titanium and 100nm of Palladium are deposited using the PVD method. Since there is a gold blanket etch later, Palladium is used as marker instead of gold to avoid damaging of the markers. The lift off involves soaking in acetone with a 50 °C of water bath for 4 hours.

Isolation is processed in a similar manner to marker fabrication. To ensure alignment with the previous step, coordinates of four markers on each corners of the sample are submitted to EBPG. The machine can then locate the marker position by detecting the backscattered electrons from the metal marker and the substrate diamond. After the development and gold etch, the isolated area is exposed to air. Subsequently, an 80 W 3-minute of oxygen plasma is employed to replace the exposed hydrogen-terminated with oxygen, resulting in an insulating surface. Two isolation test structures can be found in figure4.27, with three yellow pads in a row both at the top and



bottom of the design, facilitating testing for isolation completeness.

Figure 4.27: The L-edit design of MOSFETs and images of isolation and marker structure.

## **Ohmic Contact**

In contrast to photolithography, where the ohmic contact is formed from the gold sacrificial layer, ebeam lithography employs an addition method to apply ohmic contact to the sample, offering more choices for the ohmic metal. Therefore, a double layer of PMMA is used in this step to create an undercut for the lift-off process. A CDL is sputtered on the double layer of PMMA and later removed by gold wet etch. The PMMA is exposed, developed and ashed to remove all the resist from the area where the ohmic pad will be placed. Subsequently, room temperature gold wet etching is utilized to remove the gold and create an opening for ohmic pad deposition. The next step involves using ebeam PVD to evaporate 80nm of Palladium onto the sample surface, followed by a 4-hour acetone soak in 50 °C water bath to complete the lift off process.

#### CHAPTER 4. FABRICATION

The choice of different ohmic metals in ebeam lithography is motivated by the observation that the ohmic contact between hydrogen-terminated diamond and Pd exhibits lower contact resistance and better adhesion compared to that of Au. Fabrication and measurements are carried out to verify this assertion, and the experimental results will be discussed in greater detail in the subsequent chapter.



Figure 4.28: Ohmic level optical image. (a) ohmic pad develop (b) ohmic pad gold etch (c) Pd ohmic lift off.

#### **Oxide deposition**

After the fabrication of the ohmic pad on the diamond surface, a blank gold etch is performed to remove all the remaining gold used to protect the hydrogen termination. Then sample is mounted into PLASSYS 5 for ebeam  $Al_2O_3$  deposition. Before the start of the deposition, a one-hour anneal at 350 °C is conducted to desorb the air adsorbates between the source and drain. Subsequently, 35 *nm* of ebeam  $Al_2O_3$  was deposited on the diamond surface at a rate of 0.02 nm/s.

As discussed earlier, ebeam deposition is non-conformal, meaning that the edge wall can not be uniformly covered with deposition materials. Therefore, to address this issue, 15 nm of thermal ALD  $Al_2O_3$  is deposited on the diamond surface after the ebeam deposition. This 15 nm of ALD  $Al_2O_3$  prevents the gate metal and ohmic pad shorted. The deposition of oxide layer serves as both devices gate dielectric and protection layer over the surface hydrogen termination.

#### **Bond Pad**

Bond Pads are placed on the ohmic contact to establish a robust contact for the later measurement. A double layer of ebeam lithography resist is applied by spinning and baking it onto the sample surface. After development and ashing, the area designated for the bond pad is cleaned up. To establish direct contact between the ohmic contact and the bond pad, the oxide on the

#### CHAPTER 4. FABRICATION

bond pad position needs to be removed. To prevent potential damage to the hydrogen termination, wet etching is preferred over dry etching. The sample was immersed in 5% buffered oxide etchant for 45 seconds to eliminate the unwanted oxide on the ohmic contact. Over etching of the oxide layer is necessary to ensure proper contact formation between the bond pad metal and the ohmic contact.



Figure 4.29: Optical microscope image of bond pad oxide etching. The colour contrast shows the evidence of oxide layer over etch.

After wet etching the oxide layer, PVD is employed to deposit a metal stack consisting of 20 *nm* of titanium and 160 *nm* of gold for the bond pad metal stack. Titanium functions as the adhesion layer and enhances the durability of the gold layer. Subsequently, the sample is immersed in acetone with 50 °C water bath for overnight to remove unwanted metal and resist. Figure 4.30 displays the microscope image of the MOSFETs after bond pad lift-off process.



Figure 4.30: Optical microscope image of bond pad after lift off.

### Gate

The final structure in this fabrication process is the gate. A double layer of e-beam resist is spun and baked onto the sample surface. The EBPG is utillized to expose the gate area, followed by a 30-second development in a 23 °C 1:1 MIBK:IPA developer to remove the exposed e-beam resist. Then, an 80W oxygen plasma treatment is employed to eliminate the resist residue. The gate metal stack is deposited onto the sample surface using the PVD method. This stack consists of 20 nm of aluminium, 20 nm of platinum and 100 nm of gold, which are evaporated onto the sample surface. The sample is then submerged in acetone at 50 °C water bath overnight for lift-off process.



Figure 4.31: Optical microscope image of gate level fabrication. (a) Developed Sample after metal deposition (b) Gate metal lift off (c) Enlarge part of the gate structure.

# 4.11 Chapter Summary

This chapter delves into the techniques and processes employed in the fabrication of diamondbased devices, focusing on photo lithography, e-beam lithography, and various physical vapor deposition (PVD) methods. It also discusses the selection and application of different types of resists for lithography, notably the use of double resist layers to create undercut structures.

The chapter provides a detailed exploration of each fabrication step, including sample cleaning, resist spinning, wet etching and lift-off processes, elucidating the significance of each stage in device manufacturing.

The final sections of the chapter offer comprehensive insights into the fabrication of hydrogenterminated MESFETs and normally-off MOSFETs, highlighting the intricate procedures involved in their construction and emphasizing the importance of precise execution at each stage of the process.

# Chapter 5

# **Characterisation and Measurement**

This chapter includes a variety of techniques applied to characterise and quantify diamond samples and the devices fabricated on the diamond surface. It covers two central parts, which are surface profiling and electrical measurement. Scanning Electron Microscope (SEM), Atomic Force Microscope (AFM) and Contact Profiler Dektak XT are introduced in the part of surface profiling. In the electrical measurement part, Transmission Line Measurement (TLM), Van Der Pauw Measurement (VDP) and Field Effect Transistor (FET) are covered by the fundamental to application.

# 5.1 Surface Profiling

#### 5.1.1 SEM

In Scanning Electron Microscopy (SEM), a focused electron beam is applied to the sample surface to produce topographical images. Electron beams interact with samples, generating various signal types, including Secondary Electrons (SE), Backscattered Electrons (BSE), Auger Electrons, and Characteristic X-rays [124]. Of these signals, Secondary Electrons and Backscattered Electrons provide the majority of topographic information.



Figure 5.1: Illustration of signals generated by the interaction of incident electrons and the sample surface in Scanning Electron Microscopy.

There are two types of scattering that occur between incident electrons and specimen: elastic scattering and inelastic scattering. [125] Inelastic scattering occurs when incident electrons interact with the atoms and electrons on the sample's surface, resulting in a significant transfer of energy from the incident electrons to the target sample. This process excites loosely bound electrons, known as secondary electrons. Typically, secondary electrons have low energy and escape only a few nanometers from the sample surface before being detected. As a result, secondary electrons provide highly accurate topographic information with a resolution of approximately 10 nanometers[125].

Incident electrons deflect when interacting with the outer shell electrons or atomic nucleus of the sample, resulting in elastic scattering that generates backscattered electrons. Electrons undergoing elastic scattering can retain 60% to 80% of their initial energy. This means that electrons can travel a greater distance before being detected. thereby providing less precise topographic information with a resolution of approximately 1  $\mu m$ [125].

However, it's important to note that backscattered electrons also contain valuable compositional information. Elements with higher atomic numbers possess more positive charges, leading to the generation of a greater number of backscattered electrons. As a result, the detector captures varying amounts of backscattered electrons from different elements, creating contrast in the final image.



Figure 5.2: General SEM configuration. [126]

Figure 5.2 illustrates the configuration of a scanning electron microscope (SEM). The filament generates electron beams that initially have a width too broad for surface scanning. Consequently, SEM is equipped with a series of lenses to compress and focus this beam. The condenser lenses play a crucial role in converging and collimating the electron beams into a series of parallel streams. Within these condenser lenses, several types of apertures are applied to exclude scattered electrons, which contribute to the sharpness and shape of the electron beams.

Reducing the size of the aperture allows for an increased depth of field and a decrease in the final current, consequently enhancing the resolution of the final image. Before scanning the sample surface, the objective lens refocuses the electron beams to eliminate divergent electron beams. However, contamination on the aperture and defects in the lenses can cause beam profile distortion, leading to stretching of the final image in various directions. This optical defect is referred to as astigmatism. Therefore, stigmators located around the objective lens are employed to correct this beam shape distortion and ensure that the electron beams are focused into a circular pattern.

#### CHAPTER 5. CHARACTERISATION AND MEASUREMENT

Focused electron beams are applied to the sample surface and scanned point by point, with control provided by the scanning coils around the objective lens. Electron detectors capture secondary electrons (SE) and backscattered electrons (BSE) by applying bias to the detector[124]. The resulting signals are then displayed on the view monitor.

The scanning electron microscope offers several advantages, including high resolution, high magnification, and the ability to provide both compositional and topographic information. Additionally, with the inclusion of an Energy-Dispersive X-ray Spectroscopy (EDS) detector, excited X-rays from the sample surface can be utilized for elemental analysis.

However, it's important to note that the fundamentals of using electron beams to form images limit the range of suitable samples. When analyzing insulating samples, two common issues that arise are surface charging and image drifting. Furthermore, the application of accelerated high-energy electron beams has the potential to alter or damage the sample surface.

### 5.1.2 AFM

The Atomic Force Microscope (AFM) is a type of microscope that utilizes the interaction forces between scanning tips and the atoms and molecules on a sample's surface to provide topographic information. This information includes surface roughness, imaging, and the manipulation of surface properties [127].

Figure 5.3 shows the configuration of Atomic Forces Microscope (AFM). Two main interaction forces come into play between the tips and the sample surface. When the tip is far from the surface, forces are negligible. As the tip approaches the sample surface, attractive forces, such as Van der Waals interactions, become dominant and can be sensed even when the tip and sample are separated by more than 10 *nm*. As the tips continue to move closer to the sample surface, repulsive forces, known as Coulombic interactions, take precedence in the interaction between the tips and the sample surface. The combination of these two types of forces forms the interaction curve between the sample surface and the tips[128].



Figure 5.3: General configuration of Atomic Force Microscopy

These interaction forces cause slight movements in the connected cantilever, which are recorded by a focused laser beam. Consequently, the signal can be translated into a topographic image of the sample.

The cantilever is connected to a scanner constructed from piezoelectric material, which enables precise control of the tip's movement along the X, Y, and Z axes. By applying voltage to the scanner, it undergoes deformation, effectively maneuvering the tips across the sample surface[129].

When the tips make direct contact with the sample surface during the scanning process, they can easily become blunted or damaged. To prevent this, most AFMs employ a feedback circuit. This circuit maintains a constant distance between the tips and the sample surface by adjusting the tips' position up and down. This not only reduces damage to the AFM tips but also protects the delicate sample surface.

Several parameters are used to adjust the feedback system. The "setpoint" is the value for the amplitude or deflection that the feedback system aims to maintain. "Feedback gains" determine how quickly the system responds to the scanning signal. A slow scan rate and high feedback gains are typically chosen to achieve better scanning results. However, setting the feedback gains too high can lead to an unstable and oscillating system, causing high-frequency noise in



Figure 5.4: AFM tips and sample surface interaction curve.

the final image.

Tapping mode and contact mode are two fundamental operating modes in AFM[127]. In contact mode, the tips maintain continuous contact with the sample surface to collect topographic information. In contrast, tapping mode involves the vibration of the tips above the sample surface with intermittent contact. This approach prolongs the lifespan of the tips and reduces lateral forces. However, it's worth noting that the scanning speed in tapping mode is generally slower compared to contact mode.



Figure 5.5: Image of different width of AFM tips scan a surface with high aspect ratio and density.

#### CHAPTER 5. CHARACTERISATION AND MEASUREMENT

AFM data collection is susceptible to several common artifacts resulting from the tips, scanner, feedback, and ambient vibrations. The final AFM image is a convolution of the sample's topography and the tip's shape. Consequently, when the sample exhibits a larger aspect ratio than that of the tip, the scan results may be misrepresented. To obtain accurate results for samples with high aspect ratios, equally sharp tips are essential. This effect can lead to blind spots when scanning high-density features with wide tips. Figure 5.5 illustrates how tip thickness influences the final scan results. Additionally, there are other tip-related artifacts, such as multiple images caused by chipped tips, as shown in Figure 5.6.



Figure 5.6: Double image caused by chipped AFM tip.

Scanner-related artifacts occur when the piezoelectric material exhibits non-ideal behavior, such as creep, non-linearity, and hysteresis[128]. These artifacts become noticeable when there are image distortions in the results.

Feedback-related artifacts emerge when there is a poor trace and retrace trajectory or when the gains are set too high. In the scanned image, these two feedback-related artifacts result in blurred images or high-frequency noise.

Lastly, ambient vibration-related artifacts typically manifest as horizontal bands in the image.









## 5.1.3 Surface Profiler DektakXT

Surface Profiler Dektak XT is an equipment for step height measurement with a vertical resolution of 0.01 *nm* on 6500 *nm* range. Figure 5.8 shows the configuration of Surface Profiler.

Beneath the bridge structure, a sensor crafted from piezoelectric material and a stylus composed of diamond come together to create a measurement and feedback circuit, facilitating the collection of surface information. Initially, the measurement head descends to ensure that the stylus makes contact with the sample's surface. Subsequently, as the stylus traverses the surface, encountering raised or depressed features, it undergoes displacement either upward or downward. The sensor generates electrical signals that correspond to the stylus's displacement. Consequently, a computer processes these electrical signals, constructing a surface image of the measured area.



Figure 5.8: Configuration of Surface Profiler Dektak XT. [130]

The Surface Profiler DektakXT is widely employed for step height measurements. When compared to the Atomic Force Microscope (AFM), the Surface Profiler offers distinct advantages. It can scan larger areas at significantly higher rates, and its operation is generally more userfriendly than that of the AFM. However, the Surface Profiler has a noticeable drawback. Despite the ability to adjust the contact force, there is a risk of surface scratching by the stylus after measurements. Furthermore, the Surface Profiler is limited to scanning along a single axis, either the x-axis or y-axis, which means it cannot be used to characterize two-dimensional surface areas.

# 5.2 Electrical Measurement

#### 5.2.1 Transmission Line Measurement (TLM)

Transmission Line Measurement (TLM) is used to measure contact resistance ( $R_C$ ) between the ohmic metal and semiconductor as well the sheet resistance ( $R_S$ ) of the semiconductor.

A series of same size ohmic contact metal are deposited on the semiconductor surface with vary distance  $(L_x)$  between each two adjacent pads. When a bias applied on two adjacent contact pads, current against voltage curve gives the resistance  $(R_T)$  between those two pads.



Figure 5.9: Optical Microscope Image of TLM structure on hydrogen terminated diamond surface.



Figure 5.10: TLM gap graph.

The resistance measured above contains the resistance from three different parts those are the resistance between probes and ohmic metal  $R_M$ , contact resistance  $R_C$  and the semiconductor resistance  $R_{semi}$ .

$$R_T = 2R_M + 2R_C + R_{semi} \tag{5.1}$$

Since typically  $R_M \ll R_C$ ,  $R_M$  is negligible, 5.1 can be simplified to,

$$R_T = 2R_C + R_{semi} \tag{5.2}$$

Because

$$R_{semi} = R_S \frac{L}{W} \tag{5.3}$$

where  $R_S$  is the sheet resistance, then we can get

$$R_T = 2R_C + \frac{R_S}{W}L \tag{5.4}$$

With applying bias on each two adjacent ohmic pads,  $R_T$  against  $L_x$  curve can be plotted as figure 5.11.



Figure 5.11: Ideal Plot of  $R_T$  against  $L_x$ .

#### 5.2.2 Van Der Pauw Measurements (VDP)

Van der Pauw structure is an efficient way to characterise electrical properties, such as carrier mobility, carrier concentration, sheet resistance and carrier polarisation.

Figure 5.12 shows one of the basic principles (Hall effect) used for VDP structure to characterise the carriers properties. A magnetic field (*B*) applied perpendicularly to the surface of the substrate, in a direction perpendicular to the flow of electric current (*I*), propels electrons to move towards both sides of the substrate. The direction of electron movement is perpendicular to both the current and the magnetic fields. This behaviour leads to the accumulation of charges on both sides of the substrate, creating an internal electric field. As the number of charges increases, the newly generated internal electric field balances the force exerted on moving charges by the external magnetic field. Therefore, the moving charges can continue to move along the direction of the current. The potential associated with the electric field generated by the accumulated charges if known as the Hall voltage or  $V_h$ .



Figure 5.12: Hall effect

Nanometrics 550 Hall Kit 4 probes system is used for sheet resistance and hall coefficient measurement. Figure 5.13 shows the configuration of the equipment. Two of those probes push current through the sample surface and the other two measure the bias on the other two pads.



Figure 5.13: Nanometrics 550 Hall Kit 4 probes hall measurement system.

Figure 5.14 illustrates a microscope image of the VDP structure fabricated on a hydrogen terminated diamond surface. The middle hydrogen terminated area is designed to restrict the current



Figure 5.14: Microscope image of VDP structure for hall effect measurement.

flow and make sure a symmetric geology for the measurement which improve the carrier properties and sheet resistance measurement accuracy. The isolation is accomplished by oxygen plasma, and isolation structure was used to test the isolation. More details can be found in Chapter 4. When the probes get contact with pad 1 and pad 2, they inject current (I) into the structure, and the resulting voltage (V) can be measured by the other two probes over pad 3 and pad 4. These values of I and V are used to calculate the resistance ( $R_1$ ). Subsequently, the probes apply current through pad 1 and pad 3 while measuring the voltage across pad 2 and pad 4 to obtain a second resistance ( $R_2$ ). In the case of a uniform conducting sample surface,  $R_1$  and  $R_2$  should have the same value. Therefore, the sheet resistance can be calculated as follow,

$$R_S = \frac{\pi R}{\ln 2} \tag{5.5}$$

When the current flows through pad 1 and pad 4 with a positive magnetic field applied to the substrate, Hall voltage ( $V_h$ ) can be measured across pad 2 and pad 3 using the other two probes. This process is then repeated by applying a negative magnetic field. The polarisation of the measured  $V_h$  indicates whether the majority carriers are holes (p-type) or electrons (n-type). Equation 5.6 demonstrates how to calculate hall coefficient using the measured  $V_h$ , applied current I and magnetic field B.

$$R_h = \frac{V_h}{IB} \tag{5.6}$$

With hall coefficient, the carrier concentration and mobility can be calculated as following equation, where  $n_s$  is the 2D carrier concentration, q is the unit charge, and  $\mu$  is the carrier mobility.

$$n_s = \frac{1}{qR_h} \tag{5.7}$$

$$\mu = \frac{R_h}{R_S} \tag{5.8}$$

Hall measurement is crucial for characterizing the hydrogen terminated diamond surface. It is used to measure the surface conductivity degradation after completing the fabrication process. Additionally, mobility measurement contribute to the understanding of charge transport under low electric fields.

#### 5.2.3 Field Effect Transistor (FET)

To understand the quality of transistor on hydrogen terminated diamond surface, it's necessary to carry out DC measurements on the devices. The measurement includes the output characteristics measurement, known as  $I_d - V_d$  curve, and transfer characteristics measurement, known as  $I_d - V_g$  curve.



Figure 5.15: Simplified FET DC characteristics graph.

#### **Output Characteristics**



Figure 5.16: Example FET output characteristics.

Three probes are used to perform DC measurements with the source grounded. In the output characteristic, known as  $I_dV_d$  graph, voltage is applied and swept between the source and drain. The bias applied between the gate and source is stepped after each sweep between the source and drain. This measurement is used to obtain the maximum drain current  $I_{dmax}$ .



Figure 5.17: Simplified FET DC characteristics for  $R_{on}$ 

In addition, DC measurement can provide the on-resistance  $R_{on}$ . Figure 5.17 demonstrates all the components of resistance included in the on-resistance. As previously discussed, *TLM* measurement provide information about the contact resistance  $R_C$ . According to the measurement result from *VDP* measurement, one can calculate the resistance between the source-gate channel and gate-drain channel,  $R_{ch}$ . When the device operates in the linear region, the on resistance can be calculated by Ohm's Law. With this information, the resistance under the gate,  $R_G$ , at a particular gate bias can be extracted.

$$R_{on} = 2R_C + 2R_{Ch} + R_G \tag{5.9}$$

One more important piece of information that can be extracted from output measurement is offstate breakdown voltage. To determine the breakdown voltage, the gate channel is held in the "off" state. In the case of hydrogen-terminated diamond devices, a positive bias is applied on the

gate. Then the voltage applied on the drain is swept from low to high until there is a significant increase in current, at which point the voltage is referred as the breakdown voltage.

#### **Transfer Characteristics**

Transfer characteristics  $(I_d - V_g)$  are obtained by sweeping voltage between the source and gate while incrementally adjusting the bias applied to the drain terminal. Transconductance  $g_m$  is an important parameter of a FET device, representing the change in drain current  $(I_{ds})$  at a given drain voltage  $(V_{ds})$  due to a change in gate voltage  $(V_{gs})$ . [30] It reflects the efficiency of changing drain current with respect to gate voltage under a given drain voltage.



Figure 5.18: Example FET transfer characteristics.

Figure 5.18 shows the transfer characteristics, where the green plot represents the transconductance. In an ideal situation, where the parasitic resistances are neglected, intrinsic transconductance can be described as below,

$$g_m^* = \left(\frac{dI_{ds}}{dV_{gs}}\right)_{V_{ds}} \tag{5.10}$$

However, if we take into account the voltage drop between the source and gate contacts, extrinsic transconductance can be defined as follows,

$$g_m = \frac{g_m^*}{1 + g_m^* R_s}$$
(5.11)

In equation 5.11,  $R_s$  represents the resistance between the source ohmic metal and channel region under the gate, and therefore comprises both the source ohmic contact resistance and any potential access resistance associated with the channel between the source and gate contacts,  $R_{Ch}$ . As discussed in Chapter Two and outlined in equation 2.4, we know that the drain current  $I_{ds}$  is a function of the channel's width-to-length ratio (W/L) and carrier's mobility ( $\mu_p$ ). Hence, high carrier mobility and width-to-length ratio contribute to a high transconductance.

Another important parameter extracted from the transfer characteristic is the threshold voltage  $(V_{th})$ , which is the minimum voltage required to maintain the conductive channel between the source and drain terminals. There are different methods to extract the threshold voltage. One method involves using peak transconductance to determine  $V_{th}$ . In this approach, a gate voltage is identified when the transconductance reaches its peak, and this voltage is then used to find the intersection point between the specific gate voltage and the  $I_d - V_g$  curve. The tangent line at this point is used to determine the threshold voltage. The second method involves plotting the  $I_d - V_g$  curve on a logarithmic scale. An off-state current is identified, which is then used to extract the threshold voltage. In this work, the second method was used to extract the threshold voltage.

## 5.3 Chapter Summary

This chapter provides a detailed examination of the equipment used to characterize both topographic and electrical properties, including AFM, SEM, and Surface Profiler, which are employed to obtain surface images, surface roughness measurements, and step heights. Furthermore, we delve into electrical characterization methods such as TLM and VDP, which offer valuable insights into sheet resistance ( $R_S$ ), contact resistance ( $R_C$ ), carrier mobility ( $\mu$ ) and carrier concentration ( $n_p$ ).

In the latter part of this chapter, we introduce FET device measurements. Output characteristics play a significant role in determining parameters such as maximum current, breakdown voltage, and on-resistance. Additionally, we extract the crucial parameter of transconductance  $(g_m)$  and threshold voltage  $(V_{th})$  from transfer characteristics. All these methods collectively contribute to the fabrication of hydrogen-terminated diamond FET devices and enhance our understanding of their behavior.

# Chapter 6

# **Surface Termination Results**

As discussed in the previous chapter, plasma-assisted hydrogen termination is currently the most popular and well-developed method for terminating the diamond surface with hydrogen. However, it is noted that hydrogen plasma can roughen the diamond surface [71]. Additionally, the utilization of hydrogen plasma and a high vacuum chamber can result in escalated costs and restricted accessibility to the capability. High-temperature annealing of diamond in a hydrogen atmosphere has also been reported; however, the process typically requires pure hydrogen, a high vacuum environment, or diamond with large surface area such as diamond powder [73]– [75].

This chapter explores a new method, known as "Spillover", for terminating the diamond surface. The aim of this process is to utilise a lower-cost and more accessible method for terminating the surface of the diamond wafer with hydrogen, without the need of a high vacuum chamber or pure hydrogen.

## 6.1 Initialisation experiment

Single polished CVD diamond wafer provided by Element Six with a (100) surface orientation and dimensions of  $4.7 \times 4.7 \text{ mm}$  were used. To remove any potential hydrocarbon and oxygenated compounds from the sample surface, the samples were immersed in freshly prepared aqua regia ( $HNO_3 : HCl$  in a 1:1 ratio) and boiled on a hotplate for 10 minutes. Subsequently, a boiled solution consisting of a mixture of  $H_2SO_4 : HNO_3$  in a 3:1 ratio was employed for oxidation to achieve a fully oxygen-terminated diamond surface and to remove any other metal contamination.

The sheet resistivity of samples was measured using a Nanometrics HL5500PC Hall Effect measurement system. Four probes made direct contact with the diamond surface. After the

acid cleaning step described above, the measurement four probes hall results indicated that all samples surface were insulating, suggesting that the samples are most likely oxygen terminated as the second oxidation step of the acid cleaning process mainly designed for surface oxygen termination.

Subsequently, PLASSYS II was employed for the deposition of platinum (Pt) onto the diamond samples, with a thickness of 100 nm and a deposition rate of 1 nm/s. To ensure that only one edge of the sample was covered with metal, a cleaned silicon shadow mask was utilized. Figure 6.1 illustrates the method used for mounting the samples on the holder.



Figure 6.1: Image of masking the samples for metal deposition.

The spillover process was conducted in the School of Chemistry at the University of Glasgow. Prior to initiating the spillover experiment, the samples underwent degassing in a vacuum at  $1 \times 10^{-3}$  mbar. The temperature of the furnace was then gradually increased to either 800 °C or 900 °C at a rate of 5 °C/min and held at that temperature for 30 minutes. This degassing process was implemented to attempt to cleanse the diamond samples' surface and desorb all oxygen termination from the surface, leaving free dangling carbon bonds. It has been reported that the oxygen termination on diamond surface desorb at 600°C under vacuum (1 × 10<sup>-3</sup> mbar) and 900°C for hydrogen termination [92].

Subsequently, the furnace temperature was then reduced to 500 °C at a rate of 5 °C/*min*. Once the temperature stabilized, a mixture consisting of 5% of  $H_2$  and 95% of Ar was introduced into the furnace to terminate the diamond surface. The termination process varied in duration, followed by the cooling of the furnace to room temperature.

#### Results

The first experiment for sample one was conducted with the following parameters: 800 °C for 30 minutes for the degassing step, followed by 500 °C for 4 hours for the Spillover step. After the Spillover, the 4 probes hall measurement shows that only the area near the metal strip is

conducting. However the sheet resistance  $(3.6 \times 10^5 \Omega/\Box)$  was observed to be one order of magnitude higher than that typically reported for diamond hydrogen terminated by hydrogen plasma.

Sample	Degassing (°C)	Duration	Spillover (°C)	Duration	$R_s(\Omega/\Box)$
Sample1	800	0.5	500	4	$3.6 \times 10^{5}$
Sample1	800	2	500	4	$4 \times 10^4$
Sample1	900	2	500	12	$1.6 \times 10^{4}$
Sample2	800	0.5	500	4	$> 2 \times 10^{6}$
Sample2	900	2	500	12	$1.5  imes 10^4$
Sample3	800	1	500	4	$> 2 \times 10^6$
Sample3	×	×	500	12	$> 2 \times 10^6$

Table 6.1: Experiment parameters of three different samples.

In the second experiment, the degassing duration was extended from 30 minutes to 2 hours, while all other parameters remained the same as the first experiment. The 4 probes hall kit measurement indicated that the sheet resistance  $(4 \times 10^4 \ \Omega/\Box)$  was one order of magnitude lower than in the first experiment.

For the last experiment, the degassing temperature was increased to 900 °C for 2 hours, and the termination step was prolonged to 12 hours. Consequently, the sheet resistance dropped to  $1.6 \times 10^4 \Omega/\Box$ . However, it was noted that the conductivity on the surface was not continuous.

Two sets of experiments were conducted on sample two. 800 °C 30 minutes degassing and  $500^{\circ}\text{C}$  4 hours Spillover was conducted in the first set of experiments. The 4 probes Hall measurement indicated that the surface remained insulating after the experiment.

For the second set of experiments, the degassing temperature and duration were increased from 800°C 30 minutes to 900°C 2 hours, and the Spillover duration was extended from 4 hours to 12 hours. The 4 probes hall kit measurement revealed that the surface became conducting after the spillover, with a sheet resistance of  $1.5 \times 10^4 \Omega/\Box$ .

Two sets of experiments were designed for the third sample. In the first experiment, the degassing process involved heating to 800 °C for 1 hour to desorb the oxygen termination, followed by 500 °C for 4 hours for the hydrogen termination process. However, the Hall measurement showed that the surface remained insulating after the experiment.

In the second experiment for the third sample, the degassing process was omitted, and a 12-hour termination step at 500 °C was utilized to terminate the surface. Nevertheless, the surface remained insulating after the experiment.

The three sets of experiments conducted on sample one demonstrate that a degassing process

#### CHAPTER 6. SURFACE TERMINATION RESULTS

involving higher temperature (900  $^{\circ}$ C) and longer duration (2 hours) leads to an improvement in surface conductivity by one order of magnitude. Furthermore, prolonging the spillover duration also contributes to enhancing termination results.

The two sets of experiments carried out on sample two, repeating the first and the last sets of experiments conducted on sample one, confirm that a degassing temperature of 900 °C for 2 hours and a 12-hour spillover with hydrogen gas inlet are the optimal parameters for hydrogen-terminating the diamond surface. This observation is assumed to be attributed to the increased efficiency of oxygen desorption at 900 °C. However, more study in details need to be done to confirm the assumption.

In the first set of experiments on sample three, only the degassing duration was extended. The insulation of the diamond surface after the spillover indicates that the temperature used for degassing is more critical than the duration. Additionally, the last set of experiments conducted without a degassing process indicates the necessity of the degassing process for the subsequent termination process. It is assumed that this is because the degassing process desorbs the oxygen termination, exposing free dangling carbon bonds that can bond with hydrogen atoms during the spillover process.

Overall, these experiments confirm the necessity of the degassing process and explore the optimal parameters for both the degassing and spillover processes. Two additional sets of experiments were designed and completed to verify the efficacy of the spillover process.

# 6.2 Spillover with optimal parameters

As both Pt and Pd have been well studied as catalytic metals for the spillover process [131], [132], in this experiment, both metals were introduced as new variables to explore whether different catalytic metals affect the hydrogen termination of the diamond surface.

Three polished CVD diamond samples with surface roughness between 1nm to 2nm, designated as SC16, SC22 and SC10, were provided by Element Six, with a (100) surface orientation and dimensions of  $4.7 \times 4.7$  mm. The same acid and solvent cleaning process employed in the initialization experiment was carried out to clean and oxidize the diamond surface. Subsequently, 4 probes Hall kit measurements were conducted to confirm that all sample surfaces were insulating.

Next, the contact angle of the surface was measured using the KRUSS DSA30 Drop Shape Analyser to assess surface wettability. It has been reported that an oxygen-terminated diamond surface can transition from hydrophilicity to hydrophobicity with surface termination changing to hydrogen [133]. The water dose was optimized at  $0.5\mu l$  to achieve the highest accuracy of the

measurement. Following this measurement, a methanol cleaning step was performed to remove any potential contamination residues.

PLASSYS II was utilized for the deposition of platinum (Pt) with a thickness of 100 nm on sample SC22, while PLASSYS IV was employed for the deposition of 100 nm of palladium (Pd) on sample SC16. The deposition rates for Pd and Pt were 0.3 nm and 1 nm per second, respectively. Pd and Pt act as catalyst to split hydrogen molecular into atoms which are expected to connect with the free dangling carbon bonds to hydrogen terminate the diamond surface [86], [134]. Sample SC10 was set as a blank control experiment where no metal was deposited on the sample surface. 5 minutes ultrasound alcohol (IPA) followed with a 5 minutes ultrasound acetone cleaning was conducted after the metal deposition to remove potential contamination induced by the holder and silicon shadow mask. Subsequently, the sample surface roughness was measured by atomic force microscopy (AFM).



Figure 6.2: Diamond sample were split into 25 areas and 4 probes hall measurement is carried in each area following Spillover.

Similar to previous experiments, the spillover process was conducted in the School of Chemistry. Prior to the start of the spillover, both samples were degassed in a vacuum at  $1 \times 10^{-3}$  mbar. The temperature of the furnace was then raised to 900 °C at a rate of 5 °C per minute and held at 900 °C for 2 hours.

Subsequently, the furnace temperature was reduced to 500°C at a rate of 5°C per minute. Once the temperature stabilized, a mixture consisting of 5%  $H_2$  and 95% Ar was introduced into the furnace to terminate the diamond surface. The termination process lasted for 12 hours, followed by the cooling of the furnace to room temperature.

A series of measurements were performed on the surface to characterize the samples. To investi-

gate whether the surface conductivity changes with increasing distance between the sample and the metal strip, the samples were divided into 25 areas, and Hall measurements were conducted on each small area of the diamond surface. Additionally, atomic force microscopy (AFM) was utilized to characterize the surface roughness, and contact angle measurements were performed to confirm that the surface termination had been altered.



#### **Four Probes Sheet Resistance Measurement**

Table 6.2: Sheet resistance of SC16 after spillover (100nm of Pd strip).



Table 6.3: Sheet resistance of SC22 after spillover (100nm of Pt metal strip).

The initial results we are discussing pertain to the Hall measurements. Prior to the spillover process, the surfaces of both samples were found to be insulating. However, after the experi-

ment, both samples exhibited surface conductivity similar to that reported for sheet resistance of plasma hydrogen-terminated diamond i.e.  $\approx 10 K\Omega/\Box$  [69], [135].

Sample SC16, with palladium (Pd) as the catalyst, demonstrated an average conductivity of 13.2  $k\Omega/\Box$ . Notably, this sheet resistance was slightly lower than that of sample SC22, which utilized platinum (Pt) as the catalyst metal, showing an average resistance of 16.9  $k\Omega/\Box$ .

As detailed in the experimental section, the surfaces of all three samples were partitioned into 25 smaller sections, and 4 probes Hall kit measurements were conducted on each individual segment. Table 6.2 and 6.3 present the sheet resistance of sample SC16 and SC22 following the spillover process. The colour shading of each small section represents the magnitude of the sheet resistance in that area, with darker colours indicating higher sheet resistance.

It is evident that, generally, the colour depth of SC16 is lighter than that of SC22, indicative of lower sheet resistance in SC16. Additionally, the more uniform colour shading on SC16 suggests that its sheet resistance distribution is more homogeneous, although SC22 exhibits a few sections with sheet resistance around  $11 k\Omega/\Box$ .

Furthermore, regarding the surface resistance of the two samples, we did not observe significant changes in sheet resistance gradients with increasing distance between the sampling sections and the metal catalyst. In the table, Row A is adjacent to the metal strip, while Row E represents the furthest section from the strip. To some extent, this suggests that the spillover occurring on the diamond surface is unaffected by the distance between the metal and the sampling point, at least for the sample dimensions inspected. Karim et al. reported that hydrogen spillover is influenced by the substrate material; if the surface is reducible, the spillover effects effectively reduce the oxide without being affected by the distance [86]. In the context of this study, we hypothesize that the 900 °C degassing process desorbs the oxygen termination on the diamond surface, leaving free dangling carbon bonds, thereby rendering the surface reducible. Consequently, the sheet resistance shows no significant relationship to the distance from the catalyst metal strip.

The blank control sample, SC10, without catalyst metal on the surface shows no measurable surface conductivity after the same Spillover process as the other samples with catalyst metal. It is assumed that the splitting of hydrogen molecular by catalyst metal, Pt and Pd, contributing to the hydrogen termination of the diamond surface. Which means, in this work, with low concentration of hydrogen gas as the hydrogen sources, catalyst metal plays crucial roles in terms of hydrogen termination.


Table 6.4: Sheet resistance of SC10 after spillover (Blank control group).

#### **AFM and Contact angle Measurement**

Atomic force microscopy (AFM) was employed to characterize the surface roughness before and after the spillover process. The surface roughness of sample SC16 is slightly lower, and the roughness of sample SC22 slightly increased. Blank control sample SC10 shows a very consistent surface roughness before and after the hydrogen termination. Considering the reported surface roughness changes from 0.3nm to 1.6nm before and after the hydrogen termination with hydrogen plasma treatment[71] and potential measurement errors caused by different measurement spots on the sample surface, those roughness changes in this work is very low.

Measurement	Before (nm)	After (nm)		
R1	1.98	1.51		
R2	2.08	1.55		
R3	1.55	1.39		

Table 6.5: Surface roughness of SC16 (Pd) before and after spillover.

Measurement	Before (nm)	After (nm)		
R1	1.69	1.61		
R2	0.74	N/A		
R3	1.17	1.47		

Table 6.6: Surface roughness of SC22 (Pt) before and after spillover.

The surface contact angle of sample SC16 increased from  $35.5^{\circ}$  to  $65^{\circ}$ , while that of sample SC22 increased from  $42.5^{\circ}$  to  $90^{\circ}$  after the spillover process. This indicates that both samples

Measurement	Before (nm)	After (nm)	
R1	1.08	1.01	
R2	1.22	1.25	
R3	1.14	0.9	

Table 6.7: Surface roughness of SC10 (control) before and after spillover.



Figure 6.3: Contact angle measurement of SC16 before (left) and after (after) the spillover.



Figure 6.4: Contact angle measurement of SC22 before (left) and after (after) the spillover.



Figure 6.5: Contact angle measurement of SC10 before (left) and after (after) the spillover.

became more hydrophobic after the spillover. This observation aligns with literature findings suggesting that the wettability of an oxygen-terminated diamond surface is higher than that of a hydrogen-terminated diamond surface [136]. The contact angle measurement of blank control sample shows the contact angle keeps the same before( $36.2^{\circ}$ ) and after( $38.2^{\circ}$ ) the Spillover process.

Overall, these results indicate that SC16 and SC22 underwent hydrogen termination after the spillover process, although parameters may need further optimization to achieve a more uniform surface termination. Comparing with the results of the blank control sample S10 whose surface is insulating before and after the Spillover, the catalyst metal strip is playing crucial roles by splitting the hydrogen molecular into atoms in this work thus contribute to the Spillover.

## 6.3 Chapter Summary

This chapter presents the experiments conducted for surface hydrogen termination using the Spillover-based process. Seven sets of experiments were designed in the initialization phase to explore various parameters, including degassing temperature, duration, and spillover duration. The results highlight the necessity of a 900 °C two-hour degassing process to desorb oxygen termination on the diamond surface, thereby exposing free dangling carbon bonds. Subsequently, these bonds can connect with hydrogen atoms split by the catalyst metal, platinum or palladium, leading to hydrogen termination of the diamond surface.

The optimal parameters identified in the initialization phase were then applied to two samples, employing Pd and Pt as catalyst metals, respectively. Both samples exhibited surface conductivity after the spillover, with increased hydrophobicity indicating a change in surface termination. AFM results for sample SC16 and SC12 revealed minimal change in surface roughness posttermination. Blank control sample SC10 was carried out to verify the existence of the catalyst metal contribute to the surface termination by splitting the hydrogen into atoms.

In summary, a method for hydrogen termination of diamond surfaces using the thermal spillover process has been explored. The lowest achieved sheet resistance was  $13.2 k\Omega/\Box$  with *Pd* as catalyst metal. These results are consistent with literature recorded values for plasma hydrogen terminated diamond. Additionally, the surface of SC16 exhibited minimal roughening after the spillover process, highlighting an advantage of using this method for diamond surface termination over plasma methods. This work appears to be the first study of the thermal hydrogen termination on diamond wafer with low concentration of hydrogen and temperature below 900 °C.

# **Chapter 7**

# **Alternative Fabrication Results**

In chapter 2, we established that the hydrogen-terminated diamond surface is highly susceptible to damage during fabrication and processing. Even low-power plasmas, such as oxygen ashing, can disrupt or replace this termination. This poses a significant challenge in the fabrication of devices on diamond surfaces, as oxygen plasma is commonly employed for the removal of resist residue post resist development. Furthermore, ambient environmental factors, including elevated temperatures, can variably desorb air adsorbates on the hydrogen-terminated diamond surface, which are required to instigate transfer doping in air-exposed H-diamond.

To safeguard the surface termination against damage during the fabrication process, gold is frequently utilised as a sacrificial layer. This chapter explores a few alternative fabrication methods to replace the gold sacrificial layer approach. Lift-off resist was investigated as protective layers in lieu of gold.

Another strategy to preserve the surface termination involves the avoidance of oxygen plasma usage. Consequently, hydrogen plasma is explored as an alternative method for resist residue removal.

Furthermore, the latter section of this chapter delves into the utilisation of Pd as a replacement for Au in contact metal applications to establish ohmic contact. As previously discussed, Pd has been documented to form ohmic contacts as Au. This segment of the experiment employs the Transmission Line Model (TLM) method to investigate the viability of substituting gold with palladium for the formation of ohmic contacts.

## 7.1 Lift-off Resist as protective layer

Gold serves as a sacrificial layer in the fabrication of devices on hydrogen-terminated diamond surfaces, playing three pivotal roles. Firstly, it shields the hydrogen termination throughout the fabrication procedure. Secondly, the sacrificial layer serves as an ohmic contact metal once the gate level is established. Following gate level development, gold undergoes over etching via wet etch to create the channel and expose the area for subsequent gate deposition. The last acts as a charge conduction layer if the process involves e-beam lithography. This method offers the advantage of safeguarding the surface from environmental damage. However, it presents drawbacks. Wet etching for gate window opening and ohmic contact formation result in a self-aligned gate level, positioning the gate centrally within the source-drain channel. Additionally, wet etching yields less controlled edge wall profiles for ohmic contact, potentially increasing contact area and reducing channel length accuracy.

Lift-off resist (LOR) is a UV and electron beam-insensitive resist type, susceptible to TMAH etching. Control over wet etch profile is achieved through spinning rate, baking time, and developer concentration adjustments [137]. LOR exhibits good adhesion to various materials and compatibility with both photo and e-beam resists, rendering it ideal for bi-layer lift-off processes. Controlled wet etch profiles enable undercutting in the range of 1 to 2  $\mu m$ . Moreover, minimal residue post-wet etch may obviate the need for ashing or descum processes [138]. Thus, exposure of the hydrogen surface termination to oxygen plasma can be circumvented. Utilising LOR as a lift-off layer facilitates an additive process for ohmic metal, enabling the formation of ohmic contacts with precisely controlled sidewalls. Additionally, as discussed in the fabrication chapter, the gold sacrificial layer forms the final ohmic pad by wet etching in the middle of the source and drain. With the use of the LOR process, ohmic pads and gates are defined through two separate steps, enabling the adjustment of the gate position. This flexibility allows for precise control over the gate alignment relative to the source and drain, which is beneficial for optimizing device performance and characteristics.

#### **TLM and VDP structure**

Figure 7.1 depicts the simplified fabrication steps for TLM and VDP structures on hydrogenterminated diamond utilising LOR and e-beam resist as a bi-layer lift-off method. Two single polished hydrogen terminated samples A and B with similar sheet resistance were used for the fabrication. Initial cleaning of hydrogen-terminated single polished diamond samples was conducted using acetone and IPA to eliminate any potential surface contamination. Subsequently, Hall Kit measurements were performed to assess sheet resistance.



Figure 7.1: Fabrication steps of LOR method (Left) and gold sacrificial layer method(Right). The ohmic pad in LOR method is formed by lift-off method. The ohmic pad in gold sacrificial layer method is formed by the wet etch of the gold deposited in the first step. Former method results in a metal contacts with better profiles, and the later on gate fabrication is not limited to the ohmic metal steps. The gold sacrificial form the ohmic pad by over-etching the gold sacrificial layer which leads to a rough profile. In addition, the opened windows for wet etch is also used for gate deposition, which leads to a self-aligned gate level.

#### CHAPTER 7. ALTERNATIVE FABRICATION RESULTS

A layer of LOR 3A was spun (6000 rpm/s for 45 seconds) and baked (180 °C for 5 minutes) onto the surface as the first resist layer, followed by the spun (5000 rpm/s for 60 seconds) and bake (180 °C for 2 minutes) of e-beam resist atop the LOR. Following e-beam lithography, the e-beam resist undergoes development with 1:2 IPA:MIBK developer at 23 °C for 30 seconds. The sample was then subjected to oxygen plasma ashing for 1 minute at 80 W power to eliminate PMMA resist residue. A 40-second rinse in CD-26 (2.5% TMAH) was employed to etch (develop) the LOR resist beneath the developed e-beam resist. For isolation purposes, the sample was exposed to oxygen plasma once more to remove the exposed hydrogen termination.

For the ohmic metal and bond pad metal steps, Plassys 2 was utilised to deposit Au (ohmic metal 80nm) or Ti(10 nm)/Au(80 nm) (bond pad metal stack). Subsequently, an overnight rinse in SVC-14 in a 50 °C water bath was employed to strip off the resist and lift-off any unwanted metal.

The entire fabrication process encompasses the marker level, isolation, ohmic pad, and bond pad metal deposition stages. Following fabrication, the TLM VDP structure was measured using a probe station and Hall kit.





Figure 7.2 displays the optical and SEM images of ohmic pads formed using the gold wet etch and lift-off processes, respectively. It is evident that the edges of the lift-off ohmic pads are smoother than those of the gold wet-etched ones. This difference is particularly pronounced in the SEM image, where the magnification of the lift-off sample, at 35 k, is nearly double that of the wet-etched sample. Despite the higher magnification, the edge of the lift-off sample remains smoother. In terms of the roughness of the ohmic pad edge, the lift-off method evidently yields superior results.



Figure 7.3: Optical image of VDP structure for Hall measuremnt.

Hall Kit measurements are conducted both before and after fabrication. Prior to device fabrication, four probes make direct contact with the sample surfaces. Sample A exhibits a sheet resistance of  $1.8 \times 10^4 \Omega/\Box$ . Conversely, the sheet resistance of sample B measure is  $1.7 \times 10^4 \Omega/\Box$ 

Sample	Status	$R_s(\Omega/\Box)$	Status	$R_s(\Omega/\Box)$
Sample A	Before	$1.8 \times 10^{4}$	After	$2.4 \times 10^{4}$
Sample B	Before	$1.7 \times 10^{4}$	After	$6.8 \times 10^{4}$

Table 7.1: VDP measurement of Sample A and B before and after the fabrication.

Following fabrication, VDP structures were measured using the Hall Kit. The sheet resistance of samples A and B increases to  $2.4 \times 10^4 \Omega/\Box$  and  $6.8 \times 10^4 \Omega/\Box$ , respectively. This increase in sheet resistance suggests damage to the hydrogen termination or the presence of resist residue. Particularly noteworthy is the nearly threefold increase in sheet resistance for sample B.



Figure 7.4: Optical image of TLM structure.

The TLM structure is assessed post-fabrication to derive the contact resistance and sheet resistance. Figure 7.5 presents a set of TLM structure measurement results. It is evident that the structure does not exhibit ohmic contact behaviour across the inspected voltage range, where the I-V curve should ideally be a straight line, with the slope of the plot reflecting the resistance of the contact.



Figure 7.5: TLM result of sample A fabricated with LOR (Applied voltage 0 - 5 V).



Figure 7.6: TLM result of sample A fabricated with LOR below 0.5V.

The observed non-linear behaviour in this sample suggests the potential presence of resist residue on the sample surface. Although the curve is not linear within the 0 to 5 volts range, a small linear region exists below 0.5 volts. To derive the contact resistance and sheet resistance from the TLM results, only data below 0.5 volts are utilised for the calculation. Figure 7.6 depicts the data plot below 0.5 volts.

The resistance between each gap is computed, and the value for each gap is utilised to plot a graph, with gap size as the x-axis and resistance as the y-axis. Equation 5.4 elucidates that the intercept represents two times the contact resistance, while the slope corresponds to the sheet resistance over the width of the contact pad. Figure 7.7 illustrates the final plot of TLM structure measurements. The sheet resistance measures  $3.9396 \times 10^4 \Omega/\Box$ , and after normalisation, the contact resistance is  $17.1 \Omega \cdot mm$ . Notably, the contact resistance is larger than typically reported for Au on hydrogen terminated diamond which is  $0.9 - 5 \Omega \cdot mm$  [46], [93], [95]. The sheet resistance from the TLM is higher than that measured from the VDP structure. It is assumed that the influence of resist residue on the sheet resistance might vary across the sample surface.



Figure 7.7: TLM results analyse of Sample A when data extracted below 0.5 V.

Figure 7.8 depicts the measurement of one set of TLM structures on sample B. The observed I-V curve exhibits non-linear behaviour in the measured range, and the current saturates between 1 to 2V, which is similar behaviour to sample A. It is assumed that the non-linear behaviour and saturation current in a designated ohmic contact are caused by the potential resist residue, which leads to poor metal-semiconductor contact between the ohmic pad and the H-diamond surface. The increased sheet resistance measured before and after fabrication with the Hall kit and the sheet resistance (3.9396 × 10<sup>4</sup>  $\Omega/\Box$ ) extracted from sample A contribute to the hypothesis that the existence of the resist residue across the sample surface. The patchy resist residue under the Ohmic pad leads to poor Ohmic contact between the gold and the H-diamond surface. This is one of the reasons for the extremely high contact resistance extracted from the TLM measurement. The saturation of the current when the bias exceeds 1 V might be caused by the limited number of carriers or the low mobility of carriers in the TLM gap. This could result from the resist residue interfering with the H-diamond surface air adsorbates. Equation 7.1 demonstrate the current calculation with the carrier concentration,

$$I = q \cdot p \cdot \mu \cdot E \cdot A \tag{7.1}$$

where q is the elementary charge, p is the hole concentration,  $\mu$  is the hole mobility, E is the electric filed and A is the cross sectional area.



Figure 7.8: TLM result of sample B fabricated with LOR.

The electrical measurements, encompassing TLM and Hall measurements, reveal a notable increase in the sheet resistance of both samples following fabrication. The non-linear I-V curves in the measured range and saturation behaviours is assumed to be caused by the remained resist residue of LOR during the fabrication process. Extracting the contact resistance from sample A yields a value higher than the reported contact resistance between gold and hydrogen-terminated diamond. Taken together, these findings suggest that LOR remained residue on the sample surface, contributing to the observed deviations from expected electrical behaviour.

#### **Resist residue Measurement**

The electrical measurements have provided evidence suggesting the presence of resist residue beneath the ohmic contact. To validate this assertion, additional tests were conducted in this study. The initial test involved spinning LOR resist onto the sample surface and subsequently baking it. Following this, CD26 was employed to etch the resist for the same duration as used in the fabrication process. This procedure was repeated four times on the same sample, and after each iteration, the surface roughness was measured using AFM.

Figure 7.9 illustrates that after the initial spin and wet etch, the surface roughness increased by more than 0.5 nm for both samples. Subsequently, following the first spin, the roughness of both samples exhibited minimal change. This phenomenon may be attributed to the initial spin leaving a thin layer of resist on the surface, thereby resulting in the AFM scan measuring the roughness of the LOR resist rather than the sample surface itself. After this spin, the roughness stabilised at a certain level, as the potentially accumulated thin layer of LOR resist after each spin formed a surface similar to the initial layer of residue.



Figure 7.9: Sample surface roughness after four times of LOR spin and etch.

In accordance with the report by Macintyre et al. [118], another set of experiments was conducted in this study. This experiment involved spinning LOR 3A and PMMA resist onto the cleaned diamond surface, followed by a three-step fabrication process. Initially, a marker level was applied for subsequent alignment purposes. Subsequently, a 20 nm Ti layer and an 80 nm Au layer were deposited onto the sample surface. After lift-off of the marker metal, the sample was spun with the same resist stack. In this step, an array with horizontal lines was patterned onto the resist. To create a surface devoid of resist residue, a 2-minute ashing process was performed after resist development. Instead of stripping off the resist, a second array of vertical lines was patterned onto the same resist stack in the third step. Ashing was omitted in this step to ensure that the resist residue remained on the surface. To mitigate errors caused by etching the diamond with oxygen plasma affecting the experimental results, a silicon wafer was utilized in this experiment.



Figure 7.10: Simplified fabrication flow of resist residue test.(a) Cleaned silicon sample. (b) LOR 3A and PMMA spin and baking. (c) First level of grate writing and develop, potential resist residue in light green. (d) Oxygen plasma engaged to remove the potential residue. (e) After oxygen plasma the second level of grate was completed on the rest of resist left on the sample surface and developed. "Step" between clean surface and resist residue created.

Figure 7.11 presents optical images depicting the sample surface after each step of the experiment. The initial image displays the sample surface subsequent to PMMA development and LOR wet etching, with a horizontal line array patterned onto the surface. The subsequent ashing step is aimed at removing the resist residue from the initially exposed area. The middle image illustrates the surface following the second exposure, with the PMMA on the top subsequently developed. The third image showcases the sample surface post-LOR wet etching by CD26. The



Figure 7.11: Optical image of LOR residue test structure fabrication. The first step of horizontal array written (left), exposure of vertical array (mid), vertical array LOR wet etch (right).



Figure 7.12: 2D and 3D rendering AFM image of area between the first expose and second expose.

red areas denote regions without oxygen ashing. Notably, a junction area is visible, where part of the surface is cleaned (ashed) while the other part retains the condition after LOR wet etching. Subsequently, AFM was employed to measure the junction area.



Figure 7.13: Histogram graph of AFM scan.

The left peak shows the minimum peak depth and the right peak shows the maximum peak depth. The distance between two peaks represents the depth of the measured step.

Figure 7.12 presents the AFM image depicting the scanning results over the junction area between the first exposed area and the second exposed area. The scanning results reveal a step between the two exposed areas. The smallest measured depth size in this study is 7 nm, while the largest step measures 14 nm. This observation is consistent with the hypothesis that wetetched LOR leaves resist residue on the sample surface, and that this residue can be removed by oxygen plasma ashing. Figure 7.13 displays the histogram graph of the depth measurements. The minimum peak depth is 0 nm, while the maximum peak depth is 14.2 nm.

All the results collectively indicate that LOR is not suitable as a protective layer for fabrication on hydrogen-terminated diamond due to the presence of resist residue. Removal of this residue necessitates the use of oxygen plasma, which in turn damages the surface termination.

## 7.2 Hydrogen Plasma resist residue removal

Instead of relying on a protective layer to preserve the surface hydrogen termination from damage caused by oxygen plasma, hydrogen plasma was investigated as an alternative for removing resist residue. Given that the hydrogen termination of diamond surfaces can be achieved through hydrogen plasma treatment, using hydrogen plasma to remove resist residue may prevent damage to the hydrogen surface termination. The initial phase involved establishing a hydrogen plasma etch recipe capable of removing resist while maintaining a well-controlled etching rate.



Figure 7.14: CAD image of e-beam lithography job design for hydrogen plasma etching test.

The experiment entails an e-beam lithography fabrication step to create a series of arrays with different line widths on a cleaned silicon substrate. The silicon sample is first cleaned and then coated with a bi-layer of PMMA resist to establish the same resist condition as that on the diamond sample. Subsequently, the sample was developed in 2:1 MIBK:IPA developer for 30 seconds at 23 °C to develop the exposed resist. Following development, oxygen ashing is conducted to remove any resist residue and ensure a clean profile for subsequent measurement steps.

Figure 7.14 illustrates the pattern for the e-beam lithography job. Three rows and three columns of arrays are written on the sample surface. The patterns in each row share the same architec-

ture, with the line widths of each row being 100  $\mu m$ , 10  $\mu m$ , and 5  $\mu m$  from top to bottom, respectively.

To measure the etch rate of hydrogen plasma, a DEKTEK XT surface profiler is employed to scan across the surface before and after each dry etch step to compare the thickness difference. The ICP180 is utilised as the dry etch tool.

### **DEKTEK XT Measurement Results**

To establish appropriate parameters for resist etching, an initial etching process was conducted with a coil power of 900 W and platen power of 8 W. The hydrogen flow, chamber pressure, and stage temperature were set at 50 sccm, 25 mTorr, and 20 °C, respectively. The duration of the dry etch was set to 2 minutes. Prior to the dry etch, the thickness of pattern A (with 100  $\mu$ m array) was measured at 410 nm, while patterns B and C (with 10  $\mu$ m and 5  $\mu$ m arrays, respectively) measured at 350 nm. The difference in thickness between patterns of different sizes can be attributed to the proximity effect. Smaller features, having a higher density, can cause adjacent features to be overdosed with back-scattered electrons more than larger features, resulting in thinner resist after development.

Following the hydrogen dry etch, the thickness of pattern A decreased to 90 nm, yielding an etching rate of approximately 160 nm/min. Patterns B and C exhibited a thickness reduction to 40 nm, resulting in an etching rate of around 165 nm/min. While the etching rates for different patterns did not significantly differ, they were notably too fast for effective resist residue removal. Consequently, two additional sets of experiments were devised to decelerate the etching rate.

In the first modified parameter set, the coil power, which accelerates the ions in the coil, was adjusted. A series of dry etch tests were conducted with coil powers of 450 W, 300 W, 150 W, and 50 W. The thickness of the patterns was measured before and after the dry etch for each experiment. With a coil power of 450 W, the etching rates for patterns A, B, and C were approximately 99.6 nm/min, 99 nm/min, and 99.5 nm/min, respectively. Subsequently, as the coil power decreased to 300 W, the etching rates for all three patterns remained approximately the same at around 65 nm/min. Further reductions in coil power to 150 W and 50 W resulted in etching rates of 25 nm/min and 5.99 nm/min, respectively.

Figures 7.15, 7.16, and 7.17 illustrate the contact profiler measurement results of the grating structures before and after etching by hydrogen plasma. The coil power applied in this experiment was 50 W, with a duration of 10 minutes. The flow rate, platen power, chamber pressure, and stage temperature were set at 50 sccm, 8 W, 25 mTorr, and 20 °C, respectively. The height of the grating patterns for pattern A decreased from 405 nm to 344 nm, resulting in an etch rate

of 5.675 nm/min. For pattern B, the height decreased from 335.5 nm to 275 nm, giving an etch rate of 6.05 nm/min. Finally, the thickness of pattern C dropped from 330 nm to 267.3 nm, with an etch rate of 6.25 nm/min. The average etch rate with 50 W power was approximately 5.99 nm/min.



Figure 7.15: Resist thickness of pattern A before (left) and after (Right) dry etch with coil power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins.



Figure 7.16: Resist thickness of pattern B before (left) and after (Right) dry etch with coil power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins.



Figure 7.17: Resist thickness of pattern C before (left) and after (Right) dry etch with coil power of 50 W, hydrogen flow of 50 sccm and duration of 10 mins.

Figure 7.18 depicts the changes in etching rate with the decrease in coil power. As previous results shown, the reduction in power leads to a significant decrease in etching rate. Moreover, the etching rate for patterns with different feature sizes remains consistent. In order to further reduce the etching rate, two more sets of experiments were conducted with decreasing flow rates.



Figure 7.18: Hydrogen plasma etching rate of PMMA resist versus coil power change.



Figure 7.19: Hydrogen plasma etching rate of PMMA resist versus hydrogen flow rate change.

Based on the recipe with 50 W coil power, two more sets of experiemnt with decreasing flow rate were conducted. Figure 7.19 illustrates the changes in etching rate with varying hydrogen flow rates. Similar to the previous test, the etching rate remains consistent for different feature sizes. Furthermore, reductions in the flow rate led to decreases in the etching rate from 5.99 nm/min to 4.5 nm/min. For a standard oxygen plasma ashing process used in this work, ashing rate of PMMA resist is around 5 nm/min to 10 nm/min. Thus a hydrogen plasma etching rate of PMMA between 5.99 nm/min and 4.5 nm/min is ideal for the resist residue removal.

Two  $3 \times 3$  mm hydrogen-terminated diamond samples were utilised to test the etching parameters and to verify whether this process preserves the hydrogen termination on the diamond surface. VDP measurements were used to assess the surface conductivity before and after the dry etch. Four probes made direct contact with the sample surface. The sheet resistance of both samples was on the order of  $3 \times 10^4 \Omega/\Box$ . The oxygen plasma ashing process employed in this work aimed to remove resist residue with a thickness of around 10 nm. Thus a recipe with etching rate around 4.5 nm/mm was applied for the trial. The etching trial was conducted for one minute using a recipe with a coil power of 50 W, platen power of 8 W, flow rate of 20 sccm, pressure of 25 mTorr, and table temperature of 20°C. After hydrogen plasma etching, VDP measurements indicated that the sheet resistance of both samples was not measurable.

Despite achieving a well-controlled etching rate in the aforementioned experiments, when applied to hydrogen-terminated diamond samples, the Hall Kit measurements revealed that the diamond surface became not measurable after the dry etch process, contrary to hope of preserving the hydrogen termination on the diamond surface. One potential reason for the loss of surface conductivity may be attributed to the low stage temperature. It has been reported that the surface conductivity of hydrogen-terminated diamond decreases as the dry etch stage temperature decreases [139]. Another study suggests that hydrogen plasma etching conducted at room temperature tends to result in more surface defects and effectively etches off the diamond surface rather than preserving the termination [140]. Considering the findings from both reports, it is plausible that the plasma etching process employed in this study tended to etch off the surface, leading to the induction of an insulating diamond surface.

### 7.3 Pd as contact metal

This section presents the results of utilizing palladium (Pd) as an ohmic contact metal, aimed at investigating the difference between gold (Au) and palladium (Pd) ohmic contacts on hydrogenterminated diamond. As discusses in the chapter 3, Au is suitable to form an ohmic contact on hydrogen terminated diamond due to its large work function (5.1-5.47 eV) that allows the conduction band sits near or below the valence band maximum of hydrogen terminated diamond which allows efficient charge transport between the Au and hydrogen terminated diamond. Pd with a slightly higher work function (5.2-5.6 eV) than Au is energetically favourable to form ohmic contact with hydrogen terminate diamond as gold. A series of experiments were conducted for this purpose. Simple TLM structures were fabricated with either Au or Pd serving as the ohmic metal on two hydrogen-terminated diamond surfaces.



Figure 7.20: Fabrication flow of TLM with Pd or Au as ohmic contact metal.

Two single polished (100) diamond samples underwent cleaning and oxidation through an acid cleaning process as described previously. Subsequently, the samples were hydrogen-terminated using hydrogen plasma in Australia. Gold was then deposited on both samples to serve as a sacrificial layer. A bi-layer resist comprising LOR and S1805 was employed for photolithography. On sample A, 80 nm of Au was utilized as the TLM ohmic contact metal, while 80 nm if Pd was used for the ohmic contact metal on sample B. The TLM gap was set at 3  $\mu m$ , 4  $\mu m$ , 5  $\mu m$ , 6  $\mu m$ , and 7  $\mu m$ . More details in terms of the resist process see Figure 7.20 and appendix A.1



Figure 7.21: Au as ohmic metal TLM measurement results.



Figure 7.22: Pd as ohmic metal TLM measurement results.

Figure 7.21 and Figure 7.22 depict the TLM measurement results of the fabrication using Pd or Au as the ohmic metal. The sample employing Au as the ohmic metal exhibits a contact resistance of 7.52  $\Omega \cdot mm$  and a sheet resistance of 9.6  $k\Omega/\Box$ . The contact resistance of the TLM with Au as ohmic contact is within the range of literature reported ones, which is averagely around 5  $\Omega \cdot mm$  [46], [93], [95].

Conversely, the sample fabricated with Pd as the ohmic metal demonstrates a contact resistance of  $3.82 \ \Omega \cdot mm$ , and a sheet resistance of  $7.64 \ k\Omega/\Box$ . This contact resistance is slightly higher than the literature recorded  $0.4 \ \Omega \cdot mm$  to  $2.1 \ \Omega \cdot mm$  [97], [141]of contact resistance formed between the Pd and hydrogen terminated diamond. However, this slight difference may be caused by the difference in the surface conditions, such as the sheet resistance. It has been observed that contact resistance between the hydrogen terminated diamond and Pd can be as high as 43  $\Omega \cdot mm$  when the sheet resistance of the hydrogen terminated diamond reaches 75 k $\Omega/\Box$ [142]. In conclusion, in this work, the contact resistance of Pd-formed ohmic contacts is lower compared to that of Au contacts when both diamond samples exhibits similar sheet resistance.

## 7.4 Chapter Summary

This chapter primarily discusses the results obtained from various non-traditional fabrication methods aimed at replacing the Au sacrificial layer commonly used in the fabrication of H-diamond FETs. LOR was investigated as an alternative to Au to protect the surface termination from damage caused by oxygen plasma and ambient environmental factors. However, the non-linear I-V characteristics observed in TLMs fabricated with LOR suggest the presence of resist residue. Subsequent AFM scans of the cross-grating structures on the diamond surface revealed that the resist residue could reach up to 14 nm.

The potential of using hydrogen plasma as a replacement for oxygen plasma to remove resist residue while preserving the surface termination was explored. Despite setting parameters that slowed the resist etching rate to 4.53 nm/min, surface conductivity measurements revealed that the surface had become insulating. The utilization of a low stage temperature in this process may have resulted in the hydrogen plasma etching off the diamond surface rather than just removind resist residue, thereby damaging the conductivity. Further optimisation of this process in future studies may yield better results however.

In the final part of the chapter, the ohmic contacts to H-daimond formed by Au and Pd were investigated and compared. The TLM results demonstrated that Pd forms an ohmic contact on the hydrogen-terminated diamond surface, with the contact resistance being lower than that of Au ohmic contact.

# **Chapter 8**

## **Hydrogen terminated Diamond MESFETs**

As discussed in the previous chapter, metals with low work function can be utilised on hydrogenterminated diamond to form a Schottky contact. Aluminium (Al), lead (Pb), zinc (Zn), and titanium (Ti) have been reported to form Schottky contacts on hydrogen-terminated diamond[98]. Metals with high work function, such as gold (Au) and palladium (Pd), can be employed to form ohmic contacts on hydrogen-terminated diamond [97].

This chapter discusses the electrical measurement results of hydrogen-terminated diamond metalsemiconductor field-effect transistors (MESFETs) utilising an Al/Pt/Au metal stack for the gate and Au as the ohmic contact metal. Van der Pauw (VDP) and transmission line model (TLM) measurements yield results such as contact resistance ( $R_c$ ), sheet resistance ( $R_s$ ), carrier concentration ( $n_p$ ), and carrier mobility ( $\mu$ ). MESFET measurements will illustrate device properties such as maximum drain current ( $I_{dmax}$ ), transconductance (g), threshold voltage ( $V_{th}$ ), and breakdown voltage.

The devices reported in this chapter were initially measured at room temperature. To assess the stability of the devices in ambient air, measurements were repeated after two weeks of exposure. Subsequently, atomic layer deposition (ALD) of  $Al_2O_3$  was performed on the device surface to serve as encapsulation and electron acceptor. Further device measurements with  $Al_2O_3$  were conducted at high temperature (400K) to explore thermal stability.

## 8.1 VDP and TLM structures with Au as contact metal

A  $4.5 \times 4.5$  mm double-polished diamond sample was subjected to a two-step acid cleaning process, as detailed in the previous chapter(see appendix A.2), to remove potential contamination and residual fabrication structures. When the fabrication process commenced, the majority of samples were recycled, necessitating polishing to reduce surface roughness to below 1 nm.

Subsequently, the sample underwent hydrogen termination in Australia using hydrogen plasma. Upon its return, the sample was subjected to 2 to 3 days of exposure to ambient air to ensure stabilization of surface conductivity before fabrication commenced. The sample, sealed in a membrane sample box after the termination step, underwent an air exposure process, which facilitated the establishment of a layer of air adsorbates, thereby enhancing surface conductivity. Surface conductivity was monitored throughout this process using a VDP measurement process.

Following the stabilization of surface conductivity, a 50 nm gold sacrificial layer was deposited onto the surface by electron-ebeam evaporation to protect the hydrogen termination and function as the ohmic contact for devices. This step minimises the likelihood of exposure of the ohmic area and gate channel to ambient air and other steps, such as plasma ashing, which could potentially damage the surface termination. As the existed photo-lithography mask was designed to use gold sacrificial layer and ohmic metal, the device in this chapter used gold rather than palladium as ohmic contact. The fabrication process of the device has been elucidated in Chapter 4, with Figure 4.19 providing a detailed illustration of each step.



Figure 8.1: Optical image of sample with completed structures and devices fabricated with gold as sacrificial layer.

Following fabrication, a transmission line model (TLM) measurement was conducted, and Figure 8.2 presents the data extracted from this measurement. The width of the TLM structure is 75  $\mu$ m, with a slope of 381.6  $\Omega/\mu m$ . Calculating from this plot, the sheet resistance is determined to be 28.62 k $\Omega/\Box$ , consistent with the VDP-measured sheet resistance. Contact resistance can be derived from the intercept of this plot, yielding a value of 4.14  $\Omega \cdot mm$ . This contact resistance value from the reported range for gold contacts on hydrogen-terminated diamond, typically falling between 4 to 5  $\Omega \cdot mm$  [95], [143].



Figure 8.2: TLM results with sample exposed to air, day 1 and day 7 after ALD  $Al_2O_3$  deposition.

In Chapter 4, experimental results of atomic layer deposition (ALD) of  $Al_2O_3$  on hydrogenterminated diamond were discussed (Table 4.2). VDP measurements revealed that thermal deposition of ALD  $Al_2O_3$  improved surface conductivity from 9 k $\Omega/\Box$  to 6 k $\Omega/\Box$ . This suggests that ALD  $Al_2O_3$  enhances surface conductivity. Consequently, a 15 nm layer of ALD  $Al_2O_3$  was deposited on the sample surface. Subsequent measurements were performed on both the TLM structure and the MESFETs, the latter to be discussed in the following section.

The blue line in Figure 8.2 represents the plot extracted from TLM measurement results. With a slope of 339.64 and an intercept of 281.77, the sheet resistance is calculated to be 25.425  $k\Omega/\Box$  and the contact resistance to be 10.5375  $\Omega \cdot mm$ . The sheet resistance decreased by more than 3  $k\Omega/\Box$  after  $Al_2O_3$  deposition, consistent with Chapter 4's findings. However, the doubled contact resistance post-deposition was unexpected, considering the protection of the contact area by the gold ohmic pad.

	Sheet Resistance ( $k\Omega/\Box$ )	Contact Resistance ( $\Omega \cdot mm$		
Air Adsorbates	28.620	4.14		
Thermal ALD $Al_2O_3$	25.425	10.5375		
7 days after $Al_2O_3$	23.4	3.63		

Table 8.1: Sheet resistance and contact resistance after 30 days of air expose.

Subsequent to the sample that is covered with  $Al_2O_3$  being exposed to ambient air for seven days, the TLM was remeasured. The purple dotted line in Figure 8.2 indicates that the sheet

resistance remained relatively stable. However, the contact resistance decreased from 10.537  $\Omega \cdot mm$  to 3.63  $\Omega \cdot mm$ , returning to its value after initial device fabrication. Hypothetically, any airdsorbates between the H-diamond/gold interface was disturbed during the thermal ALD  $Al_2O_3$  deposition. The reaction chamber temperature (200 °C) potentially can disturb the air adsorbates on the diamond surface. The underlying reasons for these changes in contact resistance will be explored further in subsequent measurements. However, this changes is temporary as the contact resistance measured with TLM back to normal seven days after the thermal oxide deposition.



Figure 8.3: Sheet resistance and contact resistance before and after  $Al_2O_3$  and under different temperature.

To assess the thermal stability of the sample following ALD  $Al_2O_3$  coverage, high-temperature TLM measurements were conducted. The sample was placed on a heating stage and sealed under low vacuum conditions. The stage temperature was increased stepwise from room temperature (300 K) to 350 K, 400 K, and 450 K. At each temperature stage, TLM measurements were performed. Additionally, the stage temperature was maintained at each level for 30 minutes, with two measurements conducted at each stage, except for the highest temperature of 450 K, which was measured once for safety.

Figure 8.3 illustrates the sheet resistance and contact resistance of the TLM at various stages and temperatures. The sheet resistance of the sample surface exhibited slight fluctuations, ranging between 24 k $\Omega/\Box$  and 21 k $\Omega/\Box$  from room temperature to 450 K, returning to 22.5 k $\Omega/\Box$  thereafter. This indicates that the surface conductivity of hydrogen-terminated diamond under ALD  $Al_2O_3$  oxide remains relatively stable between room temperature and 450 K. In Crawford

et al report, the exposed to air hydrogen terminated diamond shows sheet resistance drop at around 345 K[144].

However, the contact resistance began to increase at 350 K. With continued temperature elevation, the contact resistance rose from 3.63  $\Omega \cdot mm$  at room temperature to 8.733  $\Omega \cdot mm$  at 350 K, with slight increments observed during each 30-minute temperature hold. The highest contact resistance measured was 35.87  $\Omega \cdot mm$  at 450 K.

A hypothesis posits that residual air adsorbates beneath the gold ohmic pad induce the temperature sensitivity of the hydrogen-terminated diamond. The gold forms an ohmic contact with the hydrogen-terminated diamond. However, the gold and H-diamond surface are just sticked together with weak Van der Wal forces. As the temperature rises, some air adsorbates are disrupted, which affects the contact between the gold and the hydrogen-terminated diamond. This hypothesis requires further investigation in future work. A potential experiment involves depositing gold on the hydrogen-terminated diamond after high-temperature (400 °C) annealing, which is used to remove the air adsorbates. The hypothesis can be confirmed if subsequent I-V testing of the sample at elevated temperatures shows constant contact resistance.

After the high-temperature measurements, the sample was stored at room temperature for 6 months. Subsequent TLM measurements showed that the sheet resistance remained consistent at 24 k $\Omega/\Box$ , consistent with measurements before high-temperature testing and after oxide deposition. The contact resistance, which had increased to 35.87  $\Omega \cdot mm$  at 450 K, decreased to 4.1625  $\Omega \cdot mm$ , matching the value measured after oxide deposition at room temperature. This decrease in contact resistance supports the hypothesis that the increase during high-temperature measurement resulted from residual air adsorbate disruption beneath the gold. Moreover, the temperature caused disruption is restored after the temperature drop back to the room temperature.

Similarly, this disrupt may occur on the hydrogen-terminated diamond surface between two TLM ohmic pads. However, it is assumed that defects potentially formed within the ALD  $Al_2O_3$  layer may act as electron acceptors, enhancing surface conductivity. The ALD  $Al_2O_3$  deposition process at 200°C can partially remove air adsorbates, allowing defect-mediated surface conductivity enhancement to dominate. This explains the stable sheet resistance despite the significant increase in contact resistance. Kevin et al. also observed enhanced thermal stability of surfaces covered with low work function material ( $V_2O_5$ ) at temperatures up to 520 K [144].

Hall kit measurements were conducted before fabrication commenced. Four probes made direct contact with the sample's four corners, yielding a sheet resistance of 35.99 k $\Omega/\Box$ . The higher sheet resistance of this sample, compared to reported values (2-20 k $\Omega/\Box$  [69], [145]), may be attributed to surface defects resulting from the sample recycling process. The higher sheet re-

sistance measured by VDP, compared to TLM-extracted values, may result from the inaccuracy of the VDP measurement. Since this measurement was not conducted with lithographically defined VDP structures on the sample surface, the sheet resistance results might vary from the real values and the values extracted from the TLM structures (28.62  $k\Omega/\Box$ ).

Subsequently, VDP measurements were conducted again after ALD  $Al_2O_3$  deposition with a VDP structures. The results showed a decrease in sheet resistance post- $Al_2O_3$  deposition (24.17 $k\Omega/\Box$ ), consistent with TLM measurements. As this VDP measurement were conducted on properly lithographically defined VDP structure, this result is relatively reliable. The posthigh-temperature VDP measurements indicated the sheet resistance wasn't affected by the high temperature measurements. This results matches the finding concluded from the TLM measurements.

Table 8.2 presents the measurement results of the sample before and after oxide deposition.

	Sheet Resistance ( $k\Omega/\Box$ )		Mobility $(cm^2/V \cdot s)$	Concentration $(cm^2)$	
Air Adsorbates		35.99	NaN	NaN	
ALD $Al_2O_3$		24.17	79.1	$3.26 \times 10^{12}$	
After H	gh	25.7	79.1	$3.07 \times 10^{12}$	
Temperature					

Table 8.2: VDP measurement of the sample.

To conclude this section, both VDP and TLM structures were measured using a Hall Kit and probe station before and after ALD  $Al_2O_3$  deposition. The thermal deposition of ALD  $Al_2O_3$  on the sample surface resulted in improved surface conductivity, as evidenced by the drop in sheet resistance measured by both TLM and VDP structures. A increase of contact resistance after the ALD  $Al_2O_3$  deposition was observed. However, seven days after the oxide deposition, TLM structures were remeasured, and the contact resistance returned to the level observed before the oxide deposition while the sheet resistance remained at a reduced amount.

Subsequently, the sample was subjected to elevated temperatures ranging from 350K to 450K. The sheet resistance remained relatively stable within this temperature range. The observed increase in contact resistance is assumed to be attributed to the potential disruption of air adsorbates under high temperatures, a phenomenon observed after both the thermal oxide deposition and during high-temperature measurements. Subsequent TLM and VDP measurements showed that the sheet resistance and contact resistance returned to its pre-high-temperature level.

This series of measurements confirms the efficacy of thermal ALD  $Al_2O_3$  deposition in improving surface conductivity by 17% and highlights the impact of high temperatures on contact resistance, attributed to the air residue disruption.

This comprehensive analysis provides valuable insights into the thermal stability and surface properties of hydrogen-terminated diamond samples, essential for optimizing device performance and reliability in various applications.

### 8.2 MESFET measurement before and after air exposure

In the previous section, immediate measurements were conducted on the devices post-fabrication, followed by an additional measurement after covering the sample with 15 nm of ALD  $Al_2O_3$ . Both TLM and VDP results indicated an improvement in surface conductivity after  $Al_2O_3$  deposition. This section will delve into the results of room temperature measurements of the air H-Diamond MESFETs.

The gate metal stack for the devices comprises 20 nm of aluminium (in contact with the Hdiamond), 20 nm of platinum, and 80 nm of gold on the top.



Figure 8.4: Output characteristics for  $2\mu m$  MESFETs with H-diamond surface expose to air. Drain current is shown in blue solid line, and gate leakage current is shown in orange dot line.

In the acquired device I-V output characteristics depicted in Figure 8.4, it is evident that for a positive gate-source voltage, the device is turned off. As discussed in the previous chapter, the hydrogen-terminated diamond surface with air adsorbates forms a hole accumulation area, providing a channel for current flow. When a positive voltage is applied to the gate, holes near the surface are repelled, thus closing the channel when the voltage reaches threshold.

Conversely, when a negative voltage is applied to the gate, more holes are attracted to the sur-

face, increasing the charge density with the channel, allowing more current to flow through the channel. Increasing the negative voltage applied to the gate leads to higher drain current. When a fixed negative gate voltage is applied, increasing the drain voltage initially places the device in the linear region, where the on resistance can be extracted. As the drain-source voltage keep increasing, the drain current reaches saturation, representing the maximum drain current that can be extracted. In this device, a maximum drain current ( $I_{dmax}$ ) of 28.3 mA/mm is observed at a gate voltage of -2.5V, with a gate length of 2  $\mu m$ . The on resistance ( $R_{on}$ ) is extracted as 90  $\Omega \cdot mm$ . While this current is lower than the typically reported 100 mA/mm for air adsorbate devices [94], it remains comparable considering the much shorter gate length (250 nm) used in that report, and the much lower sheet resistance of the sample used in that work, which was around 7 k $\Omega/\Box$ .

The orange dot line in Figure 8.4 represents the gate leakage current. With a low drain-source voltage, the gate leakage current increases as the drain voltage decreases. When a high voltage is applied to the gate, more holes accumulate near the junction interface. If the drain-source voltage is not high enough, some carriers may tunnel through the Al-Diamond junction and flow through the gate, forming a gate leakage current. For instance, when the gate voltage is -2.5 V, gate leakage current is larger when the drain voltage is below -2.5 V, and increases exponentially when the drain voltage is below 1.5 V.



Figure 8.5: Transfer characteristic of  $2\mu m$  MESFETs with H-diamond surface expose to air. Drain current is shown in blue solid line, and extrinsic transconductance is shown in orange line.

Figure 8.5 illustrates the transfer characteristic of the same 2  $\mu m$  MESFETs device. The drain-

source voltage is swept from 0 to -3.5 V with a step of 0.5 V, and the drain current is plotted on a logarithmic scale. To extract threshold voltage, at the noise floor, when the drain current in the order of  $1 \times 10^{-5}$  mA/mm, the threshold voltage ( $V_{th}$ ) occurs when the gate voltage is around 0.1 V; below this voltage, the device is fully on. For a set off-state current density of 1  $\mu A/mm$ ,  $V_{th} < 0V$  is extracted across the inspected  $V_{ds}$  range of 1 V to -2 V. The  $I_{on}/I_{off}$  ratio from the graph is of the order of 10<sup>6</sup>, which is two order of magnitude lower comparing to the reported 10<sup>8</sup> for hydrogen-terminated diamond MESFETs [146].

Transconductance reflects how efficiently a FET device can convert a change in gate voltage  $(V_{gs})$  into a change in output current  $(I_{ds})$ . As depicted in Figure 8.5, the maximum extrinsic transconductance is approximately 22 mS/mm when the drain voltage is -3.5 V. This is not as high as reported values of 137 mS/mm and 150 mS/mm [96], [143]. However, this discrepancy may be due to the much higher sheet resistance  $(29 k\Omega/\Box)$  of the sample in this work compared to the reported sheet resistance  $(11 k\Omega/\Box)$  in the literature. In addition, the much shorter gate length (250 nm to 50 nm) in the literature compared with the gate length (2  $\mu m$ ) in this work also contributes to a higher transconductance.



Figure 8.6: Output characteristics for  $2\mu m$  MESFETs with atmospheric adsorbates after 14 days of air expose.

Figure 8.6 presents the output characteristics of the same MESFET devices after exposure to air for 14 days. The plot shows a slight increase in maximum drain current from 28.3 mA/mm to 31.5 mA/mm. The turn-on resistance after exposure is 88  $\Omega \cdot mm$ , slightly lower than before the 14 days of air exposure. The highest gate leakage current is slightly shifted higher from 3.9 mA/mm to 4.3 mA/mm. This increase in maximum drain current and decrease in turn-

on resistance may be attributed to air exposure of the sample. During the fabrication process, although the surface was protected by the gold sacrificial layer, there is still a risk of disrupting the air adsorbates. However, this process is reversible; once the hydrogen-terminated surface is exposed to the air, air adsorbates can build up on the surface again, thereby improving surface conductivity, leading to an increase in drain current and a decrease in turn-on resistance.

Figure 8.7 illustrates the transfer characteristic of the same MESFET devices after 14 days of exposure to air. The threshold voltage at set off-state current 1  $\mu A/mm$ , extrinsic transconductance, and  $I_{on}/I_{off}$  ratio after air exposure are 0 V, 22 mS/mm, and 10<sup>7</sup>, respectively, which is consistent with the data extracted before the air exposure.



Figure 8.7: Transfer characteristic of  $2\mu m$  MESFETs with atmospheric adsorbates after 14 days of air expose.

Multiple  $V_{ds}$  sweeps were conducted on one device with the same gate length and width. The actual sweep applied on this device is 20 times. However, all sweep results slightly fluctuated between the highest and lowest drain current; thus, only 5 sweeps, including the one with the highest drain current and the one with the lowest drain current, are plotted in this thesis. Figure 8.8 shows a very slight increase (4%) in maximum drain current after 3 sweeps, from 20.7 mA/mm to 21.6 mA/mm. Then the current started to drop back after the fourth sweep. The turn-on resistance changed slightly from  $106 \ \Omega \cdot mm$  to  $103 \ \Omega \cdot mm$ . In this case, the device is considered working consistently after 20 repeated sweeps.



Figure 8.8: Output characteristics multiple sweep with gate bias  $V_{gs} = -2.5V$ .



Figure 8.9: Off-state break test of devices with gate lengths of  $2\mu m$  and  $5\mu m$ .

Figure 8.9 illustrates the off-state breakdown test of two devices with gate lengths of 2  $\mu m$  and 5  $\mu m$ , respectively, both with a gate width of 25  $\mu m$ . To ensure the device is turned off, a +1V bias was applied to the gate. The measurement shows that the devices with gate length of 2  $\mu m$  started to breakdown when the drain voltage is around 80V. For the device with a gate length of 5  $\mu m$ , the breakdown voltage is higher than 100V, which is the limit of the measurement system

used.

In summary, longer air exposure may generally improve surface conductivity, leading to an increase in drain current and a slight decrease in turn-on resistance. However, other properties such as transconductance,  $I_{on}/I_{off}$  ratio, and turn-on resistance remain consistent before and after air exposure. Subsequent multiple sweeps on one device indicate that the device operates consistently after 20 sweeps. The off-state breakdown test demonstrates that a device with a gate width of 25  $\mu m$  and length of 2  $\mu m$  breaks down when the drain voltage reaches 80V.

	I <sub>dmax</sub>	Igleakage	Peak g	Ron	Ion/Ioff	$V_{th}$	Off-state
	(mA/mm)	(mA/mm)	(mS/mm)	$(\Omega \cdot mm)$		(V)	Break-
							down
							(V)
Air Adsorbates	28.3	3.9	22	90	106	0	N/A
14 days expose	31.5	4.3	22	88	107	0	-80

Table 8.3: Comparison of device before and after 14 days of air expose.

## **8.3** MESFET with thermal ALD $Al_2O_3$

In the preceding section, the output and transfer characteristics of the MESFET after a 14-day period of air exposure were discussed, with a potential slight improvement in drain current and worse gate current leakage observed. This section will delve into the measurement results of the device after the deposition of 15 nm of thermal ALD  $Al_2O_3$ , providing a detailed comparison between the device before and after the  $Al_2O_3$  deposition.



Figure 8.10: FET structure with atmospheric adsorbates (top) and 15nm of  $ALDAl_2O_3$  (bottom). Ohmic contacts were 80nm Au and the gate contact was 20nm Al/20nm Pt/80nm Au.

Post-oxide deposition, an additional photolithography step was performed to open resist windows in the  $Al_2O_3$  on the gate pad, source, and drain bond pads to allow for direct electrical probing. Buffered oxide etch was then utilized to remove the  $Al_2O_3$  over these contact regions. This step serves two purposes: firstly, it facilitates better contact, minimizing the need to probe through the  $Al_2O_3$  and potentially damage the metal pad; secondly, it reduces the contact resistance between the pad and probe, thereby ensuring more accurate measurement results.



Figure 8.11: Output characteristics for 5  $\mu m$  MESFETs with atmospheric adsorbates after 14 days of air expose.



Figure 8.12: Output characteristics for 5  $\mu m$  MESFETs with ALD  $Al_2O_3$ .

Figure 8.11 displays the output characteristics of a 5  $\mu m$  long gate MESFET after 14 days of air exposure and before the deposition of ALD  $Al_2O_3$ . The maximum drain current is 16.3 mA/mm when the gate-source voltage is at -2.5 V. The drain current maximum is lower than that discussed above for the 2  $\mu m$  gate device, which is most likely caused by a larger gate length. The gate leakage current increases with the decrease in drain-source voltage, with the maximum leakage gate current being 2.49 mA/mm. The linear working region is when the drain-source current is between 0 to -1.5 V when the gate voltage is at -2.5 V. The on resistance extracted from this linear region is 99  $\Omega \cdot mm$ . The device operates in the saturation region when the drain voltage is lower than -1.5 V.

Figure 8.12 illustrates the output characteristics of the same device after the deposition of 15 nm ALD  $Al_2O_3$ . The maximum drain current is 37.5 mA/mm, which is twice the drain current before the oxide deposition. The device operates under the linear region when the drain-source voltage is between 0 and -4 V, a wider range than that of the air adsorbates devices. The on resistance is calculated as 90  $\Omega \cdot mm$ . The on resistance includes the 2 times the ohmic contact resistance  $(R_c)$ , channel resistance under the gate  $(R_{Gch})$ , and the source-to-gate and drain-to-gate channel resistance  $(R_{ch})$ . As discussed in the previous section, the deposition of ALD  $Al_2O_3$  reduces the sheet resistance, which can reduce the channel resistance and subsequently the on resistance.

$$R_{on} = 2R_c + R_{ch} + R_{Gch} \tag{8.1}$$

The previous section discussed the TLM and VDP measurements before and after the ALD  $Al_2O_3$  deposition. An improvement of the sheer resistance was observed from both measurements. This indicates the oxide might provide a higher concentration of carrier on the H-diamond surface, thus improve the maximum drain current of the MESFETs.

Apart from the improvement in maximum drain current and slightly reduced on resistance, the gate leakage current has improved dramatically. The maximum gate leakage current dropped from around 2.5 mA/mm to 0.05 mA/mm, which is a two-order-of-magnitude reduction in the gate leakage current. The decrease in the gate leakage further suggests that there is a larger barrier between the gate-diamond interface, which reduces the current flow through the gate.

Figure 8.13 and 8.14 illustrate the transfer characteristics for this device before and after the  $Al_2O_3$  deposition. The threshold voltage shifted more depletion mode from -0.5 V to 0.6 V, further indicating that the  $Al_2O_3$  provides higher carrier concentration, which takes a higher gate voltage to turn off the channel between the source and drain contact. Another piece of evidence showing that carrier concentration change is the  $I_{on}/I_{off}$  ratio, which changed from  $10^6$  to  $10^4$ . This change is caused by a higher off-state drain leakage current, which is increased due to the improvement of surface conductivity. Additionally, the transconductance after the oxide deposition (16 mS/mm) is slightly lower than that before the deposition, which is 18
mS/mm. As the extrinsic transconductance highly limited by total source access resistance, so it might be influenced gently by sheet resistance, gate contact and ohmic contact.



Figure 8.13: Transfer characteristics for 5  $\mu m$  MESFETs with atmospheric adsorbates after 14 days of air expose.



Figure 8.14: Transfer characteristics for 5  $\mu m$  MESFETs with ALD  $Al_2O_3$ .

	<i>I<sub>dmax</sub></i>	Igleakage	Peak g	Ion /	V <sub>th</sub>	Ron	$R_c$	$R_s$	R <sub>ch</sub>	<i>R</i> <sub>Gch</sub>
	mA/mm	mA/mm	mS/mm	$I_{off}$	(V)	$\Omega \cdot mm$	$\Omega \cdot mm$	$\mathrm{k}\Omega/\Box$	$\Omega \cdot mm$	$\Omega \cdot mm$
Air	16.3	2.49	18	$10^{6}$	-0.5	99	4.15	28.6	28.6	33.5
ALD	37.5	0.05	16	104	0.6	90	3.6	23.4	23.4	36
$Al_2O_3$										

Table 8.4: Comparison of device before and after oxide deposition.

In summary, the maximum drain current of the devices improved from 16.3 mA/mm to 37.5 mA/mm after the deposition of ALD  $Al_2O_3$ . Due to the improvement in surface conductivity, the  $I_{on}/I_{off}$  ratio dropped because of the increase in off-state drain leakage current. The two-order-of-magnitude reduction in gate leakage current, lower transconductance indicate some chemical or physical modification to the gate contact occurred during the  $Al_2O_3$  deposition process. Table 8.4 lists all the parameters extracted from the experimental results. The final calculation of the gate channel resistance after the oxide deposition slightly increased compared to that before the deposition. In conclusion, the ALD  $Al_2O_3$  improves the surface conductivity, possibly explained by defects formed in the oxide growth process serving as acceptors. Further study required to explore the reduction of the gate leakage current.

### 8.4 MESFET high temperature measurement

In the final set of measurements conducted on this  $Al_2O_3$  encapsulated sample, high-temperature measurements were performed. The sample was placed on a stage capable of heating up and sealed under a low to medium vacuum chamber. Measurements were taken at room temperature (300 K), 350 K, 400 K, and 450 K, respectively. To test the stability over the duration of heating, the stage was maintained at 300 K and 400 K for 30 minutes, and the sample was remeasured. In this section, the output and transfer characteristics of the same device with a gate width of 50  $\mu$ m and a length of 5  $\mu$ m at different temperatures will be discussed and compared.

Figure 8.15 and Figure 8.16 illustrate the output characteristics of one device at different temperatures. At room temperature, the maximum drain current is approximately 39.6 mA/mm when the gate-source voltage is -2.5 V and the drain-source voltage is -10 V. However, as the temperature increases, the maximum drain current starts to decrease. At 350 K, it drops to 27.4 mA/mm, and after 30 minutes at this temperature, it further decreases to 22.7 mA/mm. Subsequently, at 400 K, 400 K after 30 minutes, and 450 K, the maximum drain current is measured as 13.7 mA/mm, 11 mA/mm, and 10 mA/mm, respectively. This decrease in maximum drain current with elevated temperature may be caused by the disruption of air adsorbates under the gate, source, and drain contacts. As observed in the TLM measurements, the sheet resistance fluctuates slightly between 21 k $\Omega/\Box$  and 24 k $\Omega/\Box$  with elevated temperature, while the contact resistance increased from 3.8  $\Omega \cdot mm$  to 35.8  $\Omega \cdot mm$  with elevated temperature, thus providing



Figure 8.15: Output characteristics at 300 K, 350 K and 350 K for 30 mins.



Figure 8.16: Output characteristics at 400 K, 400 K after 30 mins and 450 K.

further evidence that the temperature induced change in IV may predominantly be due to the alteration of the H-diamond/contact interfaces.

In the room temperature graph, the device cannot be turned off when the gate voltage is at 1 V. This could be attributed to the improved surface conductivity due to the oxide deposition. A similar phenomenon was observed in the previous section when comparing air adsorbates devices and devices after oxide deposition. A higher gate voltage was required to turn off the device after oxide deposition, as observed in the transfer characteristics comparison.

The gate leakage current remained at the same level, around 0.2 mA/mm, until 30 minutes after the temperature was elevated to 350 K. Subsequently, the leakage current increased to 6.05 mA/mm, which is one order of magnitude higher than the leakage before reaching this temperature. This increase occurred because the elevated measurement temperature provides carriers more energy to overcome the Schottky barrier between the Al and hydrogen-terminated diamond, allowing more current to flow through the gate.

Simultaneously, an opposite direction of current flow of 5.8 mA/mm was observed through the drain and source when the drain-source voltage was below -0.6 V, after the temperature was kept at 350 K for 30 minutes. This current increased with elevated temperature and peaked at -17.5 mA/mm after the temperature was kept at 400 K for 30 minutes, before decreasing to 7.33 mA/mm at 450 K. This current in the area where no bias was applied on the drain can be attributed to the gate leakage current. When a low bias is applied on the drain side and a comparatively high gate bias is applied, holes can be attracted to the gate and tunnel through the barrier to form a gate leakage current. As the temperature increases, carriers acquires more thermal energy, allowing more current to flow through the gate. When the leakage current reaches a certain level, 5.8 mA/mm in this study, a strong enough electric field is created on the source side. Consequently, the source is on a comparatively negative bias compared to the drain, where no or low bias is applied. As a result, holes start to flow from the drain to the source due to the bias difference, leading to the observation of opposite direction current in the output characteristics plot. This current decreases as the drain bias increases, thereby equalizing the bias difference between the drain and source.

	Ron	$R_c$	$R_s$	R <sub>ch</sub>	R <sub>Gch</sub>
	$(\Omega \cdot mm)$	$(\Omega \cdot mm)$	$(\Omega/\Box)$	$(\Omega \cdot mm)$	$(\Omega \cdot mm)$
Room Temperature	114.5	3.63	23.44	23.44	60.36
350K	128	8.73	22.37	22.37	65.81
350K 30 minutes	133.97	10.63	21.51	21.51	69.70
400K	153.58	19.92	21.83	21.83	70.09
400K 30 minutes	166.14	21.35	21.29	21.29	80.86
450K	192.99	35.87	22.6	22.6	76.05

Table 8.5: Extracted resistance from the output characteristics plot at different temperature.

Using the sheet resistance, contact resistance extracted from the TLM measurement at different temperatures, and the on resistance from the output characteristics plot, the channel resistance and gate channel resistance can be calculated. Table 8.5 lists the calculated results of various resistances at different temperatures. The turn-on resistance increases with elevated temperature from 114.5  $\Omega \cdot mm$  to 192.99  $\Omega \cdot mm$  at 450 K, and the gate channel resistance also increases with temperature rise. This increase in gate channel resistance may be caused by the same mechanism that increases the contact resistance—the desorption of air adsorbates under the Al reducing the surface conductivity.

Figure 8.17 illustrates the transfer characteristics of the devices at different temperatures. The  $I_{on}/I_{off}$  ratio of this device at room temperature is only around 10<sup>1</sup>. However, as discussed in the output characteristics plot, this device is not turned off when the gate voltage is 1 V. Therefore, the  $I_{off}$  value is not accurately represented in this plot. The transfer characteristics are measured when the gate voltage is between 1 V and -2.5 V. Thus, the  $I_{on}/I_{off}$  ratio is not accurately captured in this scenario. For the measurements of the device at other temperatures, as the temperature increases, both the on-drain-source current and the off-drain current decrease. Consequently, the  $I_{on}/I_{off}$  ratio remains in the order of 10<sup>2</sup>.

The threshold voltage decreases with higher temperature. Which might caused by the thinner accumulation channel with the temperature increase. With decreased accumulation channel, the voltage required to turn off the device is lower.



Figure 8.17: Transfer characteristics at different temperature.

Figure 8.18 plots the transconductance of the device at different temperatures. The transconduc-

tance decreases with elevating temperature from 17 mS/mm at room temperature to 6 mS/mm at 450 K. This reduction in transconductance indicates the performance degradation of the device, and the gate amplification ability is decreased. With temperature increase, carriers acquires more thermal energy to overcome the Schottky barrier between the Al and H-diamond interface, which affects the gate control over the device. Furthermore, the keep increasing contact resistance affect the transconductance too.



Figure 8.18: Transconductance at different temperature.

To summarise, the device's drain current decreased with elevated temperature. The increased gate channel resistance and ohmic contact resistance might be caused by the disruption of the air adsorbates under the gate and ohmic metal. The elevated temperature provides carriers more thermal energy to overcome the Schottky barrier between the Al and hydrogen-terminated diamond, thus increasing the gate leakage current. The drastically increased gate leakage current creates a comparatively negative field under the source end, which is responsible for the opposite flow direction of the drain-source current when the drain bias is low. The reduction in transconductance with elevated temperature confirms that the gate degrades under high temperatures.

#### 8.5 Chapter Summary

This chapter examined devices fabricated with Au as a sacrificial layer to protect the surface termination. The devices were measured both with air adsorbates and following the deposition of 15 nm of thermal ALD  $Al_2O_3$ . Subsequently, high-temperature measurements were conducted between 300 K and 450 K.

To comprehend the device behaviour, TLM and VDP structures were fabricated on the same sample to extract parameters such as contact resistance, sheet resistance, carrier mobility, and carrier concentration. The VDP and TLM measurements indicated that the oxide deposition improved the surface conductivity of the hydrogen-terminated diamond. There were no significant changes observed in contact resistance before and after the oxide deposition. However, TLM measurements at different temperatures showed that the sheet resistance stabilized between 24 k $\Omega/\Box$  and 21 k $\Omega/\Box$ . Nevertheless, the contact resistance increased drastically from 3.63  $\Omega \cdot mm$  to 35.87  $\Omega \cdot mm$  at 450 K. This increase in contact resistance might be caused by the modification of air adsorbates under the gold ohmic pad. Subsequent TLM structure measurements confirmed the drop in contact resistance to pre-high temperature measurement levels, providing further evidence for this hypothesis.

The MESFET with Al/Pt/Au as the gate metal stack was measured after fabrication and 14 days after exposure to air. A slight increase in the maximum drain current was observed, potentially due to the re-establishment of air adsorbates on the diamond surface. Then, a comparison of the device before and after the 15 nm  $Al_2O_3$  deposition was conducted. The oxide deposition doubled the maximum drain current and reduced the gate leakage current by two orders of magnitude by providing more carriers. However, more study required to investigate the lower gate current leakage after the oxide deposition.

Subsequent high-temperature measurements illustrated that the device started to degrade once the temperature reached 350 K. The decrease in drain current, transconductance, gate channel resistance, and increase in gate leakage current imply that the gate loses its amplification ability with elevated temperature.

### **Chapter 9**

# Accumulation Channel H-Diamond MOSFETs

In the previous chapter, a H-diamond MESFETs was presented. The results demonstrated that ALD  $Al_2O_3$  improved the surface conductivity of hydrogen-terminated diamond and enhanced thermal stability. Although the oxide effectively reduced the on-state gate leakage, the off-state source-drain current increased, thereby affecting the  $I_{on}/I_{off}$  ratio by two orders of magnitude. To mitigate the off-state source-drain current, an accumulation (enhancement) channel of MOS-FET was fabricated and will be discussed in this chapter.

The currently reported normally-off hydrogen-terminated diamond MOSFETs exhibit some limitations in terms of the fabrication process or device properties. Devices employing gate oxides such as  $HfO_2$  and  $Al_2O_3$  as insulator layers typically show maximum drain currents below 40 mA/mm [147], [148]. Another strategy involves partially oxide-terminating the exposed diamond surface in the drain-source channel or doping the channel with nitrogen to induce impurities and create an energy barrier between the source and drain [110], [149]. However, the reported drain currents range between 5 to 20 mA/mm. Yu et al. [150] reported normally-off diamond devices with decent drain currents over 300 mA/mm and an  $I_{on}/I_{off}$  ratio over 10<sup>7</sup> by using Si-terminated drain-source gaps and boron-doped source and drain. By using h-BN as a dielectric material, Y. Sasama et al. [151] achieved normally-off hydrogen-terminated diamond devices with a maximum drain current around 200 mA/mm, high carrier mobility over  $600 \ cm^2V^{-1}s^{-1}$ , and an  $I_{on}/I_{off}$  ratio of 10<sup>8</sup>. Instead of using silicon termination or h-BN as dielectric materials, in this work, PVD  $Al_2O_3$  was investigated and used as dielectric material on clean H-diamond surface to fabricate normally-off hydrogen-terminated diamond MOSFETs.

### 9.1 **PVD** $Al_2O_3$ deposition

As discussed previously, the conductivity of air-exposed H-diamond might be attributed to the presence of air adsorbates, which provide a hole accumulation layer near the diamond surface [43]. By removing the air adsorbates from the H-diamond, the surface conductivity also vanishes [43]. In this case, it is assumed that the hydrogen termination provides an unpinned Fermi level in the H-diamond. The Fermi pinning effect denotes the phenomenon wherein the Fermi level within a semiconductor material is fixed or "pinned" at a specific energy level near the interface between two materials. This can arise from defects or chemical bonding that trap charge carriers, thereby affecting altering the Fermi energy level near the interface [30], [152].

With the support of these theories, a normally-off device can be achieved by creating a hydrogenterminated surface without any air adsorbates between the source and drain contacts. This cleaned surface (the surface without air adsorbates) creates an energy barrier between the source and drain contacts. However, since the hydrogen termination provides an unpinned Fermi level, once a gate bias is applied, the energy barrier between the source and drain will be decreased, thereby attracting carriers under the source and drain contacts to accumulate under the gate area and forming an accumulation channel to allow current flow between the drain and source. Thus, this normally-off device is also referred to as an accumulation channel H-diamond MOSFET. Figure 9.1 illustrates the energy band diagram.



Figure 9.1: Energy diagram between the gate channel and source drain area. (a) Band diagram without bias applied on the gate. (b) When a bias applied on the gate.

Oxides like  $V_2O_5$  and  $Mo_2O_3$  were explored in this group. Both oxides, with LUMO below the maximum valence band of H-diamond, were discovered to improve the surface conductivity by acting as an acceptors. ALD  $Al_2O_3$  was discussed in the previous chapter, where an improvement of the surface conductivity was discovered. It is assumed that some defects in the ALD  $Al_2O_3$  function as additional acceptors, thereby enhancing the surface conductivity of Hdiamond. To achieve the normally-off device discussed above, an oxide that neither behaves as a surface acceptor nor damages the surface termination is required. A 3  $mm \times 3 mm$  diamond sample was cleaned and hydrogen terminated in Australia. The sheet resistance was measured using a Hall Kit by contacting the surface with four probes until the sheet resistance stabilized. Then, silver dag was applied to the four corners of the sample to provide better contact during measurements. The measured sheet resistance after applying the silver dag was around 31.24 k $\Omega/\Box$ .

Subsequently, the sample was subjected to physical vapor deposition in PLASSYS 5, an one hour 300 °C annealing was performed when the vacuum level of the chamber reached to  $1 \times 10^{-7}$  mbar. This step is designed to remove most of the air adsorbates on the hydrogen terminated diamond surface[43]. Then 15 nm of  $Al_2O_3$  was deposited on the sample surface after the chamber temperature back to room temperature. The sheet resistance was measured again using the Hall Kit, with four probes penetrating through the silver dag at the four corners of the sample. The measured sheet resistance increased to 867.6 k $\Omega/\Box$  after the deposition.

To ensure that the oxide did not damage the hydrogen termination, the sample was rinsed in 10:1 buffered oxide etch (BOE) for 5 minutes. The hydrofluoric acid (HF) in the etch solution removed the oxide. A subsequent Hall Kit measurement conducted after oxide removal showed that the sheet resistance dropped back to 54.61 k $\Omega/\Box$ .

The experiments conducted here verified that E-beam vapor deposition of  $Al_2O_3$  does not provide a lowest unoccupied molecular orbit to act as electron acceptors, thus improving the sheet resistance while suppressing the surface conductivity of hydrogen-terminated diamond. After removing the oxide, the recovery of the sheet resistance indicated that the increase in sheet resistance was not due to damaged hydrogen termination.

	Air Expose	$Al_2O_3$ deposition	After BOE etch
$R_s (k\Omega/\Box)$	31.24	867.6	54.61

Table 9.1: Hall Kit sheet resistance measurement of the sample (1) in air (2) after 1 hour 300 °C annealing and 15 nm of  $Al_2O_3$  deposition (3) $Al_2O_3$  removed by 5 minutes BOE wet etch.

The suppression of surface conductivity of hydrogen-terminated diamond after  $Al_2O_3$  deposition can be explained as follows. Firstly, the  $Al_2O_3$  may act as a passivation layer, preventing the hydrogen termination from contacting air and forming air adsorbates, thereby reducing the sheet resistance. During the annealing process, most of the air adsorbates could be removed theoretically. The oxide deposition could then isolate the surface from the air to form an adsorbates layer. After removing the oxide, air adsorbates re-established upon exposure to air. A more detailed study is needed to understand this phenomenon better. Gated VDP structures fabrication with different annealing temperature before the oxide deposition can be used to confirm the air-adsorbates removal. In addition, XPS can be used to detect the energy band bending and oxide quality.

### 9.2 MOSFETs measurement

The fabrication process of these MOSFETs has been detailed in Chapter 4, and the complete fabrication flow can be seen in Figure 4.26. A 4.5 mm×4.5 mm single polished diamond sample was hydrogen-terminated in Australia. The sheet resistance of the sample before fabrication was measured as 50 k $\Omega/\Box$  which is higher than the desirable resistance ( $\approx 10 k\Omega/\Box$ ) used for H-diamond device fabrication in most of those reported literature. Figure 9.2 illustrates the cross-section of the sample in the final steps. After depositing the ohmic pads and before the PVD  $Al_2O_3$  deposition, a 350 °C in-situ annealing process was conducted to desorb air adsorbates from the diamond surface between the the source and drain contacts. It has been reported that air adsorbates on H-diamond can be fully removed at 400 °C[43], based on the experiment discussed in the previous section where a surface conductivity drop and recover on H-diamond was confirmed after 300 °C annealing, 350 °C was chosen as a safe temperature to remove air adsorbates on the H-diamond sample. Following deposition of 35 nm of PVD  $Al_2O_3$  was conducted to serve as the dielectric material between the gate and the H-diamond. To prevent shorting between the gate and source-drain contacts, a 15 nm layer of ALD  $Al_2O_3$  was deposited on top of the PVD oxide to cover all edge walls.



Figure 9.2: MOSFETs cross section structure.

The solid red line represents the 2DHG. The dotted red line represents the reduced carrier density 2DHG due to the 350 °C desorption process.

Figure 9.3 illustrates the output characteristics of MOSFETs with a gate/channel length of 1  $\mu$ m and a width of 25  $\mu$ m. The gate voltage ranges from 0 to -14 V, with a maximum drain current of 110 mA/mm recorded at -14 V gate voltage. Notably, (and despite the initial higher material  $R_{sh}$  values) this drain current surpasses that achieved by MESFETs which had a shorter gate length (2  $\mu$ m) discussed in the preceding chapter. Previous studies report the highest drain current maximum, ranging between 110 mA/mm and 150 mA/mm, for MOSFETs employing oxide as

an insulator [153], [154]. Comparatively, the drain current attained in this study is within this range.



Figure 9.3: 25  $\mu m$  wide, 3  $\mu m$  gate length AC-MOSFETs output characteristics.  $V_{gs}$  sweep from 0 to -14 V with step of 1 V.



Figure 9.4: 25  $\mu m$  wide, 3  $\mu m$  gate length AC-MOSFETs transfer characteristics.  $V_{ds}$  sweep from 0 to -8 V with step of 1.5 V.

The on resistance, when the gate voltage is -14 V, is determined as 42  $\Omega$ ·mm, marking a re-

duction of over fifty percent from the resistance observed in MESFET devices. Furthermore, examination of the output graph reveals negligible gate leakage current across the inspected range, estimated at approximately  $10^{-7}$  mA/mm.

The transfer characteristics are depicted in Figure 9.4. The drain voltage was swept from 0 to -8 V, while the gate voltage stepped from 0 to -14 V in increments of 1.5 V. Notably, the  $I_{on}/I_{off}$  ratio exceeds 10<sup>9</sup>, marking a three-order-of-magnitude enhancement compared to MESFET devices discussed in the preceding chapter, and surpassing by three orders of magnitude the reported MOSFETs with  $Al_2O_3$  as the insulator, achieving a drain current of 135 mA/mm [154]. This notably higher ratio indicates superior switching behaviour, lower leakage current, and consequently, a better signal-to-noise ratio.

At the noise floor level, the threshold voltage is extracted as -4 V. At a set off-sate drain current 1  $\mu A/mm$  was extracted as -5 V, significantly surpasses the reported 1.8 V [154] around 7 V.

A depletion region calculation of the  $V_{th}$  is carried as follows, parameters used for the calculation are, metal work function  $\varphi_m = 4.33$  eV, diamond surface work function  $\varphi_s = 1.45$  eV,  $t_{ox} = 50$ nm, dielectric constant  $\kappa_{ox} = 9$ , carriers concentration  $N_a = 2 \times 10^{12} \text{ cm}^{-2}$ , intrinsic H-diamond carrier concentration  $n_i = 10^{11} \text{ cm}^{-2}$  [151], permittivity of free space  $\varepsilon_0 = 8.85 \times 10^{-12}$  F/m. Here an ideal H-diamond surface without any adsorbates is assumed, with this ideal surface, Fermi level is assumed close to the middle of the band gap of the diamond. Considering the band gap of the diamond is 5.5 eV, the ideal Fermi level suppose to be 2.75 eV below the conduction band minimum. Due to the hydrogen termination induced negative electron affinity which is 1.3 eV below the conduction band minimum. Thus, a 1.45 eV of work function of H-diamond surface is used for the calculation.

In  $V_{fb}$  calculation, a perfect interface without trapped and fixed charges, where  $Q_f$  is assumed to be 0,

$$V_{fb} = \varphi_{ms} - \frac{Q_f}{C_{ox}} = \varphi_m - \varphi_{ms} = 2.88 \ eV$$
 (9.1)

In  $\varphi_f$  calculation room temperature 300 K is applied,

$$\varphi_f = \frac{kT}{q} ln(\frac{N_a}{n_i}) = 0.07 \ eV \tag{9.2}$$

where K is Boltzmann constant  $1.38 \times 10^{-23}$  J/K, T is the absolute temperature and q is elementary charge  $1.6 \times 10^{-19}$  C, capacitance under the gate metal is calculated as below,

$$C_{ox} = \frac{\varepsilon_{ox}}{t_{ox}} = \frac{\kappa_{ox}\varepsilon_0}{t_{ox}} = 1.59 \times 10^{-7} \, F/cm^2 \tag{9.3}$$

the depletion charge density is calculated as below,

$$Q_d = q \times N_a = 3.2 \times 10^{-7} \, C/cm^{-2} \tag{9.4}$$

Finally, the threshold voltage can be calculated as,

$$V_{th} = V_{fb} + 2\varphi_f + \frac{Q_d}{C_{ox}} = 2.88 + 0.14 + 2.01 = 5.03 V$$
(9.5)

This calculated threshold voltage aligns well with the extracted threshold voltage (-5 V) with set-off current in the order of  $10^{-3}$  mA/mm. And it explains the removal of the air-adsorbates on the H-diamond surface contributes to this large threshold voltage.

The transconductance extracted from this device measures approximately 22 mS/mm, a value within the range reported for hydrogen-terminated diamond devices designed for normally off operation. Notably, as this device represents the first iteration utilizing this fabrication method, numerous parameters remain open for optimization, potentially enhancing transconductance. For instance, adjusting the thickness of the oxide layer, as demonstrated in Liu et al.'s study [154], where a thinner oxide layer (100 nm) substantially increased transconductance (1410 mS/mm). Additionally, reducing the gate length could enhance gate control and subsequently improve transconductance. Furthermore, optimizing oxide deposition to mitigate interface traps and defects could enhance carrier mobility and consequently increase transconductance.

Ungated TLM structures were fabricated on the sample and measured using the probe station. The highest current recorded from the TLM structure was  $1.6 \times 10^{-3}$  mA/mm. This remarkably low current confirms the low surface conductivity post-PVD  $Al_2O_3$  deposition, indicating a scarcity of carriers in the channel available for conduction when a bias is applied across the TLM ohmic pad. As per design this is most likely attributed to the desorption of air adsorbates during the 350 °C annealing process [43], coupled with the presence of the  $Al_2O_3$  passivation layer that suppresses the 2DHG formation.

Conversely, in the present MOSFET devices, a hydrogen-terminated surface with minimal transfer doping is achieved between the source and drain channels. In the absence of gate bias, there is little active hole population in the channel and minimal carriers traverse between the source and drain, rendering the device normally off. Upon applying a negative gate bias, holes are drawn to the surface of the hydrogen-terminated diamond, forming a hole accumulation layer. This layer establishes a channel between the drain and source. Consequently, when a bias is applied to the drain terminal, current flows from drain to source, with the energy barrier between the hydrogen-terminated diamond and  $Al_2O_3$  preventing carriers from flowing to the gate.

As discussed in at the start of this section, hydrogen termination provides unpinned Fermi level

on the diamond surface. In this investigation, the hydrogen termination on the diamond surface and the interface between the hydrogen termination and PVD  $Al_2O_3$  both provide unpinned Fermi levels. The presence of these unpinned Fermi levels facilitates hole accumulation. The high drain current is indirect evidence of this theory. And more investigation such as XPS and KPFM which can measure the band bending on the interface and work function can be done to confirm.



Figure 9.5: MOSFETs off-state breakdown measurement.

When the device is in the off state ( $V_g = 0$  V), with more negative voltage applied to the drain terminal, carriers on the drain side accumulate enough energy to tunnel through the energy barrier between the gate channel and the source-drain side, potentially leading to off-state breakdown. Figure 9.5 illustrates the off-state breakdown measurement of this device, revealing that breakdown occurs at approximately 22.5 V drain-source voltage. Notably, this breakdown voltage is lower than the 80 V achieved in MESFET devices. This discrepancy is most certainly attributed to the architecture of these MOSFET devices, wherein the gate contact extends over the source and drain terminals as separated by only the ALD  $Al_2O_3$  layer.

As discussed earlier in this section, the deposition of ALD  $Al_2O_3$  aims to prevent gate shorting with the source and drain, with only around 15 nm of  $Al_2O_3$  separating the edge wall of the source-drain contact and the gate. This thin oxide layer may prove insufficient, with the barrier potentially too thin to prevent high conduction when drain voltage reaches high levels. Consequently, optimizing the device architecture or increasing the thickness of the oxide layer may enhance the off-state breakdown voltage.



Figure 9.6: Benchmarks of threshold voltage and drain current maximum of the reported stateof-the-art normally-off hydrogen termination diamond MOSFETs.

### 9.3 Split CV measurement and mobility extraction

By measuring the capacitance between the gate and channel and integrating the CV curves, the carrier density can be extracted. With the calculated carrier density, the effective carrier mobility can be deduced using the following equation:

$$\mu_{eff} = \frac{1}{R_{sh} \cdot n \cdot q} \tag{9.6}$$

Here,  $R_{sh}$  represents the channel sheet resistance, *n* stands for the 2D carrier density, and *q* is the elementary charge constant [162], [163].

Split CV was conducted in this study to measure the gate-channel capacitance ( $C_{gc}$ ). By grounding the drain and source terminals, a voltage range of -13 V to 1 V was applied to the gate terminal, while the capacitance was measured from the source terminal. Figure 9.7 illustrates the capacitance against voltage plot.

However, as depicted in the cross-section of the device in Figure 9.8, the overall capacitance under the gate terminal encompasses three components: the gate-to-source capacitor ( $C_{gs}$ ), the gate-to-drain capacitor ( $C_{gd}$ ), and the gate-to-channel capacitor ( $C_{gc}$ ).  $C_{gs1}$  and  $C_{gd1}$  are the capacitors between the gate edges and the source and drain edges with 15 nm of ALD  $Al_2O_3$  in the middle. It is necessary to deduct the parasitic capacitance between the gate and drain contacts,



Figure 9.7: MOSFETs CV measurement at 1 MHz.

as well as between the gate and source contacts, from the overall capacitance to extract the effective device capacitance. In the off-state, the measured capacitance should be attributed to the total parasitic capacitance between the gate and the ohmic contacts, from figure 9.7 suggesting a range for the parasitic capacitance between  $5.5 \times 10^{-14}$  F and  $5.9 \times 10^{-14}$  F.



Figure 9.8: The capacitor distribution in the AC-MOSFETs.

To estimate the parasitic capacitance through calculation, several assumptions were made. First, neglecting the parasitic capacitances  $C_{gs1}$  and  $C_{gd1}$ . Compared to the 1 µm overlapping length between  $C_{gs}$  and  $C_{gd}$ , the overlapping length between the gate edge and the source ( $C_{gs1}$ ) and drain ( $C_{gd1}$ ) edges is only around 25 nm. This makes these two parasitic capacitors negligible. Assuming uniform oxide deposition, with a total thickness (d) of 50nm for two layers of  $Al_2O_3$ , and considering a gate extending 1µm over the source and drain sides, with a channel width

of  $25\mu$ m, the total area (*A*) of the parasitic capacitor is  $50 \ \mu m^2$ . A dielectric constant ( $\kappa$ ) of 6.6 was utilized, selected based on the 15nm of low temperature plasma enhanced ALD  $Al_2O_3$  and 35nm of PVD  $Al_2O_3$ . In addition, Figure 9.7 illustrates a parasitic capacitance at the level of  $5.9 \times 10^{-14} F$  before the  $V_{gs}$  reach the threshold voltage(-5 V). According to this parasitic capacitance extracted from the C-V measurement and the permittivity of free space ( $\varepsilon_0 = 8.85 \times 10^{-12} F/m$ ), a dielectric constant around 6.6 was calculated. A typical dielectric constant for  $Al_2O_3$  is between 9 to 10. However the ALD grow  $Al_2O_3$  dielectric has been measured as low as 7.1 when the growth temperature is around  $100^{\circ}$ C[164]. The low chamber temperature ( $100^{\circ}$ C) and plasma-enhanced ALD process applied in this work could induce a low dielectric constant  $Al_2O_3$  might contribute to a lower dielectric constant. However, this aspect requires further investigation for verification.



Figure 9.9: Carrier density versus gate voltage in logarithmic scale.

With the measured capacitance and estimated parasitic capacitance, the effective capacitance  $(C_{eff})$  can be calculated. By integrating the area between the effective capacitance, the charge density  $(Q_i)$  can be determined:

$$Q_i = \int C_{eff} d(V_{gs}) \tag{9.7}$$

The charge density thus obtained can be translated into carrier density when divided by the elementary charge q. After normalization, the carrier density against gate voltage is plotted in Figure 9.9. Before the gate voltage reaches -5 V, a small number of carriers on the order of  $2 \times 10^9 \text{ cm}^2$  are observed, possibly originating from residual air adsorbates on the hydrogen-

terminated diamond surface or the noise floor of the measurement. As reported by Maier et al., complete desorption of air adsorbates on hydrogen-terminated diamond occurred at 400 °C under vacuum conditions [43]. Therefore, the 350 °C annealing temperature utilized in this study might have left some air adsorbate residue on the surface, contributing to the low carrier concentration below the threshold voltage. As the gate voltage increases, the carrier density reaches the accumulation area after -10 V, consistent with the CV measurement results.



Figure 9.10: 25  $\mu m$  wide, 1  $\mu m$  gate length AC-MOSFET Output characteristics.

To extract the carrier mobility, the equation 9.6 mentioned at the beginning of this section should be employed. Output characteristics were conducted to derive the turn-on resistance ( $R_{on}$ ). Figure 9.10 illustrates the output characteristics, with the maximum drain current recorded as 58 mA/mm, lower than the device discussed previously, possibly due to uneven termination and oxide deposition. The linear region is selected at  $V_{ds}$  equal to -2 V, and with increasing gate voltage, turn-on resistance is calculated using  $I_{ds}$  and  $V_{ds}$ .

Transfer characteristics are also depicted in Figure 9.11. The  $I_{on}/I_{off}$  ratio is on the order of 10<sup>9</sup>, with maximum transconductance extracted as 17 mS/mm. The threshold voltage falls between -4 V and -5 V. All parameters from the transfer characteristics are consistent with the previously demonstrated device.



Figure 9.11: 25 µm wide, 1 µm gate length AC-MOSFET Transfer characteristics.



Figure 9.12: TLM measurement results after  $Al_2O_3$  etched off.

In the final test for this device, TLM measurements were conducted to extract the contact resistance, following the calculation of turn-on resistance from the output characteristics. Due to the scarcity of holes in the channel area resulting from the annealing process, TLM measurements could only be performed after removing the oxide layer and allowing air adsorbates re-establish, forming a hole accumulation layer on the hydrogen-terminated diamond surface. A 10-minute rinse in 1:10 buffered oxide etch was carried out to completely remove the oxide from the sample surface, ensuring no residual oxide remained. TLM measurements were then conducted post-etching. Figure 9.12 illustrates the TLM measurement results, with the intercept of the plot calculated as 226.73, resulting in a contact resistance of 11.3  $\Omega \cdot mm$ .

Though this value exceeds the reported contact resistance formed between hydrogen-terminated diamond and Pd in Chapter 7, it's important to consider the difference in sheet resistance between this sample (50 k $\Omega/\Box$ ) and the sample used in Chapter 7 (8.6 k $\Omega/\Box$ ). The higher contact resistance may stem from this disparity in sheet resistance. Given that the FET source and gate contact width is 25  $\mu m$ , the contact resistance can be converted to 453.46  $\Omega$ . It's assumed that the contact resistance measured here remains consistent when the surface is covered with  $Al_2O_3$ . Consequently, the total resistance between the two ohmic contact pads includes the resistance of the channel and two contact resistances resulting from the metal-substrate contact. As the area under the metal was protected throughout the measurement and fabrication process, a constant contact resistance with or without the oxide layer is assumed.



Figure 9.13: Sheet resistance versus gate source voltage.

With the turn-on resistance from the output characteristics and contact resistance from the TLM measurement, the resistance in the channel can be calculated using the following equation:

$$R_{channel} = R_{on} - 2R_c \tag{9.8}$$

Then, the sheet resistance can be determined using the equation:

$$R_s = \frac{R_{channel}}{W \times L} \tag{9.9}$$

where L and W represent the length and width of the gate channel, respectively. Figure 9.13 illustrates the extracted sheet resistance versus gate voltage. With the calculated  $R_s$ , carrier density *n*, and elementary constant *q*, the mobility can be computed using equation 9.6.



Figure 9.14: Mobility versus gate source voltage.

Figure 9.14 illustrates the calculated two-dimensional hole gas (2DHG) mobility under different gate biases. For lower gate bias ( $\leq 6 V$ ), the mobility is observed to be very close to 0, attributed to the very high  $R_{sh}$  in the channel when the device is in its off-state (Figure 9.11). As the gate voltage increases, holes from the adjacent source and drain 2DHG areas are attracted to the gate area, forming a hole accumulation region under the gate channel. At approximately -5V, which corresponds to the threshold voltage extracted from the transfer characteristics plot, the channel forms and expands with the increasing gate voltage. When the gate voltage exceeds -8 V, the mobility reaches saturation due to increased scattering. With a carrier concentration of around  $2 \times 10^{12} cm^2$ , the maximum mobility extracted from this device is approximately  $105 cm^2/V \cdot s$ . This mobility is not as high as the values reported by Sasama et al. [151]. However, considering that the annealing temperature (350 °C) is not optimal for removing air-adsorbates, increasing the annealing temperature to 400 °C could result in a hydrogenated diamond surface with fewer surface defects. With a hydrogenated diamond surface exhibiting fewer surface defects, a higher mobility could, in theory, can be achieved.

In this section, the dielectric constant assumed for the calculation is 6.6. This choice is supported by the fact that "thin" ALD  $Al_2O_3$  films (less than 35 nm) have been reported to exhibit a dielectric constant around 7. Additionally, the PVD oxide layer likely contains significant defects, which can further reduce the dielectric constant. The lowest parasitic capacitance observed, above  $5.5 \times 10^{-14}$  F as extracted from Figure 9.7, suggests a minimum dielectric constant of approximately 6.3. On the other hand, the maximum possible parasitic capacitance is below  $6 \times 10^{-14}$  F, implying a dielectric constant around 6.7.

If the dielectric constant were to vary within this range, from 6.3 to 6.7, the carrier mobility would consequently range from 75  $cm^2/V \cdot s$  to 140  $cm^2/V \cdot s$ . This sensitivity underscores the importance of accurately determining the dielectric constant, as it significantly influences the calculated carrier mobility.

However the extracted mobility figures are well aligned with those reported previously for Hdiamond which provides some confidence in the accuracy of the extraction process.



Figure 9.15: Benchmarks of carrier mobility versus the carrier density of the reported state-ofthe-art hydrogen termination diamond.

### 9.4 Chapter Summary

In this chapter, an Accumulation Channel of hydrogen-terminated diamond MOSFETs has been demonstrated. Through a 350 °C anneal, the majority of air adsorbates were removed prior

to  $Al_2O_3$  deposition. A 35 nm PVD  $Al_2O_3$  layer was utilized as the insulator beneath the gate, while a 15 nm layer of low-temperature plasma-enhanced ALD  $Al_2O_3$  was applied atop the PVD oxide to prevent gate short-circuiting with the source and drain ohmic pads.

The maximum drain current of 110 mA/mm was extracted with a channel length of 1  $\mu$ m and width of 25  $\mu$ m. The  $I_{on}/I_{off}$  ratio exceeding 10<sup>9</sup> was obtained from the transfer characteristics. The threshold voltage and transconductance were approximately -4 V and 22 mS/mm, respectively. The off-state breakdown voltage was measured at around -20 V.

By employing split CV measurement, the carrier concentration and mobility were extracted as  $2 \times 10^{12} \ cm^2$  and  $105 \ cm^2/V \cdot s$ , respectively. These calculations were based on the assumption of a dielectric constant of the oxide stacks being 6.6 and a parasitic capacitance of  $5.84 \times 10^{-14}$  F.

Overall, an Accumulation Channel device has been fabricated and measured. Numerous avenues for future work exist to enhance device properties. Improvements in the quality and adjustment of oxide thickness could enhance transconductance and off-state breakdown voltage. Gated Hall bar fabrication could offer a more accurate method for measuring mobility and concentration. Additionally, modifying the gate size may alter drain current and parasitic capacitance. These avenues present promising directions for optimizing device performance.

### Chapter 10

### **Conclusions and Future Work**

### **10.1** Conclusions

Diamond has been demonstrated as an highly promising material for high-power and high-frequency electronic devices due to its wide band gap, high thermal conductivity, and intrinsic mobility. In this work, the main purpose was to investigate the limitations of the existing fabrication methods of MOSFETs on H-diamond, thereby developing a novel method to improve the material and device properties, such as carrier mobility ( $\mu$ ), drain current ( $I_{ds}$ ),  $I_{on}/I_{off}$  ratio, and threshold voltage ( $V_{th}$ ).

In this work, Chapter 6 explored "Spillover" to hydrogen-terminate the diamond surface with catalyst metals (Pd and Pt) and hydrogen gas. With a designed 2-hour degassing process at 900°C followed by a 12-hour spillover process at 500°C, VDP measurements showed that the surface altered from insulator to conductor with sheet resistance around 13.2  $k\Omega/\Box$ , which is comparable to plasma-based hydrogen termination processes. The changes in surface hydrophilicity before and after the experiment also indicate the termination changes from oxygen to hydrogen. The absence of a requirement for a high vacuum system and no changes to the surface roughness are two advantages of this method over the traditional plasma-enhanced hydrogen termination process for diamond.

Chapter 7 explored alternative fabrication methods to replace the conventional gold sacrificial method in H-diamond FET fabrication. Lift-off resist was explored as an alternative protective layer; however, TLM measurements indicated the potential existence of resist residue. Further cross-fabrication verified around 14 nm of resist residue left on the surface. Subsequently, hydrogen plasma was investigated to remove the resist residue without damaging the surface termination. By altering the plasma power, stage temperature, and hydrogen flow rate, a slow etching recipe with an etching rate of 4.53 nm/s was established. Unfortunately, the H-diamond

surface turned into an insulator after the etching process. A potential explanation is that hydrogen plasma etches off diamond under low-temperature etching conditions. The final method explored in this chapter was Pd ohmic contact. TLM results illustrated that, on two H-diamond samples with similar sheet resistance, Pd contact showed lower contact resistance (3.82  $\Omega \cdot mm$ ) than that of gold contact (7.52  $\Omega \cdot mm$ ).

Chapter 8 reported H-diamond MESFETs fabricated using the conventional gold sacrificial layer method. The device was characterised after fabrication, after ALD  $Al_2O_3$  deposition, and under high temperature. The measurements show that the ALD  $Al_2O_3$  improved the drain current by a factor of two and reduced the gate leakage current by two orders of magnitude. However, the subsequent high-temperature tests showed that the devices degraded with elevated temperatures from 300K to 450K. The continually increasing contact resistance extracted from the TLM structure and decreasing transconductance measured from the MESFETs indicate there might be some disruption of air adsorbates under the gate contact and the ohmic contact.

The biggest achievement of this work is the Accumulation Channel MOSFETs reported in Chapter 9. By high-temperature (350 °C) annealing in a vacuum chamber and deposition of PVD  $Al_2O_3$ , a cleaned H-diamond area (an area without air adsorbates but with hydrogen termination) was created between the source and drain contacts, thus generating a normally-off device with an  $I_{on}/I_{off}$  ratio in the order of 10<sup>9</sup>, a highest drain current of 110 mA/mm and threshold voltage < -5 V with a gate width of 25  $\mu m$  and length of 1  $\mu m$ . The subsequent split CV measurement contributed to the calculation of carrier mobility and concentration, which are 105  $cm^2/V \cdot s$  and  $2 \times 10^{12} cm^{-2}$ , respectively. This device is the first demonstrated accumulation channel H-diamond MOSFET fabricated with PVD  $Al_2O_3$ .

#### **10.2** Future work

Further work can be undertaken to optimise and verify many aspects of this research. For the spillover process, although a sheet resistance of 13.2  $k\Omega/\Box$  has been achieved, several parameters, such as higher concentrations of hydrogen gas, faster flow rates, longer degassing durations, and extended spillover durations, can be explored to improve surface conductivity and uniformity. Furthermore, XPS can be a helpful tool to characterise the surface termination and provide solid evidence if the surface has been hydrogen terminated.

Regarding alternative fabrication methods, hydrogen plasma for resist residue removal still has potential to replace oxygen plasma by increasing the stage temperature. However, the compatibility of the resist used for fabrication with elevated temperatures needs to be considered to avoid damage and reduce the accuracy of the resist. Further experiments by fabricating TLM structures with Pd or Au on the same H-diamond sample can be conducted to enable stricter comparisons. Additionally, more metals like Pb and Pt, with similar work functions to gold, could be explored to form better ohmic contacts.

Chapter 8's high-temperature measurements indicate a potential disruption of air adsorbates under the ohmic and gate contacts. Partially removing the air adsorbates with high-temperature annealing before the deposition of the contact metal could help determine if the increase in contact resistance and decrease in transconductance are caused by the disruption of air adsorbates. Fully removing the air adsorbates under the ohmic contact also helps explore the mechanism behind the ohmic contact between the H-diamond and high work function metals.

For the accumulation channel H-diamond device, Factors such as the thickness of the oxide and ohmic contact metal can be adjusted to achieve a more uniform capacitor between the gate and the substrate. The deposition of PVD  $Al_2O_3$  can be optimised to achieve a more uniform oxide layer, improving the quality of the capacitor. Additionally, since there is a small overlapping area between the gate edges and the source and drain edges, which have only 15 nm of ALD  $Al_2O_3$ , the off-state breakdown in this work might also be improved by adjusting the thickness of the contact metal and the oxide to form a better contact. Furthermore, the annealing temperature can be increased to 400°C to fully remove the air adsorbates, potentially increasing carrier mobility by reducing scattering. Moreover, some device structures, such as the gate length, can be altered to achieve different device properties. To extract more accurate mobility data and contact resistance from this type of device, gated TLM and VDP structures can be fabricated on the H-diamond after the annealing process. Further measurements, such as high-temperature measurements, can also characterise the temperature stability of this device.

# Appendix A

# Appendix

	PMMA	PMMA	S1805	LOR 3A
	50K 12%	200k 4%		
Spinning	4500rpm	4500rpm	4500rpm	5000rpm
	60secs	60secs	30secs	45secs
Baking	180°C	180°C	115°C 3mins	180°C
	2mins	2mins		5mins
Development	1:1	1:1	1:1 Microposit:	CD-26
Solvent	MIBK:IPA	MIBK:IPA	RO water	
Development	23°C	23°C	75secs	70secs
	30secs	30secs		
Ashing	80W	80W	100W 5mins	80W
	2mins	2mins		5mins

Table A.1: Parameters for the resist used in this work. PMMA is e-beam resist, and S1805 is photo-resist. LOR stands for Lift-off resist which is not sensitive to either UV light nor e-beam, however it can be lift-off and etched by TMAH.

	Two stages Acid clean
Step 1	Mix 65% <i>HNO</i> <sub>3</sub> and 37% <i>HCl</i> in the ratio of 1:1
Step 2	Acid mixture heat up to 160°C to boil the mixture
Step 3	Sample rinsed in the boiled acid for 15 minutes
Step 4	Sample rinsed in RO water for 3 times to clean off the acid residue
Step 5	Mix $65\% HNO_3$ and $95\% H_2SO_4$ in the ratio of 1:3
Step 6	Acid mixture heat up to 240°C to boil the mixture
Step 7	Sample rinsed in the boiled acid for 15 minutes
Step 8	Sample rinsed in RO water for 3 times to clean off the acid residue

Table A.2: Parameters for the two stages acid cleaning process used in this work.

Step 1	Sample load and chamber vacuum down
Step 2	Stage heat up to 800°C
Step 3	Hydrogen plasma strike
Step 4	Stage cool down to 700°C
Step 5	Stop hydrogen plasma
Step 6	Chamber cool down to room temperature
Step 7	Chamber venting

 Table A.3: Parameters for diamond hydrogen termination provided by Dr. Alastair Stacy from Australia.

### References

- M. Kasu, K. Ueda, Y. Yamauchi, A. Tallaire, and T. Makimoto, "Diamond-based RF power transistors: Fundamentals and applications," *Diamond and Related Materials*, vol. 16, no. 4-7 SPEC. ISS. Pp. 1010–1015, Apr. 2007, ISSN: 09259635. DOI: 10.1016/j.diamond.2006.12.046.
- [2] Satoshi Koizumi, Hitoshi Umezawa, Julien Pernot, and Mariko Suzuki, *Power Electronics Device Applications of Diamond Semiconductors*. Woodhead Publishing, 2018.
- [3] E. Kohn and A. Denisenko, "Concepts for diamond electronics," *Thin Solid Films*, vol. 515, no. 10, pp. 4333–4339, Mar. 2007, ISSN: 00406090. DOI: 10.1016/j.tsf.2006.07.179.
- [4] Hugh O. Pierson, *Handbook of Carbon, Graphite, and Fullerenes: Properties, Processing, and Applications.* Elsevier Science, 1993.
- [5] M. Gabrysch, S. Majdi, D. J. Twitchen, and J. Isberg, "Electron and hole drift velocity in chemical vapor deposition diamond," *Journal of Applied Physics*, vol. 109, no. 6, Mar. 2011, ISSN: 00218979. DOI: 10.1063/1.3554721.
- [6] M. Cabello, V. Soler, G. Rius, J. Montserrat, J. Rebollo, and P. Godignon, Advanced processing for mobility improvement in 4H-SiC MOSFETs: A review, May 2018. DOI: 10.1016/j.mssp.2017.10.030.
- [7] R. S. Pengelly, S. M. Wood, J. W. Milligan, S. T. Sheppard, and W. L. Pribble, A review of GaN on SiC high electron-mobility power transistors and MMICs, 2012. DOI: 10. 1109/TMTT.2012.2187535.
- [8] S. Koizumi, "n-Type doping of diamond," *Materials Research Society Symposium Proceedings*, vol. 956, no. 13, pp. 55–60, 2007, ISSN: 02729172. DOI: 10.1002/pssa. 200671407.
- M. Ding, Y. Liu, X. Lu, Y. Li, and W. Tang, "Boron doped diamond films: A microwave attenuation material with high thermal conductivity," *Applied Physics Letters*, vol. 114, no. 16, 2019, ISSN: 00036951. DOI: 10.1063/1.5083079. [Online]. Available: http://dx.doi.org/10.1063/1.5083079.
- [10] H. Sato and M. Kasu, "Maximum hole concentration for Hydrogen-terminated diamond surfaces with various surface orientations obtained by exposure to highly concentrated

NO2," *Diamond and Related Materials*, vol. 31, pp. 47–49, Jan. 2013, ISSN: 09259635. DOI: 10.1016/j.diamond.2012.10.007.

- J. Isberg, J. Hammersberg, E. Johansson, *et al.*, "High carrier mobility in single-crystal plasma-deposited diamond," *Science*, vol. 297, no. 5587, pp. 1670–1672, 2002, ISSN: 00368075. DOI: 10.1126/science.1074374.
- H. J. Looi, R. B. Jackman, and J. S. Foord, "High carrier mobility in polycrystalline thin film diamond," *Applied Physics Letters*, vol. 72, no. 3, pp. 353–355, 1998, ISSN: 00036951. DOI: 10.1063/1.120734.
- [13] G. Yang, L. Li, W. B. Lee, and M. C. Ng, *Structure of graphene and its disorders: a review*, Dec. 2018. DOI: 10.1080/14686996.2018.1494493.
- [14] A. Hirsch, *The era of carbon allotropes*, 2010. DOI: 10.1038/nmat2885.
- [15] D. D. L. Chung, "Review Graphite," Tech. Rep.
- [16] G. Peschel, "Carbon-Carbon bonds: Hybridization Peschel Carbon-Carbon bonds: Hybridization," Tech. Rep.
- [17] F. Banhart, "Elemental carbon in the sp 1 hybridization," *ChemTexts*, vol. 6, no. 1, Mar. 2020, ISSN: 21993793. DOI: 10.1007/s40828-019-0098-z.
- [18] R. A. Bonham, L. S. Bartell, ; ) K. S. Pitzer, E. Catalano, and .; L. Pauling, "Infrared and Raman Spectra," Tech. Rep. 2, 1959, p. 193. [Online]. Available: https://pubs. acs.org/sharingguidelines.
- [19] C. J. Wort and R. S. Balmer, *Diamond as an electronic material*, 2008. DOI: 10.1016/ S1369-7021(07)70349-8.
- [20] D. A. Macdonald, "Enhancement of hydrogen terminated diamond FET performance through integration of electron acceptor oxides," Ph.D. dissertation, University of Glasgow, 2019.
- [21] R. Quay, C. Moglestue, V. Palankovski, and S. Selberherr, "A temperature dependent model for the saturation velocity in semiconductor materials," Tech. Rep., 2000.
- [22] E. Bellotti, H. E. Nilsson, K. F. Brennan, P. P. Ruden, and R. Trew, "Monte Carlo calculation of hole initiated impact ionization in 4H phase SiC," *Journal of Applied Physics*, vol. 87, no. 8, pp. 3864–3871, Apr. 2000, ISSN: 00218979. DOI: 10.1063/1.372426.
- [23] D. Ji, B. Ercan, and S. Chowdhury, "Experimental Determination of Velocity-Field Characteristic of Holes in GaN," *IEEE Electron Device Letters*, vol. 41, no. 1, pp. 23–25, Jan. 2020, ISSN: 15580563. DOI: 10.1109/LED.2019.2953873.
- [24] J. H. Luong, K. B. Male, and J. D. Glennon, Boron-doped diamond electrode: Synthesis, characterization, functionalization and analytical applications, 2009. DOI: 10.1039/ b910206j.

- [25] U. F. S. D'Haenens-Johansson, J. E. Butler, and A. N. Katrusha, "Synthesis of Diamonds and Their Identification," *Reviews in Mineralogy and Geochemistry*, vol. 88, no. 1, pp. 689–753, Jul. 2022, ISSN: 1529-6466. DOI: 10.2138/rmg.2022.88.13.
- [26] S. Ferro, Synthesis of diamond, Oct. 2002. DOI: 10.1039/b204143j.
- [27] B. F.P and H. H.T, "Man-made diamonds," *Nature*, vol. 176, no. 07/09, p. 51, 1955.
- [28] R. S. Balmer, J. R. Brandon, S. L. Clewes, *et al.*, "Chemical vapour deposition synthetic diamond: Materials, technology and applications," *Journal of Physics Condensed Matter*, vol. 21, no. 36, 2009, ISSN: 09538984. DOI: 10.1088/0953-8984/21/36/ 364221.
- [29] K. G. Crawford, "Surface Transfer Doping of Diamond using Transition Metal Oxides by," 2017.
- [30] S. M. Sze and K. K. Ng, *Physics of Semiconductor Devices: Third Edition*. 2006, pp. 1–815, ISBN: 0471143235. DOI: 10.1002/9780470068328.
- [31] G. Popovici, R. G. Wilson, T. Sung, M. A. Prelas, and S. Khasawinah, "Diffusion of boron, lithium, oxygen, hydrogen, and nitrogen in type IIa natural diamond," *Journal* of Applied Physics, vol. 77, no. 10, pp. 5103–5106, 1995, ISSN: 00218979. DOI: 10. 1063/1.359320.
- [32] R. Kalish, "Doping of diamond," *Carbon*, 1999, ISSN: 00086223. DOI: 10.1016/ S0008-6223(98)00270-X.
- [33] K. Okano, H. Kiyota, T. Kurosu, and M. Iida, "Doping of diamond," *Carbon*, vol. 37, no. 5, pp. 781–785, 1999, ISSN: 00086223. DOI: 10.1016/S0008-6223(98) 00270-X.
- [34] R. Kalish, A. Reznik, C. Uzan-Saguy, and C. Cytermann, "Is sulfur a donor in diamond?" *Applied Physics Letters*, vol. 76, no. 6, pp. 757–759, 2000, ISSN: 00036951.
   DOI: 10.1063/1.125885.
- [35] Alison Mainwood, "Point Defects, Impurities and Doping," in CVD Diamond for Electrical Devices and Sensors, R. Sussmann, Ed., WILEY, 2009, ch. 3, pp. 51–55.
- [36] E. Gheeraert, N. Casanova, A. Tajani, *et al.*, "n-Type doping of diamond by sulfur and phosphorus," *Diamond and Related Materials*, vol. 11, no. 3-6, pp. 289–295, 2002, ISSN: 09259635. DOI: 10.1016/S0925-9635(01)00683-5.
- [37] I. Sakaguchi, M. N. Gamo, Y. Kikuchi, et al., "Sulfur: A donor dopant for n-type diamond semiconductors," *Physical Review B Condensed Matter and Materials Physics*, vol. 60, no. 4, R2139–R2141, 1999, ISSN: 1550235X. DOI: 10.1103/PhysRevB. 60.R2139.
- [38] S. J. Cobb, Z. J. Ayres, and J. V. Macpherson, "Boron Doped Diamond: A Designer Electrode Material for the Twenty-First Century," 2018. DOI: 10.1146/annurevanchem. [Online]. Available: https://doi.org/10.1146/annurev-anchem-.

- [39] G. Chicot, A. Fiori, P. N. Volpe, *et al.*, "Electronic and physico-chemical properties of nanometric boron delta-doped diamond structures," *Journal of Applied Physics*, vol. 116, no. 8, 2014, ISSN: 10897550. DOI: 10.1063/1.4893186. [Online]. Available: http://dx.doi.org/10.1063/1.4893186.
- [40] S. J. Sque, R. Jones, and P. R. Briddon, "Structure, electronics, and interaction of hydrogen and oxygen on diamond surfaces," *Physical Review B Condensed Matter and Materials Physics*, vol. 73, no. 8, pp. 1–15, 2006, ISSN: 10980121. DOI: 10.1103/PhysRevB.73.085313.
- [41] N. Carolina and R. June, "Negative-electron-affinity effects on the diamond (100) surface," *Physical Review B*, vol. 50, no. 8, pp. 5803–5806, 1994.
- [42] D. Takeuchi, H. Kato, G. S. Ri, *et al.*, "Direct observation of negative electron affinity in hydrogen-terminated diamond surfaces," *Applied Physics Letters*, vol. 86, no. 15, pp. 1–3, 2005, ISSN: 00036951. DOI: 10.1063/1.1900925.
- [43] F. Maier, M. Riedel, B. Mantel, J. Ristein, and L. Ley, "Origin of surface conductivity in diamond," *Physi. Rev. Lett.*, vol. 85, no. 16, pp. 3472–5, 2000, ISSN: 1079-7114.
   [Online]. Available: http://www.ncbi.nlm.nih.gov/pubmed/11690524.
- [44] M. Kasu, "Diamond field-effect transistors for RF power electronics: Novel NO2 hole doping and low-temperature deposited Al2O3 passivation," *Japanese Journal of Applied Physics*, 2017.
- [45] K. G. Crawford, D. Qi, J. McGlynn, *et al.*, "Thermally Stable, High Performance Transfer Doping of Diamond using Transition Metal Oxides," *Scientific Reports*, vol. 8, no. 1, pp. 1–9, 2018, ISSN: 20452322. DOI: 10.1038/s41598-018-21579-4. [Online]. Available: http://dx.doi.org/10.1038/s41598-018-21579-4.
- [46] D. A. Macdonald, K. G. Crawford, A. Tallaire, R. Issaoui, and D. A. Moran, "Performance Enhancement of Al2O3/H-Diamond MOSFETs Utilizing Vacuum Annealing and V2O5 as a Surface Electron Acceptor," *IEEE Electron Device Letters*, vol. 39, no. 9, pp. 1354–1357, 2018, ISSN: 07413106. DOI: 10.1109/LED.2018.2856920.
- [47] Z. Ren, J. Zhang, J. Zhang, et al., "Polycrystalline Diamond MOSFET with MoO3 Gate Dielectric and Passivation Layer," *IEEE Electron Device Letters*, vol. 38, no. 9, pp. 1302–1304, 2017, ISSN: 07413106. DOI: 10.1109/LED.2017.2727879.
- [48] M. Tordjman, K. Weinfeld, and R. Kalish, "Boosting surface charge-transfer doping efficiency and robustness of diamond with WO3 and ReO3," *Applied Physics Letters*, vol. 111, no. 11, 2017, ISSN: 00036951. DOI: 10.1063/1.4986339. [Online]. Available: http://dx.doi.org/10.1063/1.4986339.
- [49] M. Tordjman, C. Saguy, A. Bolker, and R. Kalish, "Superior Surface Transfer Doping of Diamond with MoO3," *Advanced Materials Interfaces*, vol. 1, no. 3, pp. 1–6, 2014, ISSN: 21967350. DOI: 10.1002/admi.201300155.

- [50] C. Verona, W. Ciccognani, S. Colangeli, E. Limiti, M. Marinelli, and G. Verona-Rinati, "Comparative investigation of surface transfer doping of hydrogen terminated diamond by high electron affinity insulators," *Journal of Applied Physics*, vol. 120, no. 2, 2016, ISSN: 10897550. DOI: 10.1063/1.4955469. [Online]. Available: http://dx. doi.org/10.1063/1.4955469.
- [51] M. I. Landstrass and K. V. Ravi, "Resistivity of chemical vapor deposited diamond films," *Applied Physics Letters*, vol. 55, no. 10, pp. 975–977, 1989, ISSN: 00036951. DOI: 10.1063/1.101694.
- [52] M. I. Landstrass and K. V. Ravi, "Hydrogen passivation of electrically active defects in diamond," *Applied Physics Letters*, vol. 55, no. 14, pp. 1391–1393, 1989, ISSN: 00036951. DOI: 10.1063/1.101604.
- [53] G. S. G. S.A. Grot, "The Effect of Surface Treatment on the Electrical Properties of Metal Contacts to Boron-Doped-Homoepitaxial Diamond Film," Tech. Rep.
- [54] J. Shirafuji and T. Sugino, "Electrical properties of diamond surfaces," Tech. Rep.
- [55] F. J. Himpsel, J. A. Knapp, J. A. Vanvechten, and D. E. Eastman, "Quantum photoyield of diamond(111)-A stable negative-affinity emitter," Tech. Rep. 2, 1979.
- [56] M. Kubovic and M. Kasu, "Enhancement and stabilization of hole concentration of hydrogen-terminated diamond surface using ozone adsorbates," *Japanese Journal of Applied Physics*, vol. 49, no. 11, Nov. 2010, ISSN: 00214922. DOI: 10.1143/JJAP.49. 110208.
- [57] M. Kubovic, M. Kasu, and H. Kageshima, "Sorption properties of NO2 gas and its strong influence on hole concentration of H-terminated diamond surfaces," *Applied Physics Letters*, vol. 96, no. 5, 2010, ISSN: 00036951. DOI: 10.1063/1.3291616.
- [58] H. Sato and M. Kasu, "Electronic properties of H-terminated diamond during NO 2 and O 3 adsorption and desorption," *Diamond and Related Materials*, vol. 24, pp. 99–103, Apr. 2012, ISSN: 09259635. DOI: 10.1016/j.diamond.2011.12.004.
- [59] R. Sung Gi, K. Tashiro, S. Tanaka, *et al.*, "Hall Effect Measurements of Surface Conductive Layer on Undoped Diamond Films in NO 2 and NH 3 Atmospheres," Tech. Rep. 6A, 1999, pp. 3492–3496.
- Y. Takagi, K. Shiraishi, M. Kasu, and H. Sato, "Mechanism of hole doping into hydrogen terminated diamond by the adsorption of inorganic molecule," *Surface Science*, vol. 609, pp. 203–206, Mar. 2013, ISSN: 00396028. DOI: 10.1016/j.susc.2012.12.015.
- [61] S. A. Russell, L. Cao, D. Qi, *et al.*, "Surface transfer doping of diamond by MoO3: A combined spectroscopic and Hall measurement study," *Applied Physics Letters*, vol. 103, no. 20, 2013, ISSN: 00036951. DOI: 10.1063/1.4832455.
- [62] K. G. Crawford, I. Maini, D. A. Macdonald, and D. A. Moran, "Surface transfer doping of diamond: A review," *Progress in Surface Science*, p. 100613, 2021, ISSN: 00796816.

DOI: 10.1016/j.progsurf.2021.100613. [Online]. Available: https://doi.org/10.1016/j.progsurf.2021.100613.

- [63] A. Hiraiwa, A. Daicho, S. Kurihara, Y. Yokoyama, and H. Kawarada, "Refractory twodimensional hole gas on hydrogenated diamond surface," *Journal of Applied Physics*, vol. 112, no. 12, Dec. 2012, ISSN: 00218979. DOI: 10.1063/1.4769404.
- [64] H. Kawarada, H. Tsuboi, T. Naruo, *et al.*, "C-H surface diamond field effect transistors for high temperature (400 °c) and high voltage (500V) operation," *Applied Physics Letters*, vol. 105, no. 1, pp. 1–5, 2014, ISSN: 00036951. DOI: 10.1063/1.4884828.
- [65] H. Kawarada, "High-current metal oxide semiconductor field-effect transistors on h-terminated diamond surfaces and their high-frequency operation," *Japanese Journal of Applied Physics*, vol. 51, no. 9, Sep. 2012, ISSN: 00214922. DOI: 10.1143/JJAP. 51.090111.
- [66] H. Kawarada, T. Yamada, D. Xu, *et al.*, "Diamond MOSFETs using 2D Hole Gas with 1700V Breakdown Voltage," pp. 483–486, 2016.
- [67] Z. Ren, D. Lv, J. Xu, *et al.*, "High temperature (300 °c) ALD grown Al2O3 on hydrogen terminated diamond: Band offset and electrical properties of the MOSFETs," *Applied Physics Letters*, vol. 116, no. 1, 2020, ISSN: 00036951. DOI: 10.1063/1.5126359.
- [68] S. Matsumoto and Y. Sato, "United States Patent (19) Kamo et al. (54) METHOD FOR SYNTHESIZING DIAMOND 75 Inventors: Mutsukazu Kamo, Tsuchiura," Tech. Rep.
- [69] O. A. Williams and R. B. Jackman, "Surface conductivity on hydrogen terminated diamond," *Semiconductor Science and Technology*, vol. 18, no. 3, 2003, ISSN: 02681242.
   DOI: 10.1088/0268-1242/18/3/305.
- [70] P. John and M. D. Stoikou, "Hydrogen plasma interaction with (100) diamond surfaces," *Physical Chemistry Chemical Physics*, vol. 13, no. 24, pp. 11503–11510, Jun. 2011, ISSN: 14639076. DOI: 10.1039/c1cp20099b.
- [71] K. G. Crawford, A. Tallaire, X. Li, D. A. Macdonald, D. Qi, and D. A. Moran, "The role of hydrogen plasma power on surface roughness and carrier transport in transfer-doped H-diamond," *Diamond and Related Materials*, vol. 84, no. March, pp. 48–54, 2018, ISSN: 09259635. DOI: 10.1016/j.diamond.2018.03.005.
- [72] B. Baral, S. S. M. Chan, and R. B. Jackman, "Cleaning thin-film diamond surfaces for device fabrication: An Auger electron spectroscopic study," *Journal of Vacuum Science & Technology A: Vacuum, Surfaces, and Films*, vol. 14, no. 4, pp. 2303–2307, Jul. 1996, ISSN: 0734-2101. DOI: 10.1116/1.580063.
- [73] T. Ando, K. Yamamoto, M. Ishii, M. Kamo, and Y. Sato, "Vapour-phase oxidation of diamond surfaces in O2 studied by diffuse reflectance Fourier-transform infrared and temperture-programmed desorption spectroscopy," *Journal of the Chemical Society, Faraday Transactions*, vol. 89, no. 19, pp. 3635–3640, 1993, ISSN: 09565000. DOI: 10.1039/FT9938903635.
- [74] V. Seshan, D. Ullien, A. Castellanos-Gomez, *et al.*, "Hydrogen termination of CVD diamond films by high-temperature annealing at atmospheric pressure," *Journal of Chemical Physics*, vol. 138, no. 23, 2013, ISSN: 00219606. DOI: 10.1063/1.4810866.
- [75] B. B. Pate, "THE DIAMOND SURFACE: ATOMIC AND ELECTRONIC STRUC-TURE," Tech. Rep., 1986, pp. 83–142.
- [76] R. Prins, Hydrogen spillover. Facts and fiction, May 2012. DOI: 10.1021/cr200346z.
- [77] En.wikipedia.org, Hydrogen spillover.
- [78] S. Khoobiar, "Particle to particle Migration of Hydrogen Atoms on Platinum-Alumina Catalysts from Particle to Neighboring Particles," Tech. Rep. 14, 1964, pp. 411–412.
   [Online]. Available: https://pubs.acs.org/sharingguidelines.
- [79] M. Boudart, A. W. Aldag, and M. A. Vannice, "On the slow uptake of hydrogen by platinized carbon," *Journal of Catalysis*, 1970, ISSN: 10902694. DOI: 10.1016/0021-9517 (70) 90310-6.
- [80] L. Chen, A. C. Cooper, G. P. Pez, and H. Cheng, "On the mechanisms of hydrogen spillover in MoO3," *Journal of Physical Chemistry C*, vol. 112, no. 6, pp. 1755–1758, Feb. 2008, ISSN: 19327455. DOI: 10.1021/jp7119137.
- [81] X. Sha, L. Chen, A. C. Cooper, G. P. Pez, and H. Cheng, "Hydrogen absorption and diffusion in bulk α-MoO3," *Journal of Physical Chemistry C*, vol. 113, no. 26, pp. 11 399– 11 407, Jul. 2009, ISSN: 19327455. DOI: 10.1021/jp9017212.
- [82] S. Triwahyono, A. A. Jalil, S. N. Timmiati, N. N. Ruslan, and H. Hattori, "Kinetics study of hydrogen adsorption over Pt/MoO3," *Applied Catalysis A: General*, vol. 372, no. 1, pp. 103–107, Jan. 2010, ISSN: 0926860X. DOI: 10.1016/j.apcata.2009.10. 024.
- [83] Y. Gao, E. Hu, G. Yin, and Z. Huang, "Pd nanoparticles supported on CeO2 nanospheres as efficient catalysts for dehydrogenation from additive-free formic acid at low temperature," *Fuel*, vol. 302, Oct. 2021, ISSN: 00162361. DOI: 10.1016/j.fuel.2021. 121142.
- [84] D. Messou, V. Bernardin, F. Meunier, *et al.*, "Origin of the synergistic effect between TiO2 crystalline phases in the Ni/TiO2-catalyzed CO2 methanation reaction," *Journal of Catalysis*, vol. 398, pp. 14–28, Jun. 2021, ISSN: 10902694. DOI: 10.1016/j.jcat. 2021.04.004.
- [85] Y. Ma, X. Zhang, L. Cao, and J. Lu, "Effects of the morphology and heteroatom doping of CeO2support on the hydrogenation activity of Pt single-atoms," *Catalysis Science and Technology*, vol. 11, no. 8, pp. 2844–2851, Apr. 2021, ISSN: 20444761. DOI: 10. 1039/d0cy02279a.
- [86] W. Karim, C. Spreafico, A. Kleibert, *et al.*, "Catalyst support effects on hydrogen spillover," *Nature*, vol. 541, no. 7635, pp. 68–71, 2017, ISSN: 14764687. DOI: 10.1038/nature20782.
  [Online]. Available: http://dx.doi.org/10.1038/nature20782.

- [87] M. Choi, S. Yook, and H. Kim, Hydrogen Spillover in Encapsulated Metal Catalysts: New Opportunities for Designing Advanced Hydroprocessing Catalysts, Apr. 2015. DOI: 10.1002/cctc.201500032.
- [88] J. Im, H. Shin, H. Jang, H. Kim, and M. Choi, "Maximizing the catalytic function of hydrogen spillover in platinum-encapsulated aluminosilicates with controlled nanostructures," *Nature Communications*, vol. 5, Feb. 2014, ISSN: 20411723. DOI: 10.1038/ ncomms4370.
- [89] H. Shen, H. Li, Z. Yang, and C. Li, *Magic of hydrogen spillover: Understanding and application*, Dec. 2022. DOI: 10.1016/j.gee.2022.01.013.
- [90] D. S. Pyle, E. M. A. Gray, and C. J. Webb, *Hydrogen storage in carbon nanostructures via spillover*, Nov. 2016. DOI: 10.1016/j.ijhydene.2016.08.061.
- [91] P. C. Mitchell, A. J. Ramirez-Cuesta, S. F. Parker, J. Tomkinson, and D. Thompsett, "Hydrogen spillover on carbon-supported metal catalysts studied by inelastic neutron scattering. Surface vibrational states and hydrogen riding modes," *Journal of Physical Chemistry B*, vol. 107, no. 28, pp. 6838–6845, Jul. 2003, ISSN: 15206106. DOI: 10.1021/jp0277356.
- [92] R. E. Thomas, R. A. Rudder, and R. J. Markunas, "Thermal desorption from hydrogenated and oxygenated diamond (100) surfaces," *Journal of Vacuum Science & Technology A: Vacuum, Surfaces, and Films*, vol. 10, no. 4, pp. 2451–2457, 1992, ISSN: 0734-2101. DOI: 10.1116/1.577983.
- [93] K. G. Crawford, J. D. Weil, P. B. Shah, et al., "Diamond Field-Effect Transistors with V2O5-Induced Transfer Doping: Scaling to 50-nm Gate Length," *IEEE Transactions* on Electron Devices, vol. 67, no. 6, pp. 2270–2275, Jun. 2020, ISSN: 15579646. DOI: 10.1109/TED.2020.2989736.
- [94] D. A. Macdonald, K. G. Crawford, A. Tallaire, R. Issaoui, and D. A. Moran, "Performance Enhancement of Al2O3/H-Diamond MOSFETs Utilizing Vacuum Annealing and V2O5 as a Surface Electron Acceptor," *IEEE Electron Device Letters*, vol. 39, no. 9, pp. 1354–1357, Sep. 2018, ISSN: 07413106. DOI: 10.1109/LED.2018.2856920.
- [95] C. Verona, W. Ciccognani, S. Colangeli, *et al.*, "Gate-Source Distance Scaling Effects in H-Terminated Diamond MESFETs," *IEEE Transactions on Electron Devices*, vol. 62, no. 4, pp. 1150–1156, Apr. 2015, ISSN: 00189383. DOI: 10.1109/TED.2015. 2398891.
- [96] S. A. Russell, S. Sharabi, A. Tallaire, and D. A. Moran, "Hydrogen-terminated diamond field-effect transistors with cutoff frequency of 53 GHz," *IEEE Electron Device Letters*, vol. 33, no. 10, pp. 1471–1473, 2012, ISSN: 07413106. DOI: 10.1109/LED.2012. 2210020.

- [97] W. Wang, C. Hu, F. N. Li, *et al.*, "Palladium Ohmic contact on hydrogen-terminated single crystal diamond film," *Diamond and Related Materials*, vol. 59, pp. 90–94, Oct. 2015, ISSN: 09259635. DOI: 10.1016/j.diamond.2015.09.012.
- [98] K. Tsugawa, H. Noda, K. Hirose, and H. Kawarada, "Schottky barrier heights, carrier density, and negative electron affinity of hydrogen-terminated diamond," *Physical Review B - Condensed Matter and Materials Physics*, vol. 81, no. 4, Jan. 2010, ISSN: 10980121. DOI: 10.1103/PhysRevB.81.045303.
- [99] H. Kawarada, M. Aoki, and M. Ito, "Enhancement mode metal-semiconductor field effect transistors using homoepitaxial diamonds," *Applied Physics Letters*, vol. 65, no. 12, pp. 1563–1565, 1994, ISSN: 00036951. DOI: 10.1063/1.112915.
- [100] C. Verona, W. Ciccognani, S. Colangeli, *et al.*, "V 2 O 5 MISFETs on H-Terminated Diamond," vol. 63, no. Cvd, pp. 1–7, 2016.
- [101] Z. Ren, J. Zhang, J. Zhang, and C. Zhang, "Diamond Field Effect Transistors With MoO 3 Gate Dielectric," vol. 38, no. 6, pp. 786–789, 2017.
- [102] K. Tsugawa, K. Kitatani, H. Noda, *et al.*, "High-preformance diamond surface-channel field-effect transistors and their operation mechanism," *Diamond and Related Materials*, vol. 8, no. 2-5, pp. 927–933, 1999, ISSN: 09259635.
- [103] K. Hirama', H. Takayanagil, S. Yamauchil, Y. Jingul, H. Umezawa2, and H. Kawaradal, "High-performance p-channel diamond MOSFETs with alumina gate insulator," Tech. Rep., 2007.
- [104] X. Yu, J. Zhou, S. Zhang, Z. Cao, Y. Kong, and T. Chen, "High frequency H-diamond MISFET with output power density of 182 mW/mm at 10 GHz," *Applied Physics Letters*, vol. 115, no. 19, 2019, ISSN: 00036951. DOI: 10.1063/1.5125771.
- [105] J. W. Liu, M. Y. Liao, M. Imura, *et al.*, "Interfacial band configuration and electrical properties of LaAlO 3/Al2O3/hydrogenated-diamond metal-oxide-semiconductor field effect transistors," *Journal of Applied Physics*, vol. 114, no. 8, Aug. 2013, ISSN: 00218979. DOI: 10.1063/1.4819108.
- Y. F. Wang, W. Wang, X. Chang, *et al.*, "Performance of hydrogen-terminated diamond MOSFET with bilayer dielectrics of YSZ/Al2O3," *Diamond and Related Materials*, vol. 99, Nov. 2019, ISSN: 09259635. DOI: 10.1016/j.diamond.2019.107532.
- [107] J. Liu, M. Liao, M. Imura, A. Tanaka, H. Iwai, and Y. Koide, "Low on-resistance diamond field effect transistor with high-k ZrO2 as dielectric," *Scientific Reports*, vol. 4, Sep. 2014, ISSN: 20452322. DOI: 10.1038/srep06395.
- [108] Y. F. Wang, W. Wang, X. Chang, *et al.*, "Hydrogen-terminated diamond field-effect transistor with AlO x dielectric layer formed by autoxidation," *Scientific Reports*, vol. 9, no. 1, pp. 1–5, 2019, ISSN: 20452322. DOI: 10.1038/s41598-019-41082-8.
- [109] K. Hirama, H. Sato, Y. Harada, H. Yamamoto, and M. Kasu, "Diamond field-effect transistors with 1.3A/mm drain current density by Al 2O 3 passivation layer," *Japanese*

*Journal of Applied Physics*, vol. 51, no. 9, 2012, ISSN: 00214922. DOI: 10.1143/JJAP.51.090112.

- [110] Y. Kitabayashi, T. Kudo, H. Tsuboi, et al., "Normally-Off C-H Diamond MOSFETs with Partial C-O Channel Achieving 2-kV Breakdown Voltage," *IEEE Electron Device Letters*, vol. 38, no. 3, pp. 363–366, Mar. 2017, ISSN: 07413106. DOI: 10.1109/LED. 2017.2661340.
- [111] D. Oing, M. Geller, A. Lorke, and N. Wöhrl, "Tunable carrier density and high mobility of two-dimensional hole gases on diamond: The role of oxygen adsorption and surface roughness," *Diamond and Related Materials*, vol. 97, no. June, p. 107 450, 2019, ISSN: 09259635. DOI: 10.1016/j.diamond.2019.107450. [Online]. Available: https://doi.org/10.1016/j.diamond.2019.107450.
- [112] C. E. Nebel, C. Sauerer, F. Ertl, *et al.*, "Hydrogen-induced transport properties of holes in diamond surface layers," *Applied Physics Letters*, vol. 79, no. 27, pp. 4541–4543, Dec. 2001, ISSN: 00036951. DOI: 10.1063/1.1429756.
- [113] K. Hayashi, S. Yamanaka, H. Watanabe, T. Sekiguchi, H. Okushi, and K. Kajimura, "Atomic force microscopy study of atomically flat (001) diamond surfaces treated with hydrogen plasma," Tech. Rep., 1998, pp. 1–124.
- [114] H. Jiang, F. Liu, H. Yan, L. Si, and Z. Dou, "Etching effects of hydrogen plasma treatment on diamond surfaces," *Surface and Coatings Technology*, vol. 363, pp. 12–17, Apr. 2019, ISSN: 02578972. DOI: 10.1016/j.surfcoat.2019.02.007.
- P. John and M. D. Stoikou, "Hydrogen plasma interaction with (100) diamond surfaces," *Physical Chemistry Chemical Physics*, vol. 13, no. 24, pp. 11503–11510, Jun. 2011, ISSN: 14639076. DOI: 10.1039/clcp20099b.
- [116] N. Arjmandi, "Resist Homogeneity," in Updates in Advanced Lithography, S. Hosaka, Ed., Rijeka: IntechOpen, 2013, ch. 6. DOI: 10.5772/56618. [Online]. Available: https://doi.org/10.5772/56618.
- [117] S. Thoms and P. Reynolds, "PG introduction & workflow," Tech. Rep.
- [118] D. S. MacIntyre, O. Ignatova, S. Thoms, and I. G. Thayne, "Resist residues and transistor gate fabrication," *Journal of Vacuum Science and Technology B: Microelectronics and Nanometer Structures*, vol. 27, no. 6, pp. 2597–2601, 2009, ISSN: 10711023. DOI: 10. 1116/1.3243176.
- [119] J. V. Sci, "Plasma-polymer interactions : A review of progress in understanding polymer resist mask durability during plasma etching for nanoscale fabrication," vol. 010801, no. 2011, 2015.
- [120] B. Mackenzie and T. Sharp, "Oxford Instruments ALD (Atomic Layer Deposition) Basics of ALD," Tech. Rep., 2015.
- [121] A. Hiraiwa, T. Saito, D. Matsumura, and H. Kawarada, "Isotope analysis of diamondsurface passivation effect of high-temperature H2O-grown atomic layer deposition-Al2O3

films," *Journal of Applied Physics*, vol. 117, no. 21, p. 215 304, Jun. 2015, ISSN: 0021-8979. DOI: 10.1063/1.4921824. [Online]. Available: https://doi.org/10.1063/1.4921824.

- [122] J. B. Cui, J. Ristein, and L. Ley, "Dehydrogenation and the surface phase transition on diamond "111...: Kinetics and electronic structure," Tech. Rep.
- [123] C. Su and J. C. Lin, "Thermal desorption of hydrogen from the diamond C(100) surface," *Surface Science*, 1998, ISSN: 00396028. DOI: 10.1016/S0039-6028(98)00107-1.
- [124] A. Mohammed and A. Abdullah, "SCANNING ELECTRON MICROSCOPY (SEM): A REVIEW," Tech. Rep.
- [125] W. Zhou, R. P. Apkarian, Z. Lin Wang, and D. Joy, "Fundamentals of Scanning Electron Microscopy," Tech. Rep.
- [126] A. Ul-Hamid, "A Beginners' Guide to Scanning Electron Microscopy," Tech. Rep.
- [127] S. Sinha Ray, "Techniques for characterizing the structure and properties of polymer nanocomposites," in *Environmentally Friendly Polymer Nanocomposites*, Elsevier, 2013, pp. 74–88. DOI: 10.1533/9780857097828.1.74.
- [128] B. Voigtländer, "NanoScience and Technology Atomic Force Microscopy Second Edition," Tech. Rep. [Online]. Available: http://www.springer.com/series/ 3705.
- [129] Y. F. Dufrêne, Atomic force microscopy, a powerful tool in microbiology, Oct. 2002. DOI: 10.1128/JB.184.19.5205-5213.2002.
- [130] "USER MANUAL," Tech. Rep., 2011. [Online]. Available: www.bruker-axs.com.
- [131] G. Dutta, U. V. Waghmare, T. Baidya, and M. S. Hegde, "Hydrogen spillover on CeO2/Pt: Enhanced storage of active hydrogen," *Chemistry of Materials*, vol. 19, no. 26, pp. 6430–6436, Dec. 2007, ISSN: 08974756. DOI: 10.1021/cm071330m.
- B. D. Adams, C. K. Ostrom, S. Chen, and A. Chen, "High-performance Pd-based hydrogen spillover catalysts for hydrogen storage," *Journal of Physical Chemistry C*, vol. 114, no. 46, pp. 19875–19882, Nov. 2010, ISSN: 19327447. DOI: 10.1021/jp1085312.
- [133] Y. Kaibara, K. Sugata, M. Tachiki, H. Umezawa, and H. Kawarada, "Control wettability of the hydrogen-terminated diamond surface and the oxidized diamond surface using an atomic force microscope," *Diamond and Related Materials*, vol. 12, no. 3-7, pp. 560– 564, 2003, ISSN: 09259635. DOI: 10.1016/S0925-9635(02)00373-4.
- [134] A. D. Lueking and R. T. Yang, "Hydrogen spillover to enhance hydrogen storage Study of the effect of carbon physicochemical properties," *Applied Catalysis A: General*, 2004, ISSN: 0926860X. DOI: 10.1016/j.apcata.2004.01.019.
- [135] W. Chen, D. Qi, X. Gao, and A. T. S. Wee, "Surface transfer doping of semiconductors," *Progress in Surface Science*, vol. 84, no. 9-10, pp. 279–321, 2009, ISSN: 00796816. DOI:

10.1016/j.progsurf.2009.06.002.[Online]. Available: http://dx.doi. org/10.1016/j.progsurf.2009.06.002.

- [136] M. Mertens, M. Mohr, K. Brühne, *et al.*, "Patterned hydrophobic and hydrophilic surfaces of ultra-smooth nanocrystalline diamond layers," *Applied Surface Science*, vol. 390, pp. 526–530, Dec. 2016, ISSN: 01694332. DOI: 10.1016/j.apsusc.2016.08.130.
- [137] J. Golden, H. Miller, D. Nawrocki, and J. Ross, "Optimization of Bi-layer Lift-Off Resist Process," Tech. Rep.
- [138] "LOR and PMGI Resists for Bi-layer Lift-off Processing," Tech. Rep.
- [139] Q. Zeng, L. Wang, L. Shi, *et al.*, "Optimizing hydrogen plasma etching process of preferred (110)-textured diamond film," *Surface and Coatings Technology*, vol. 228, no. SUPPL.1, Aug. 2013, ISSN: 02578972. DOI: 10.1016/j.surfcoat.2012.05.097.
- [140] Y. Yamazaki, K. Ishikawa, N. Mizuochi, and S. Yamasaki, "Structural change in diamond by hydrogen plasma treatment at room temperature," in *Diamond and Related Materials*, vol. 14, Nov. 2005, pp. 1939–1942. DOI: 10.1016/j.diamond.2005. 09.011.
- K. Xing, A. Tsai, S. Rubanov, *et al.*, "Palladium forms Ohmic contact on hydrogenterminated diamond down to 4 K," *Applied Physics Letters*, vol. 116, no. 11, 2020, ISSN: 00036951. DOI: 10.1063/1.5141775.
- [142] X. Yuan, J. Liu, S. Shao, *et al.*, "Thermal stability investigation for Ohmic contact properties of Pt, Au, and Pd electrodes on the same hydrogen-terminated diamond," *AIP Advances*, vol. 10, no. 5, May 2020, ISSN: 21583226. DOI: 10.1063/5.0008167.
- [143] S. Russell, S. Sharabi, A. Tallaire, and D. A. Moran, "RF operation of hydrogen-terminated diamond field effect transistors: A comparative study," *IEEE Transactions on Electron Devices*, vol. 62, no. 3, pp. 751–756, 2015, ISSN: 00189383. DOI: 10.1109/TED. 2015.2392798.
- K. G. Crawford, L. Cao, D. Qi, et al., "Enhanced surface transfer doping of diamond by V2O5 with improved thermal stability," Applied Physics Letters, vol. 108, no. 4, pp. 1–5, 2016, ISSN: 00036951. DOI: 10.1063/1.4940749. [Online]. Available: http://dx.doi.org/10.1063/1.4940749.
- [145] T. Wade, M. W. Geis, T. H. Fedynyshyn, *et al.*, "Effect of surface roughness and H-termination chemistry on diamond's semiconducting surface conductance," *Diamond and Related Materials*, vol. 76, no. April, pp. 79–85, 2017, ISSN: 09259635. DOI: 10.1016/j.diamond.2017.04.012. [Online]. Available: http://dx.doi.org/10.1016/j.diamond.2017.04.012.
- [146] L. Ge, Q. Zhang, Y. Peng, M. Xu, and X. Xu, "Research on Stability of H-Terminated Diamond MESFET," in 2023 20th China International Forum on Solid State Lighting

and 2023 9th International Forum on Wide Bandgap Semiconductors, SSLCHINA: IFWS 2023, Institute of Electrical and Electronics Engineers Inc., 2023, pp. 313–314, ISBN: 9798350385373. DOI: 10.1109/SSLChinaIFWS60785.2023.10399681.

- [147] Y. F. Wang, X. Chang, X. Zhang, et al., "Normally-off hydrogen-terminated diamond field-effect transistor with Al2O3 dielectric layer formed by thermal oxidation of Al," *Diamond and Related Materials*, vol. 81, no. November 2017, pp. 113–117, 2018, ISSN: 09259635. DOI: 10.1016/j.diamond.2017.11.016. [Online]. Available: https://doi.org/10.1016/j.diamond.2017.11.016.
- [148] J. W. Liu, M. Y. Liao, M. Imura, and Y. Koide, "Normally-off HfO2-gated diamond field effect transistors," *Applied Physics Letters*, vol. 103, no. 9, pp. 1–5, 2013, ISSN: 00036951. DOI: 10.1063/1.4820143.
- [149] N. Oi, T. Kudo, M. Inaba, *et al.*, "Normally-OFF Two-Dimensional Hole Gas Diamond MOSFETs Through Nitrogen-Ion Implantation," *IEEE Electron Device Letters*, vol. 40, no. 6, pp. 933–936, 2019, ISSN: 15580563. DOI: 10.1109/LED.2019.2912211.
- [150] Y. Fu, Y. Chang, X. Zhu, R. Xu, Y. Xu, and H. Kawarada, "Normally-Off Oxidized Si-Terminated (111) Diamond MOSFETs via ALD-Al2O3Gate Insulator With Drain Current Density Over 300 mA/mm," *IEEE Transactions on Electron Devices*, vol. 69, no. 8, pp. 4144–4152, Aug. 2022, ISSN: 15579646. DOI: 10.1109/TED.2022. 3186865.
- [151] Y. Sasama, T. Kageura, M. Imura, *et al.*, "High-mobility p-channel wide-bandgap transistors based on hydrogen-terminated diamond/hexagonal boron nitride heterostructures," *Nature Electronics*, vol. 5, no. 1, pp. 37–44, Jan. 2022, ISSN: 25201131. DOI: 10.1038/s41928-021-00689-4.
- [152] R. T. Tung, "Chemical Bonding and Fermi Level Pinning at Metal-Semiconductor Interfaces," Tech. Rep., 2000.
- [153] J. W. Liu, H. Oosato, M. Y. Liao, and Y. Koide, "Enhancement-mode hydrogenated diamond metal-oxide-semiconductor field-effect transistors with Y2O3 oxide insulator grown by electron beam evaporator," *Applied Physics Letters*, vol. 110, no. 20, pp. 2–7, 2017, ISSN: 00036951. DOI: 10.1063/1.4983091. [Online]. Available: http://dx.doi.org/10.1063/1.4983091.
- [154] B. Liu, T. Bi, Y. Fu, et al., "MOSFETs on (110) C-H Diamond: ALD Al2O3/Diamond Interface Analysis and High Performance Normally-OFF Operation Realization," *IEEE Transactions on Electron Devices*, vol. 69, no. 3, pp. 949–955, Mar. 2022, ISSN: 15579646.
   DOI: 10.1109/TED.2022.3147152.
- [155] Y. Kitabayashi, T. Kudo, H. Tsuboi, et al., "Normally-Off C-H Diamond MOSFETs with Partial C-O Channel Achieving 2-kV Breakdown Voltage," *IEEE Electron Device Letters*, vol. 38, no. 3, pp. 363–366, 2017, ISSN: 07413106. DOI: 10.1109/LED. 2017.2661340.

- [156] J. Su, W. Wang, G. Shao, G. Chen, and H. X. Wang, "Mobility-enhanced normally off hydrogen-terminated diamond FET with low interface state density using Al2O3/Nd gate stack," *Applied Physics Letters*, vol. 123, no. 17, Oct. 2023, ISSN: 00036951. DOI: 10.1063/5.0171832.
- [157] F. Wang, G. Q. Chen, W. Wang, et al., "High-threshold-voltage and low-leakage-current of normally-off H-diamond FET with self-aligned Zr/ZrO2 gate," *Diamond and Related Materials*, vol. 134, Apr. 2023, ISSN: 09259635. DOI: 10.1016/j.diamond. 2023.109774.
- [158] M. Zhang, W. Wang, S. Fan, *et al.*, "Normally-off hydrogen-terminated diamond field effect transistor with yttrium gate," *Carbon*, vol. 176, pp. 307–312, May 2021, ISSN: 00086223. DOI: 10.1016/j.carbon.2021.01.121.
- [159] W. Wang, Y. Wang, M. Zhang, *et al.*, "An Enhancement-Mode Hydrogen-Terminated Diamond Field-Effect Transistor With Lanthanum Hexaboride Gate Material," vol. 41, no. 4, pp. 585–588, 2020.
- [160] Z. Ren, W. Chen, J. Zhang, et al., "High Performance Single Crystalline Diamond Normally-Off Field Effect Transistors," *IEEE Journal of the Electron Devices Society*, vol. 7, no. March, pp. 88–99, 2019, ISSN: 21686734. DOI: 10.1109/JEDS.2018. 2880005.
- [161] M. Zhang, W. Wang, F. Wen, *et al.*, "Large VTHof Normally-OFF Field Effect Transistor With Yttrium Gate Material Directly Deposited on Hydrogen-Terminated Diamond," *IEEE Transactions on Electron Devices*, vol. 69, no. 7, pp. 3563–3567, Jul. 2022, ISSN: 15579646. DOI: 10.1109/TED.2022.3174047.
- [162] K. Romanjek, F. Andrieu, T. Ernst, and G. Ghibaudo, "Improved split C-V method for effective mobility extraction in sub-0.1-μm Si MOSFETs," *IEEE Electron Device Letters*, vol. 25, no. 8, pp. 583–585, Aug. 2004, ISSN: 07413106. DOI: 10.1109/LED. 2004.832786.
- [163] S. Mileusnic', M. Zivanov', and P. Habas, "MOS TRANSISTORS CHARACTERIZA-TION BY SPLIT C-V METHOD," Tech. Rep.
- [164] S. Kim, S. H. Lee, I. H. Jo, J. Seo, Y. E. Yoo, and J. H. Kim, "Influence of growth temperature on dielectric strength of Al2O3 thin films prepared via atomic layer deposition at low temperature," *Scientific Reports*, vol. 12, no. 1, Dec. 2022, ISSN: 20452322. DOI: 10.1038/s41598-022-09054-7.
- [165] S. F. Szymanski, P. Rowlette, and C. A. Wolden, "Self-limiting deposition of aluminum oxide thin films by pulsed plasma-enhanced chemical vapor deposition," *Journal of Vacuum Science & Technology A: Vacuum, Surfaces, and Films*, vol. 26, no. 4, pp. 1079– 1084, Jul. 2008, ISSN: 0734-2101. DOI: 10.1116/1.2891258.
- [166] Q. He, K. Su, J. Zhang, et al., "High Mobility Normally-OFF Hydrogenated Diamond Field Effect Transistors With BaF2Gate Insulator Formed by Electron Beam Evapora-

tor," *IEEE Transactions on Electron Devices*, vol. 69, no. 3, pp. 1206–1210, Mar. 2022, ISSN: 15579646. DOI: 10.1109/TED.2022.3147738.

[167] J. Tsunoda, M. Iwataki, K. Horikawa, *et al.*, "Low On-Resistance Vertical-Type 2-D Hole Gas Diamond MOSFETs with Trench Gate Structure," no. 001, pp. 1–7, 2021.